THE BIOPHYSICAL PROPERTIES OF PLANTAR CALLUS AND THE RELATIONSHIP BETWEEN PRESSURE AND CALLUS DEVELOPMENT AND REGRESSION

Ciaran R. F. WRIGHT

Ph.D. Thesis

2015

THE BIOPHYSICAL PROPERTIES OF PLANTAR CALLUS AND THE RELATIONSHIP BETWEEN PRESSURE AND CALLUS DEVELOPMENT AND REGRESSION

Ciaran R. F. WRIGHT

School of Health Sciences University of Salford, Salford, UK

Submitted in Partial Fulfilment of the Requirements of the Degree of Doctor of Philosophy, November 2015

'Next time someone complains that you have made a mistake, tell him that may be a good thing. Because without imperfection, neither you nor I would exist.' **Stephen W. Hawking, 2010**.

Table of Contents

Acknowledgements	xiv
Preface: Funding and my role as a PhD student	XV
Abstract	xvi
Chapter 1: Introduction	1
Chapter 2: Background and critical review of the literature	
Introduction	
Part A. Plantar callus	3
A.1 Introduction	3
A.2 Consequences of callus and corns	5
A.3 Overview of interventions for callus and corns	6
A.4 Overview of skin structure	9
A.5 Histology and biochemistry of callus development	11
A.6 Plantar pressure	14
A.6.1 Application of loads to the plantar surface	14
A.6.2 Footwear	16
A.6.3 Activity levels	
A.6.4 Bony prominences	
A.6.5 Disease	
A.6.6 Plantar pressure and callus	
A.7 Plantar shear pressure	
A.8 How the epidermis adapts to external forces	
A.9 Gaps in the research into callus	
Part B. Skin measurement devices	
B.1 Introduction to skin measurement parameters	
B.2 Skin surface hydration	

B.3 Trans-epidermal water loss (TEWL)	40
B.4 Measurement of the mechanical properties of the skin	42
B.5 Imaging of the skin	48
B.6 Measurement of skin topography	52
B.7 Gaps in skin measurement research	55
Conclusion	56
Chapter 3: Inter-rater and inter-day reliability of non-invasive instruments biophysical properties of foot skin	neasuring the
3.1 Introduction	57
3.2 Aims and objectives	58
3.3 Materials and methods	58
3.3.1 Subjects	58
3.3.2 Data collection	59
3.3.3 Statistical analysis	61
3.4 Results	63
3.4.1 Corneometer [®]	63
3.4.2 Cutometer [®]	65
3.4.3 Visioscan [®]	67
3.4.4 Order of measurement observations	69
3.5 Discussion	70
3.6 Conclusions	74
Chapter 4: The biophysical characteristics of normal and callused plantar skin	75
4.1 Introduction	75
4.2 Aims and objectives	76
4.3 Materials and methods	76
4.3.1 Subjects	76

4	4.3.2 Data collection	76
4	4.3.3 Statistical analysis	78
4.4	4 Results	81
Δ	4.4.1 Skin hydration	
Ζ	4.4.2 – Skin distensibility	
Ζ	4.4.3 Skin Topography – homogeneity	
2	4.4.4 Skin Topography – variance	
Ζ	4.4.5 Correlations between variables	
2	4.4.6 Differences between callus grades	
4.5	5 Discussion	86
4.6	5 Conclusion	92
Chapt	ter 5: Development of a device to apply external loads to plantar skin	93
5.1	I Introduction	93
5.2	2 Determining the forces and pressures to be applied by the loading device	93
5.3	3 Design of loading device	97
5	5.3.1 Components	97
5	5.3.2 Loading programme sequences	99
5	5.3.3 Design of device housing	104
5	5.3.4 Device modifications	104
5.4	4 Compression and shear load tests	106
5	5.4.1 Rationale and aim	106
5	5.4.2 Materials and methods	106
5	5.4.4 Results	107
5	5.4.5 Discussion	111
5.5	5 Shear load cell tests	112
5		
	5.5.1 Rationale and aim	112

5.5.3 Results	112
5.5.4 Discussion	113
5.4.6 Conclusion	113
5.6 Compression tests	114
5.6.1 Aims and methods of compression tests	114
5.6.2 Compression test results on plywood	114
5.6.3 Compression test results on medium density EVA material	
5.5.4 Discussion	
5.5.6 Conclusion	126
Chapter 6: A pilot study to investigate whether laboratory controlled plantar loadin	g causes callus-
like thickening in healthy skin overlying the plantar metatarsal area	
6.1 Introduction	127
6.2 Preliminary tests	128
6.2.1 Variability of positioning foot in insole	
6.2.2 Adjustment of plantar pressures applied	129
6.2.3 Duration of pressures applied	
6.3 Pilot study methodology	130
6.3.1 Participant selection	130
6.3.2 In-shoe pressure measurement	130
6.3.3 Positioning the foot in relation to the loading device	
6.3.4 Measurement of skin biophysical properties	
6.3.5 Activity monitoring	134
6.3.7 Barefoot pressure measurement	135
6.3.8 Foot loading data	136
6.3.9 Data analysis	
6.4 Case study – Subject 1	138
6.4.1 Participant profile	
6.4.2 Loading profile	138

6.4.3 Skin data	139
6.5 Case study – Subject 2	142
6.5.1 Participant profile	142
6.6 Case study – Subject 3	143
6.6.1 Participant profile	143
6.6.2 Loading profile	143
6.6.3 Skin data	144
6.7 Case study – Subject 4	147
6.7.1 Participant profile	147
6.7.2 Loading profile	147
6.7.3 Skin data	148
6.8 Case study – Subject 5	151
6.8.1 Participant profile	151
6.8.2 Loading profile	151
6.8.3 Skin data	152
6.9 Discussion	155
6.10 Conclusion and implications for further research	161
Chapter 7: The effect of pressure reduction on the biophysical properties of plantar callus	163
7. 1 Introduction	163
7.2 Review of the literature	163
7.2.1. Efficacy of customised insoles	163
7.2.2. Methods of insole design and manufacture	165
7.2.3. Insole materials	166
7.2.4. Pressure relieving additions to insoles	169
7.2.5. Insoles used to treat plantar callus	171
7.2.6. Conclusion	173
7.3 Study aims and objectives	173

7.4 Methodology	.173
7.4.1. Subjects	.173
7.4.2. Insole design and manufacture	.174
7.4.3. Data collection	.175
7.4.4. Statistical analysis	.177
7.5. Results	.178
7.5.1. Participant and pressure profile	.178
7.5.2. Skin hydration data	.178
7.5.3. Skin distensibility data	.180
7.5.4. Skin homogeneity data	.182
7.5.5. Skin variance data	. 183
7.6 Discussion	.185
7.7 Conclusion	.191
Chapter 8: Overall discussion and conclusion	. 192
8.1 Summary of findings	.192
8.2 Implications for clinicians and researchers	.197
8.3 Recommendations for further research	.198
8.4 Conclusion to Ph. D. Thesis	.201
Appendix 1: Considerations for loading device improvements for further research projects u	ısing
shear pressures	. 203
Improve shear movement	.203
Improvements in the accuracy of pressure delivery	.203
Improvements in controlling loading cycles	.204
References	. 205

List of Tables

Table A.1 – Epidermal structure	10
Table A.2 – Dermal-Epidermal junction and dermis	10
Table A.3 – Plantar skin characteristics	10
Table A.4 – Biochemical triggers in callus and corn development	12
Table A.5 Other studies reporting significant pressure increases beneath calluses	26
Table A.6. Peak pressure changes following callus debridement	26
Table A.7 - Gaps in the research into callus	35
Table B.1 – List of measurement devices and alternative measurement modalities	
Table 3.1 – Corneometer [®] inter-rater reliability	64
Table 3.2 – Corneometer [®] inter-day reliability	65
Table 3.3 – Cutometer [®] inter-rater reliability	66
Table 3.4 – Cutometer [®] inter-day reliability	67
Table 3.5 – Visioscan [®] homogeneity inter-rater reliability	68
Table $3.6 - \text{Visioscan}^{\text{B}}$ homogeneity inter-day reliability	68
Table 3.7 – Visioscan [®] variance inter-rater reliability	69
Table 3.8 – Visioscan [®] variance inter-day reliability	69
Table 4.1 Correlation coefficient between biophysical parameters at each skin site	86
Table 5.1 – Normal pressures under each metatarsal head	95
Table 5.2 – Normal plantar pressures per kilogram of bodyweight	95
Table 5.3 – Details of main device components	98
Table 5.4 – Results of compression tests on plywood	115
Table 5.5 – Results of compression tests on EVA	120
Table 6.1 Subject 1 loading profile	139
Table 6.2 Subject 1 smallest detectable change for hydration	140
Table 6.3 Subject 1 smallest detectable change for stiffness	141
Table 6.4 Subject 1 smallest detectable change for homogeneity	141
Table 6.5 Subject 1 smallest detectable change for variance	142
Table 6.6 Subject 3 loading profile	144
Table 6.7 Subject 3 smallest detectable change for hydration	145
Table 6.8 Subject 3 smallest detectable change for stiffness	146
Table 6.9 Subject 3 smallest detectable change for homogeneity	146
Table 6.10 Subject 3 smallest detectable change for variance	147

Table 6.11 Subject 4 loading profile	148
Table 6.12 Subject 4 smallest detectable change for hydration	149
Table 6.13 Subject 4 smallest detectable change for stiffness	150
Table 6.14 Subject 4 smallest detectable change for homogeneity	150
Table 6.15 Subject 4 smallest detectable change for variance	151
Table 6.16 Subject 5 loading profile	152
Table 6.17 Subject 5 smallest detectable change for hydration	153
Table 6.18 Subject 5 smallest detectable change for stiffness	154
Table 6.19 Subject 5 smallest detectable change for homogeneity	154
Table 6.20 Subject 5 smallest detectable change for variance	155
Table 7.1 – Pressure changes beneath forefoot using pressure relieving additions re-	eported
in literature	169
Table 7.2 Pressure reduction group participant profile	178
Table 7.3 Percentage change between time points at each skin site	180
Table 7.4 Percentage change between callus and control sites at each time point	180
Table 7.5 Smallest detectable change for hydration data	180
Table 7.6 Percentage change between time points at each skin site	181
Table 7.7 Percentage change between callus and control sites at each time point	181
Table 7.8 Smallest detectable change for distensibility data	182
Table 7.9 Percentage change between time points at each skin site	183
Table 7.10 Percentage change between callus and control sites at each time point	183
Table 7.11 Smallest detectable change for homogeneity data	183
Table 7.12 Percentage change between time points at each skin site	184
Table 7.13 Percentage change between callus and control sites at each time point	185
Table 7.14 Smallest detectable change for variance data	185

List of Figures

Figure A.1 – Prevalence of corns and callus reported in literature
Figure A.2 – Gender differences reported in the literature
Figure A.3 – Plantar skin histology specimen11
Figure A.4 – Molecular model for callus development
Figure A.5 - Vertical (compressive) and antero-posterior (shear) forces under the foot
during gait15
Figure A.6 – Causes of plantar callus16
Figure A.7 – Specimens of mouse ear epidermis after 7 days of daily friction application 33
Figure B.1 - Measurement principle of the Cutometer
Figure B.2 – Diagram of the Reviscometer® probe46
Figure B.3 – Low frequency ultrasound image of tissue overlying the first metatarsal head
Figure B.4 – OCT image of plantar heel skin51
Figure B.5 - Callus photograph and corresponding dermatoscope image53
Figure B.6 - Visioscan [®] image of callused skin54
Figure B.7 - Visioline [®] replica of a callus strip55
Figure 3.1 – Skin measurement sites
Figure 4.1 – Skin measurement sites
Figure 4.2 – Mean and 95% CI hydration data for each of the five measurement sites82
Figure 4.3 – Mean and 95% CI distensibility data for each of the five measurement sites.
Figure 4.4 – Mean and 95% CI homogeneity data for each of the four measurement sites.
Figure 4.5 – Mean and 95% CI variance data for each of the four measurement sites85
Figure 5.1 – Anatomy of loading device
Figure 5.2 – Diagram showing limiter switch positions
Figure 5.3 – Device housing showing holes in wall and aperture on platform to allow
loading of forefoot
Figure 5.4 – Replaced components
Figure 5.5 - Compression cell measurements: posterior shear, device locked at 'in'
position107
Figure 5.6 – Shear cell measurements: posterior shear, device locked in 'in' position 108

Figure 5.7 - Compression cell measurements: posterior shear, device locked in 'out'
position
Figure 5.8 – Shear cell measurements: posterior shear, device locked in 'in' position 109
Figure 5.9 – Compression cell measurements: anterior shear, device locked at 'in' position
Figure 5.10 – Shear cell measurements: anterior shear, device locked at 'in' position 110
Figure 5.11 - Compression cell measurements: anterior shear, device locked at 'out'
position110
Figure 5.12 – Shear cell measurements: anterior shear, device locked at 'out' position .111
Figure 5.13 – Results of shear cell tests
Figure 5.14 - Relationship between pressure applied and force delivered to device on
plywood115
Figure 5.15 – 2 Bars compression test data
Figure 5.16 – 2.5 Bars compression test data
Figure 5.17 – 3 Bars compression test data
Figure 5.18 – 3.5 Bars compression test data
Figure 5.19 – 4 Bars compression test data
Figure 5.20 – 4.5 Bars compression test data
Figure 5.21 – 5 Bars compression test data
Figure 5.22 – 5.5 Bars compression test data119
Figure 5.23 – 6 Bars compression test data
Figure 5.24 – Relationship between pressure applied and force delivered to device on EVA
material
Figure 5.25 – EVA 2 Bars compression test results
Figure 5.26 – EVA 2.5 Bars compression test results
Figure 5.27 – EVA 3 Bars compression test results
Figure 5.28 – EVA 3.5 Bars compression test results
Figure 5.29 – EVA 4 Bars compression test results
Figure 5.30 – EVA 4.5 Bars compression test results
Figure 5.31 – EVA 5 Bars compression test results
Figure 5.32 – EVA 5.5 Bars compression test results
Figure 5.33 – EVA 6 Bars compression test results
Figure 6.1 – Bontrager et al. (1997) Pedar [®] mask used in study using percentages of insole
length and width

Figure 6.2 – The skin measurement sites	134
Figure 6.3 – Subject 1 raw hydration data	140
Figure 6.4 – Subject 1 raw distensibility data	140
Figure 6.5 – Subject 1 raw homogeneity data	141
Figure 6.6 – Subject 1 raw variance data	142
Figure 6.7 – Subject 3 raw hydration data	145
Figure 6.8 – Subject 3 raw distensibility data	145
Figure 6.9 – Subject 3 raw homogeneity data	146
Figure 6.10 – Subject 3 raw variance data	147
Figure 6.11 – Subject 4 raw hydration data	149
Figure 6.12 – Subject 4 raw distensibility data	149
Figure 6.13 – Subject 4 raw homogeneity data	150
Figure 6.14 – Subject 4 raw variance data	151
Figure 6.15 – Subject 5 raw hydration data	153
Figure 6.16 – Subject 5 raw distensibility data	153
Figure 6.17 – Subject 5 raw homogeneity data	
Figure 6.18 – Subject 5 raw variance data	
Figure 7.1 – The skin measurement sites of the callus	176
Figure 7.2 – Raw skin hydration data	179
Figure 7.3 – Raw skin distensibility data	
Figure 7.4 – Raw skin homogeneity data	
Figure 7.5 – Raw skin variance data	

Acknowledgements

I have been very fortunate during the process of my PhD to receive support and guidance from many individuals. I would firstly like to thank my supervisors Professor Chris Nester, Dr Anita Williams and Dr Farina Hashmi for their guidance, patience and inspiration throughout this PhD. Their input has helped me develop into a confident and independent researcher and I will never forget all of the support given to me during difficult times. I would like to thank Mr Paul Busby for all of his input in designing and building the skin loading device based on my specifications, and also for the countless modifications he made to it during my early pilot tests. I would like to thank the prosthetics and orthotics workshop technicians Mr Bernie Noakes and Mr Geoff Riley for their assistance and for letting me use their equipment in the workshop whilst I was building the platform for my test rig and pushing back my somewhat limited boundaries of woodwork. I would like to thank my colleague and friend Dr Daniel Parker for his assistance during my PhD, for being a test 'guinea pig' during piloting, and for manufacturing the insoles used in my final study. I would like to thank Mrs Sue Royle and all of the reception and podiatry staff in the Podiatry Clinic, for letting me recruit from their patient records and attend various clinics in the process. I would like to say a big thank you to all of my colleagues from the office that have helped to make my PhD journey enjoyable and have provided friendship and support through all of the ups and downs of the past few years. I would like to thank Reckitt Benckiser who provided my income and funded the projects I worked on. I would like to thank my mum, dad, sister, grandparents and my girlfriend Simone for all of their unselfish support throughout the past few years and for being there for me when I needed them most. I am so grateful for everything they have done for me. Finally, I would like to thank all of the volunteers who served as study participants during my PhD; their time and generosity has been greatly appreciated.

Preface: Funding and my role as a PhD student

My PhD was sponsored by Reckitt Benckiser who own Scholl foot care brand. The sponsorship is part of a prior research contract concerned with better understanding of corns, callus and heel fissures, and the efficacy of Scholl treatments for these skin conditions. Scholl involvement in the PhD has been at the level of conceiving the general topic of the PhD, i.e. the relationship between pressure and callus, and the need to better understand the properties of callus.

During my PhD I have been involved in all aspects of all of the studies contained in the thesis. The work detailed in Chapters 3, 4 and 7 were undertaken as part of larger projects and involved other researchers. My role in the instrument reliability study (Chapter 3) was protocol development, data collection, processing and analysis. My role for skin characterisation study (Chapter 4) was assisting in protocol development, some data collection, and all data processing and analysis. Some of the data was collected by a HPC registered podiatrist to ensure effective delivery of the contract research work by the research group. However, I processed and analysed everything used in this thesis. I undertook all aspects of the work described to design and implement the loading device (Chapter 5), supported in the device design and production by Mr Paul Busby (he completed CAD diagrams and advised on technical aspects and manufacturing). I am responsible for all aspects of the skin loading pilot study (Chapter 6). In the final study which concerned pressure reduction using orthotic insoles (Chapter 7), I worked with Dr Daniel Parker who designed the insoles used. I was involved in protocol design and collected all of the skin data and some of the in-shoe pressure data. I also processed and analysed all data for this chapter.

At the time of writing, work undertaken for the reliability study (Chapter 3) and skin characterisation study (Chapter 4) have been published (Hashmi et al., 2015a, Hashmi et al., 2015b). The publications include the callus work detailed in these chapters but also work on corns and heel fissures which were included in the Reckitt Benckiser research contract. In addition two further papers have been submitted for publication which detail clinical trials concerning the treatment of callus (Hashmi et al., under review) and heel fissures (Hashmi et al., under review) comparing standard podiatric treatment with off-the-shelf home treatments. This work extended from the characterisation study.

<u>Abstract</u>

Plantar calluses are common skin lesions which often require professional treatment by podiatrists. They commonly present under prominent areas such as the metatarsal heads and can cause significant discomfort during ambulation. Furthermore, they are one of the known risk factors for ulceration in individuals with systemic conditions such as diabetes mellitus. Anecdotal evidence suggests that mechanical factors contribute to calluses and there are numerous studies linking callus to increased peak plantar pressure. However, whether callus is a result of increased pressure or vice versa remains unclear. Skin on other areas of the body has been shown to respond to external loading forces, but no research has specifically investigated the relationship between callus and pressure.

A critical review of the literature explored the methods used for skin profiling through biophysical skin measurement. Skin hydration, distensibility and topography were revealed to be useful measurement parameters to characterise plantar skin and for this study, three devices were chosen for testing these parameters. However, as these devices have not received much attention for testing plantar callus in previous research, the first study investigated their repeatability on normal and callused plantar skin. These devices were shown to provide adequate measures of skin properties so they were then used in a larger scale study investigating the biophysical characteristics of normal and callused plantar skin. It was found that callused skin was less hydrated, less distensible, and rougher in texture than normal plantar skin.

Work was then undertaken to develop a device that could apply loads to plantar skin in a safe manner so that the skin's response to external loading could be assessed. A subsequent pilot study was conducted to assess whether normal plantar skin in individuals prone to callus would display callus-like skin changes as a result of increased vertical pressure applied by the skin loading device over a minimum period of six weeks. After the skin loading period, no effect could be observed in normal plantar skin properties. The reasons for this are explored in depth. A study was then undertaken in order to assess the effect of plantar pressure reduction in callused skin over a period of 12 weeks. Pressure reduction was achieved by using customised insoles worn by the study participants. No change in callused skin properties was observed and the reasons for this are explored in depth. These studies provide a strong starting point in understanding the link between pressure and callus and provide a foundation for further research.

Chapter 1: Introduction

Plantar calluses are hyperkeratotic skin lesions which commonly present under areas of pressure such as the metatarsal heads (Singh, 1996). They are extremely common, reportedly affecting up to 68% of individuals (White and Mulley, 1989). The lesions are around two to three times thicker than normal plantar skin (Thomas et al., 1985) and can be very painful (Helfand, 2003) which can lead to mobility problems in elderly individuals (Menz and Lord, 2001). In individuals suffering from diabetes, calluses can also increase the risk of ulceration (Murray et al., 1996) which subsequently decreases quality of life (Firth et al., 2011).

It is believed that external insults, such as poorly fitting shoes, contribute to callus development (Singh, 1996) and there is sufficient evidence to link calluses to increased peak pressures under the forefoot (Menz et al., 2007). However little is known about this relationship and whether callus causes increased pressures or vice versa. While hyperkeratotic-like skin responses have been observed as a result of applying loads to human skin (Rubin, 1949), no studies have been conducted on the foot which specifically show increased loading to be a precursor to callus development. This area of study is important because until more is known about the relationship between callus and loading, preventative strategies cannot progress.

In order to investigate the relationship between callus and loading, researchers must first know about the properties of normal and callused skin so the skin's response to loading can be quantified. However, knowledge in this area is also scant. Therefore, the aims of this thesis are to (1) investigate the biophysical properties of normal and callused skin; and (2) quantify the skin's response to increases and decreases in external loads.

The work undertaken for this thesis satisfies the criteria for a PhD as outlined in the Descriptor For A Higher Education Qualification At Level 8: Doctoral Degree, published as part of the framework for higher education qualifications in England, Wales and Northern Ireland (QAA, 2008, p.23). The first section of this thesis (Chapter 2) critically reviews the available literature on plantar callus to identify more precisely the research gaps. The review also examines different biophysical devices available that are designed to measure skin properties and will discuss potentially useful devices to study important skin parameters. Work undertaken for this chapter demonstrates the acquisition of a substantial knowledge base essential for doctoral candidates. Chapter 3 describes a study which aimed to assess the reliability of biophysical instruments for use on callused and normal skin

types, and demonstrates research techniques and knowledge of academic enquiry. Chapter 4 presents a larger scale project to characterise normal and callused skin using these devices, adding new knowledge through an original research project. Chapter 5 describes the development of a device which was designed to apply external loads to the foot. Chapter 6 details a pilot study which assessed whether it was possible to generate a callus-like skin response through load application using the device developed in Chapter 5. Chapter 7 describes a study that was conducted to investigate any changes in callus properties following reduction of plantar pressures using orthotic insoles. Chapters 5–7 are examples of an ability to design, implement, and evaluate research, making informed judgements on research issues, problem solving and tying together skills and attributes gained during the earlier stages of the PhD. Finally, Chapter 8 discusses the work in this thesis as a whole and how it contributes to future research, and clinical practice; and thus the contribution to knowledge made through this PhD thesis.

Chapter 2: Background and critical review of the literature

Introduction

The aims of this critical review of the literature are to provide an overview of plantar callus including its prevalence, consequences and treatments. Literature on skin anatomy, callus histology and biochemistry; and vertical and shear pressures will also be reviewed and the research gaps highlighted. The second part of the review will deal with different devices that are designed to measure the biophysical properties of the skin and will consider the devices that might be most helpful in plantar skin research. The gaps in the literature relating to these devices will also be discussed.

Part A. Plantar callus

A.1 Introduction

Plantar callus and corns are hyperkeratotic lesions which commonly present under areas of pressure such as the metatarsal heads. Hyperkeratosis refers to increased thickening of the stratum corneum, and it has been suggested that in normal skin this is a physiological mechanism to protect underlying soft tissues from mechanical stress (Thomas et al., 1985, Singh, 1996), but in plantar skin these lesions can become symptomatic and as a result cause problems. Calluses can be differentiated from corns as they normally occupy a greater surface area and are diffuse, while corns are well defined and circular with a central visible core (Singh, 1996, Freeman, 2002). The most common areas for callus development on the forefoot include the first, second and fifth metatarsal heads respectively but 'roll-off' calluses are also common on the medial aspect of the first metatarsal head and interphalangeal joint (Woodburn and Helliwell, 1996, Potter and Potter, 2000a, Springett et al., 2003, Menz et al., 2007, Spink et al., 2009).

Corns and callus are a common problem and historically presented to NHS podiatry (Farndon et al., 2009), accounting for up to 68% of foot complaints reported in the literature (Figure A.1) (Ebrahim et al., 1981, Black and Hale, 1987, White and Mulley, 1989, Robbins et al., 1996, Helfand et al., 1998, Dunn, 2004, Spink et al., 2009). The highest rates have been observed in community (White and Mulley, 1989, Dunn, 2004, Spink et al., 2009) and workplace surveys (Springett et al., 2003), and those attending foot screening programmes (Robbins et al., 1996); while the lowest were observed in hospital patients or those attending hospital clinics (Ebrahim et al., 1981, Black and Hale, 1987)

and those recruited in retirement homes (Helfand, 2003). It could be hypothesised given the potential link between the development of these lesions and pressure that there is a relation to levels of activity. Therefore the populations investigated in hospitals, clinics and care homes may be less active than those populations in community and workplace settings.



Figure A.1 – Prevalence of corns and callus reported in literature

Most of the literature into prevalence of these lesions has focussed on the elderly population (Ebrahim et al., 1981, Black and Hale, 1987, White and Mulley, 1989, Helfand et al., 1998, Dunn, 2004, Spink et al., 2009), presumably due to an increased prevalence of foot problems with age. However other groups such as the working population, athletes, and the homeless population also have a high prevalence (Robbins et al., 1996, Adams, 2002, Springett et al., 2003). Gender differences have been reported in several studies

(Figure A.2) with females showing a higher prevalence of corns and callus, as well as general non-traumatic foot problems (Gorter et al., 2000). The reasons for this could stem from choice of footwear, for example studies have shown that wearing high heels increases forefoot peak pressures (Snow et al., 1992, Mandato and Nester, 1999, Yung-Hui and Wei-Hsien, 2005) and shifts pressure from the lateral to the medial forefoot (Snow et al., 1992, Yung-Hui and Wei-Hsien, 2005).



Figure A.2 – Gender differences reported in the literature

A.2 Consequences of callus and corns

The primary consequence of plantar hyperkeratoses in otherwise healthy individuals have been listed as pain, burning, and tenderness during gait (Helfand, 2003) which is troublesome as the lesions are located at weight bearing areas. Foot pain has been shown to have a whole host of consequences of its own; a negative impact on balance, walking, and climbing stairs (Menz and Lord, 2001), and it increases the risk of falls (Benvenuti et al., 1995, Mickle et al., 2010). Aside from pain and associated disability, the presence of hyperkeratotic lesions in individuals with diabetes can greatly increase the risk of ulceration (Murray et al., 1996, Reiber et al., 1999, Sage et al., 2001, Nishide et al., 2009). In turn, the presence of foot ulcers, particularly non-healing ulcers, in patients with rheumatoid arthritis and diabetes contributes to a decreased quality of life that can be attributed to decreased mobility, which affects social function and leisure activities (Ribu et al., 2008, Firth et al., 2011). While the above studies show that pain from calluses may affect mobility, there is a lack of qualitative literature on patient perspectives of living with

callus and how these lesions affect their quality of life. These types of studies would be useful and may help to increase the volume of research conducted in the area.

A.3 Overview of interventions for callus and corns

The routine treatment for callus and corns is sharp or scalpel debridement. The profession that usually carries out this skilled procedure is that of podiatry. Aside from possible effects on plantar peak pressures (discussed in depth later), treatment studies have shown that debridement of callus may significantly reduce pain (Redmond et al., 1999, Woodburn et al., 2000, Balanowski and Flynn, 2005) and improve walking function (Balanowski and Flynn, 2005) in elderly adults and patients with rheumatoid arthritis. Only two randomised controlled trials have been conducted (Davys et al., 2005, Landorf et al., 2013) using real and sham treatments. While both the debridement and sham groups in each study showed improvements in pain scores, no significant improvement in pain reduction between treatment groups was found. Furthermore, Davys et al. (2005) found no significant difference in walking function between the two groups. Interestingly they found a nonsignificant reduction in pressure in the debridement group and an increase in pressure in the sham group compared to baseline immediately post-intervention. This could potentially be due to placebo effect. Studies assessing the placebo effect have reported that suggestion that an active treatment has been given can produce a placebo effect (reviewed by Price et al., 2008). The 9% increase in pressure found in the sham group could have been due to an expectation of improvement due to treatment, possibly resulting in increased loading during gait as a result of being less tentative with weight bearing over a painful callus. The small (< 5%) increase in contact area in the sham group may illustrate this. There was also an increase in walking speed in both groups (9% and 12.3% for scalpel debridement and sham groups respectively) which could have led to higher peak pressure values.

Siddle et al. (2013) conducted a randomised controlled trial that compared the efficacy of using scalpel debridement alongside a combined therapeutic approach to a control group receiving just a combined therapeutic approach over an 18 month period. They found that while pain was significantly reduced in both groups, there was no significant difference between the groups which suggests that scalpel debridement offers no benefit over combined therapy alone in individuals with rheumatoid arthritis. Again, there was no significant improvement in walking function. The authors suggest that

debridement should only be used in emergency cases where tissue integrity is at a serious risk (Siddle et al., 2013).

Aside from scalpel debridement, there are alternative approaches to callus and corn treatment. Salicylic acid, another widely used treatment, has been involved in research in corn treatment on healthy subjects (n = 198). The authors (Lang et al., 1994) found salicylic acid to be significantly more effective at removing corn tissue than placebo treatments with 62.4% of corns removed compared with 17.9% (p < 0.001). This may be due to the fact that salicylic acid activates plasmin which causes desquamation, and increases levels of plasminogen activators while decreasing inhibitors (Heda and Roberts, 2008). However, far more work focussing on corns is needed to better understand their physiology.

Other less common approaches have also been studied. A study by Colagiuri et al. (1995) on patients with diabetes (n = 9) presenting with plantar callus assessed the effect of a yearlong treatment using a custom made thermal pliable plastic insole worn seven hours per day. Clinical assessment of photographs taken before and after the study, by consensus of the three researchers who were blinded to patient and treatment, suggested that there was a significant improvement in callus grade (p < 0.02) compared with patients treated by scalpel debridement (n = 11) where no significant reduction was found. The orthotic patients were checked after 1-3 weeks and then at three monthly intervals as with the debridement group but no details on how the integrity of the orthotic insoles was assessed and whether or not they were routinely replaced are reported. The actual pressure reduction caused by the orthotic was also not measured. The poor level of callus reduction in the debridement group may have highlighted a need for more regular debridement which might have given better results. The limited qualitative measures employed in this study for assessing the skin and the interventions are a major pitfall. The results would have been more credible had quantitative measures of pressure changes and changes in callus lesions been performed. However, despite its drawbacks, this study is useful because it indicates the potential efficacy of orthoses as a treatment modality, but also highlights the need for much better quality research to understand the pathophysiology of callus.

A subsequent study looking at the effects of pressure relieving measures in adolescents with diabetes presenting with plantar callus has been conducted (Duffin et al., 2003). The authors recruited 211 participants for plantar pressure measures. They also assessed pressure reducing qualities in a small subgroup of individuals with callus (n = 17) and without callus (n = 17). They found over both groups that cushioning, orthoses without

cushioning and combined cushioning and orthoses significantly reduced peak plantar pressures (p = 0.001, 0.05 and < 0.001 respectively). Out of their sample, they recruited 23 subjects who used custom made orthotic insoles and 67 control subjects who did not use insoles and reassessed the calluses after a year. A significant reduction in in-shoe pressure was found (p = 0.0003) in the orthoses group when walking with no intervention but no significant difference was found in the control group. Out of the group that wore insoles over one year, six had callus at the beginning of the trial and in two the callus fully resolved. In the subjects who did not wear insoles, seven had callus at the beginning of the trial, and all seven still had callus after one year. However, no detail is given as to whether individuals were permitted to use foot treatments and no quantitative evaluation of the plantar skin as a whole is presented. Furthermore, no information is given as to whether the insoles were still delivering effective pressure reduction at the follow-up appointment. This brings into question whether the results were due to pressure reduction or other factors such as occlusion of the skin caused by the insole material, or use of topical treatments. While flawed, this study does indicate a possible link between pressure reduction and callus regression.

Khan et al. (1996) conducted a blinded randomised controlled trial of thirty patients comparing marigold tegetes erecta treatment to a placebo and found a significant reduction in callus dimensions and pain in the treatment group (p < 0.001). Furthermore, the authors found that applying a protective pad over the marigold preparation was significantly more effective than withholding the protective pad for both pain and callus dimensions (p < 0.001).

Akdemir et al. (2011) assessed the efficacy of a topical treatment containing cantharidin, salicylic acid and podophylin (Canthacur CS), which is applied after debridement. The sample included 65 plantar and 7 palmar calluses and corns. The patients were followed up over the period of a year with a maximum of four treatments over this period if necessary. Clinical assessment and patient satisfaction data was collected at each appointment. The authors reported a 100% success rate for all participants over a year with 79.2% of lesions resolved after one treatment, and 91.7%, 98.6% and 100% of calluses resolved after the second, third and fourth treatments respectively. However, no quantitative measures of the lesions were taken, no control group was used, and no randomisation or blinding was employed which makes the credibility of the trial questionable. Barnes and Brocklesby (2011) assessed the effect of using a silicone gel sheet (Cica-Care) on four patients with plantar callus and two with plantar scar tissue.

They reported a reduction in lesion dimensions and pain score in all cases but no quantitative data was presented in the paper to support this. As this employed a case study research design, the authors did not randomise participants into groups or blind participants.

The literature surrounding the treatment of callus is somewhat limited with the most rigorous studies concentrated to scalpel debridement treatment. While other studies have reported callus reduction with less common methods of treatment (Colagiuri et al., 1995, Khan et al., 1996, Akdemir et al., 2011, Barnes and Brocklesby, 2011) only one of these (Khan et al., 1996) provided quantitative measures, namely the dimensions of the callus. All others have relied upon subjective clinical examination which may be insensitive to the true physiological changes (or lack of) created by a treatment. This highlights the need for quantitative measures of callus morphology in order to accurately characterise lesions and thereafter measurement of treatment efficacy. However, aside from measuring dimensions of the lesions, there is a lack of other quantitative measures of callus properties, which may offer more sensitive measures of callus.

A.4 Overview of skin structure

The epidermis, the most superficial of the two skin layers, is composed primarily of keratinocytes which account for around 95% of its cells (Mackie, 2003). Keratinocytes are in a constant state of self-renewal with new cells being produced in deep layers to replace those shed superficially. As the cells migrate superficially, they change from living cells to dead sheets of keratin (Wigley et al., 2008). The anatomy of the epidermis, the dermal-epidermal junction and the features of plantar skin are shown in Tables A.1 – A.3 and Figure A.3.

Table A.1 – Epidermal structure (in the order of deep to superficial)

Layer	Features
Stratum	One-cell-thick wall of dividing keratinocytes of a columnar/cuboidal shape which replace
basale	those lost superficially (Wigley et al., 2008).
Stratum	Composed of keratinocytes with a characteristic 'prickly' appearance due to the presence
spinosum	of desmosomes, important structural filaments which aid in cell cohesion, maintaining
-	structure (Young et al., 2006, Wigley et al., 2008).
Stratum	The layer where keratinisation begins. Within this layer, lamellar granules appear and
granulosum	merge with the cell membrane, and these release glycophospholipids into the intercellular
-	space forming the main constituent of the water permeability barrier (Smith et al., 1982,
	Potts and Francoeur, 1990, Imokawa et al., 1991, Wigley et al., 2008).
Stratum	In plantar (and palmar) skin, there is thought to be an additional thin layer of incompletely
lucidum	keratinised cells (Wigley et al., 2008). However, this might also be an artefact of histology
	specimen processing (Mackie, 2003).
Stratum	Accounts for much of the epidermal thickness (the precise thickness of this layer varies
corneum	with skin site), and consists of keratin sheets left by dead keratinocytes. The stratum
	corneum is replaced in around 16 days (Thomas et al., 1985, Wigley et al., 2008).

Table A.2 – Dermal-Epidermal junction and dermis

Structure	Role		
Dermal-	It is composed	d of types IV, V and VII collagen and functions to anchor the epidermis to	
epidermal	the dermis, providing mechanical support and forming a permeable barrier between the		
junction (DEJ)	two skin layers (Briggaman and Wheeler, 1975, Briggaman, 1982).		
Dermis	The deeper of the two skin layers, the dermis, is composed of collagen and elastic		
	networks and contains nerves, vessels and lymphatics and provides mechanical strength		
	(Wigley et al., 2008).		
	Papilliary	Contains finer layers of (type III) collagen and is responsible for providing	
	layer	mechanical support to the epidermis. It connects to the basal epidermal	
		layer through the DEJ via keratin filaments and collagen fibrils. Finger-	
		like rete ridges containing types I and III collagen interdigitate with rete	
		pegs of the epidermis to provide epidermal anchorage (Wigley et al.,	
		2008).	
	Reticular	Contains thicker type I and III collagen fibres and elastic fibres and gives	
	layer	the skin much of its strength (Wigley et al., 2008).	

Table A.3 – Plantar skin characteristics

Feature	Role
Thick epidermis (due to stratum	Protection of dermis and cubcutaneous tissues (Palastanga
corneum)	and Soames, 2012).
Epidermis and dermis tightly bound	Well-developed rete ridges prevent horizontal skin
together	displacement, allowing for improved grip (Young et al.,
-	2006).



Figure A.3 – Plantar skin histology specimen showing the two layers of skin, the epidermis (Ep) and dermis (D). Within the epidermis, the stratum corneum (K), stratum granulosum (G), sweat ducts (As) and rete ridges (RR) are all visible (Young et al., 2006 p.184).

A.5 Histology and biochemistry of callus development

Plantar skin is known to be anatomically different from skin on other sites of the body (Table A.3; Figure A.3), and has been found to have a particular prominence of the keratin filaments K6, K7, K9, K16 and K17 which relate to the mechanical demands place upon it (Swensson et al., 1998). Histologically, callused skin has been shown to be markedly different from normal plantar skin. Thomas et al. (1985) observed callus to be 2-3 times thicker (p < 0.001) than normal plantar stratum corneum (normal stratum corneum = 123 ± 12 cell layers; callused stratum corneum = 349 ± 67 cell layers). The callus corneocytes had a similar surface area to normal plantar corneocytes. However, their volume was increased and their density was decreased. The greater number of cell layers present in callused stratum corneum indicates increased proliferation and the decreased density of the cells and indicates that the cells are not as well differentiated as normal plantar corneocytes in this layer (Thomas et al., 1985). It could be suggested that the increased rate of cell production in callus is a factor in the poorer cell differentiation as they are not given sufficient time to fully mature.

Biochemically, it has long been suspected that increased corneocyte cohesion plays a role in hyperkeratosis (Rubin, 1949). Kim et al. (2010) confirmed this through immunohistochemistry and found several agents to be increased in anterior heel skin (which displays callus-like properties). Keratin genes K9 and K14 are expressed in greater quantities, as are the proteins involcurin, filaggrin, caspase 14, and CaSR, while PAR2 was found to be decreased (Kim et al., 2010). Levels of specific adhesion proteins (CDSN, DSG1 and DSC1) were also elevated. Corns have been linked to increased plasminogen activator inhibitor levels which decreases the rate of differentiation and desquamation (shedding of the stratum corneum) (Heda and Roberts, 2008). Collectively these increased levels of chemical triggers would likely cause an increased rate of proliferation and increased cell cohesion (Table A.4, Figure A.4) which would directly affect the mechanical properties of the skin and also the skin's barrier function.

Feature	Role	
Keratin K9	Activated by the gene Wnt5a and is expressed primarily in plantar keratinocytes of	
	epidermal ridges above the stratum basale, and is seen during terminal differentiation	
	(programmed cell death) (Knapp et al., 1986, Swensson et al., 1998, Rinn et al., 2008).	
	It has been suggested that this specific keratin aids in reinforcing plantar skin making it	
	more resistant to mechanical stresses (Swensson et al., 1998).	
Keratin K14	Has an integral role in the skin barrier and is mutated in various skin diseases (Engelke	
	et al., 1997, Jensen et al., 2000, Santos et al., 2002).	
Protease	Responsible for increasing the intracellular concentration of calcium ions and has thus	
activated	been shown to inhibit lamellar body secretion of lipids, delay epidermal barrier	
receptor 2	recovery and increase terminal differentiation (Demerjian et al., 2008, Jeong et al.,	
(PAR2)	2008).	
Calcium sensing	Important in modulating calcium physiology. In the stratum basale, induces a pathway	
receptor (CaSR)	which causes differentiation resulting in an increased rate of epidermal permeability	
	barrier formation and cell cohesion (by activating E-Cadherin) (Turksen and Troy,	
	2003, Tu et al., 2008).	
Caspase 14	Increases the rate of corneocyte differentiation and terminal differentiation as a	
	response to skin barrier damage (Raymond et al., 2007, Demerjian et al., 2008). It also	
	has a crucial role in degradation of filaggrin and thus the generation of natural	
	moisturising factors (NMFs) (Hoste et al., 2011).	
Filaggrin	Profilaggrin is the precursor to filaggrin and is formed in the stratum granulosum as	
	part of the keratohyalin granules but later breaks down into filaggrin towards the	
	stratum corneum. It is a structural protein involved in cohesion of keratin filaments and	
	thus plays a part in the skin barrier. It also contributes to the formation of cornified	
	envelopes (Simon et al., 1996, McGrath and Uitto, 2008, Sandilands et al., 2009, Hoste	
	et al., 2011). It is degraded in the stratum corneum by Caspase 14 into amino acids	
	(Hoste et al., 2011).	
Involucrin	Expressed in the suprabasal layers of the epidermis and acts as scaffolding in the wall	
	of the cornified cell envelope (Yaffe et al., 1992, Steinert and Marekov, 1997).	
Plasminogen	Expressed by keratinocytes at different stages of differentiation and converts	
activators	plasminogen into plasmin. Plasminogen activators and their inhibitors have a role in	
	terminal differentiation changes in cell morphology, formation of the cornified cell	
	envelope and are distributed in the same regions as involucrin (Chen et al., 1993). They	
	are also involved in regulation of desquamation (Lyons-Giordano and Lazarus, 1995).	

Table A.4 – Biochemical triggers in callus and corn development



Figure A.4 – Molecular model for callus development (Kim et al., 2010 p.500). CaSR (calcium sensing receptor); KRT (keratin); CDSN, DSG1, DSC1 (adhesion proteins); PAR2 (protease-activated receptor 2).

The above studies can be used to hypothesise what might happen to the barrier properties of callused skin. The skin barrier is provided by lipids which help to prevent water loss from the skin and protect it from external chemical insults (Madison, 2003), thus the skin's barrier function is dependent on its lipid profile. The lipids are expressed during maturation of corneocytes and have several roles. They are situated in the intercellular spaces and regulate the permeation of water to prevent desiccation through their multi-lamellar organisation (which contributes to the skin's water holding and barrier function); as well as assisting in corneocyte cohesion in the stratum corneum (Elias and Friend, 1975, Elias et al., 1981, Smith et al., 1982, Wertz and van den Bergh, 1998, Imokawa et al., 1991, Baroni et al., 2012). The decreased differentiation of callused corneoocytes observed by Thomas et al. (1985), which could in part be due to a decrease in PAR2 (Kim et al., 2010), may alter the lipid profile of the skin. In cases where cells of the stratum corneum hyperproliferate, such as in plantar callus, the cells might not have adequate time to fully differentiate which reflects what was observed in callus by Thomas et al. (1985). Because the cells have not been given time to differentiate properly, this could lead to fragile cornified cell envelopes, which has been observed in soap-induced

dry skin (Harding et al., 2003). In biochemical experiments, these fragile cell envelopes stain positive for involucrin in the superficial aspect of the stratum corneum or in areas where the skin barrier is disrupted (Hirao et al., 2001); this was one of the chemical triggers found in increased quantities in callus-type skin by Kim et al. (2010). Thomas et al. (1985) noted that, at the most superficial aspect of the stratum corneum, there was a higher desquamation rate which might reflect scaling as seen in dry skin conditions as a result of fragile cornified envelopes. The fragile cornified envelopes will have a negative impact on the skin's barrier function (Wickett and Visscher, 2006). This will lead to an increased trans-epidermal water loss (TEWL) and decreased skin hydration as a result (Baroni et al., 2012). Decreased barrier function (and resulting decreased hydration) can also directly result from mechanical insults to the skin (Baroni et al., 2012). If there is a link between callus and external loading, the delivery of the loads to the callused site may directly affect the skin barrier prior to causing alterations in chemical triggers.

There is also evidence to suggest that hydration affects the mechanical properties of the skin. Studies have observed that increased moisture content of the stratum corneum leads to degradation of corneodesmosomes and intercellular lipids which probably leads to decreased cohesion resulting in increased desquamation (Warner et al., 1999, Bouwstra et al., 2003, Wu et al., 2006). It would be expected that in cases such as callused skin, skin stiffness is likely to accelerate with decreased hydration as cohesion is increased between keratinocytes.

The impact of the structural and biochemical changes in plantar skin has never been tested. As hypothesised above, an increased TEWL and decreased stratum corneum hydration is likely to be a feature of callused skin due to altered lipid profile while the increased cell production and cohesion of the corneocytes will increase the thickness of the stratum corneum. This is likely to increase the stiffness of the skin. Measures of these features could be useful therefore in characterising the properties of calluses and corns and changes in these due to interventions. However, to date there are few reports on these characteristics in plantar skin.

A.6 Plantar pressure

A.6.1 Application of loads to the plantar surface

The main ground reaction forces that occur during gait can be described as follows: the heel strikes the ground producing a vertical peak force approximately 1.2 times body

weight, coinciding with a posterior shear force of 0.2 times bodyweight (17% of vertical component). As the centre of mass raises to its highest point and the body is directly above the limb, a trough in vertical force is seen as load is transferred from the heel to the forefoot. At this point the vertical force is 0.7 times body weight and the horizontal force becomes 0. The heel then lifts and the bodyweight is transferred onto the forefoot (which is already loaded at this point) which then begins to propel the body forwards, causing a second vertical force peak of 1.2 times bodyweight which coincides with an anterior shear force of 0.2 times bodyweight (17% of vertical component) that passes through the forefoot. The vertical and horizontal forces then reduce as bodyweight is transferred to the opposite foot during the toe-off movement (Richards, 2008). This process produces a graph with two peaks for vertical loading force and a posterior and anterior peak for horizontal shear force (Figure A.5). The medio-lateral force magnitudes are relatively small compared to the antero-posterior shear forces (Richards, 2008).



Ground Reaction Forces: Normal Gait

Figure A.5 – Vertical (compressive) and antero-posterior (shear) forces under the foot during gait (adapted from Kirtley, 2014). The second vertical peak and posterior to anterior peak represents the forces acting under the forefoot.

Pressure is the measure of force applied to an area (measured in kPa = 1,000 N/m²), and is often used in clinical biomechanics because tissue damage relates to, not how much force is applied, but to how much tissue is involved in transmitting the force, its area (Richards, 2008). A large magnitude of force may not be damaging over a large area,

i.e. 1000N through the foot and ankle, but if this were to occur beneath a small area, e.g. a metatarsal head, this could then become damaging. Knowledge of how force is distributed beneath specific sites of the foot – how much pressure is applied – is imperative in understanding when skin may be at risk of developing ulcers (in the diabetes literature), or in the context of this thesis, how pressures relate to callus.

There is an underlying assumption that callus is formed as a result of the skin reacting to increased external pressure placed upon it and there is some evidence to suggest this is the case (Bevans and Bowker, 1999). Figure A.6 shows the factors which have been linked to callus development. Extrinsic factors influencing callus development are thought to include (1) footwear and (2) activity levels (Singh, 1996, Freeman, 2002, Grouios, 2004) which might both lead to increased plantar pressures during each step taken and increase the overall load experienced by the skin over a cumulative period. Intrinsic factors include foot deformities causing (3) bony prominences, and (4) those which may be secondary to altered physiology and disease (e.g. diabetes). The intrinsic factors may relate to age, sex and race (Singh, 1996, Freeman, 2002, Grouios, 2004). Below is a summary of how each of these factors is linked to pressure changes beneath the forefoot.



Figure A.6 – Causes of plantar callus

A.6.2 Footwear

Footwear factors that are thought to contribute to callus production include poorly fitting footwear such as wearing shoes that are too small or an incorrect shape for the foot;

irregularities within the footwear such as seams; wearing shoes that are not designed for the purpose for which they are being used; and not wearing footwear (Singh, 1996, Grouios, 2004). As far as the plantar pressure literature is concerned, there is an abundance of papers suggesting that high heeled shoes contribute to increased plantar forefoot pressures. With increasing heel height, the heel is offloaded and the forefoot pressure not only increases but also shifts from lateral to medial (Mandato and Nester, 1999, Speksnijder et al., 2005, Yung-Hui and Wei-Hsien, 2005, Ko et al., 2009, Cong et al., 2011) with in-shoe pressure increases reaching up to 71% (p < 0.05) beneath the first metatarsal with the just the introduction of a 2cm heel compared with a flat sole (Mandato and Nester, 1999). These studies suggest that increased heel height pronates the foot (Cong et al., 2011) and shifts the centre of mass anteriorly and medially (Ko et al., 2009). Interestingly, wearing high heels also significantly increases posterior shear force (Cong et al., 2011) which could increase the risk of skin problems as animal models have reported skin breakdown to occur faster with the introduction of shear force (Goldstein and Sanders, 1998).

Other footwear factors may also contribute to increased pressures. Branthwaite et al. (2013) found that toe box shape can influence pressures on the toes and over the metatarsal heads. The authors found significant differences between round, square and pointed toe boxes beneath all but the second metatarsal head. The percentage differences between conditions ranged from 3.3% (between square and round toe boxes at the third metatarsal head. P = 0.005), to 67.6% (between pointed and square toe boxes at the fifth metatarsal head. P = 0.000). Changes in pressure were attributed to toe box shape not conforming to foot anatomy (Branthwaite et al., 2013).

The presence of studs in football boots contribute to significantly increased pressures over the first, fourth, and fifth metatarsal heads when compared with running shoes (Carl et al., 2014). Percentage differences here ranged from 35.5% (at the fourth and fifth metatarsal head of the non-preferred foot. P < 0.001) to 42.6% (at the first metatarsal head of the preferred foot p = 0.002). Choice of running shoe can also affect peak forces with flat, racing style running shoes increasing peak forces under the lateral forefoot by 16.2% (p value not available) compared with normal cushioned training shoes (Queen et al., 2010). The hardness of the sole can contribute to pressures experienced under the forefoot. Medium and hard shoe soles have been found to lead to significantly increased pressures under the forefoot compared with soft soles (11% and 15.9% differences respectively. P < 0.001 and 0.009 respectively). Differences between medium and hard

soles were noted to be non-significant (4.3% difference. P = 0.114). This cannot be accounted for by contact area, as there was only a 1% decrease in hard compared to soft-soled shoes in this area, and a significant increase (10%. P = 0.001) increase in contract area in the midfoot between these conditions (Lane et al., 2014).

A.6.3 Activity levels

It has also been suggested that increased intensity of exercise, overtraining and excessive loads can increase callus risk (Grouios, 2004). Current evidence suggests that with an increase in activity levels comes an increase in the number of steps taken per day. A recent study with a large sample (n = 1,136) of US citizens has shown significant relationships between the number of steps taken per day with a pedometer and self-reported physical activity. The number of steps per day was shown to increase linearly with the number of days of strenuous exercise per week (p < 0.001). There was also a linear increase in number of steps per day with those who reported themselves to be physically active (p < 0.001) (Bassett et al., 2010). This introduces the point that with exercise comes increased accumulated loads over time which may have an impact on plantar skin.

In addition to increasing the number of steps taken per day, there is also evidence that exercise intensity can increase plantar pressure values. The largest volume of literature available on exercise intensity focusses on walking and running. Burnfield et al. (2007) reported that walking and running generate significantly higher pressures (p < 0.001) under the forefoot than stair climbing (48.6% and 48.2% respectively) and recumbent biking (83.8% and 83.7% respectively) exercises. For both walking and running, speed has been shown to have a significant effect. Burnfield et al. (2004) reported that with increased walking speed from slow (3.42 km/h) to medium (4.8 km/h) and fast (5.82 km/h), peak pressures significantly increased (p < 0.006) in the central and medial forefoot. An increase in speed from slow to medium (40.4% increase) brought an increase in peak pressure of 8.7% and 22.2% in the central and medial metatarsal heads respectively. Increasing the speed from medium to fast (9.1% increase) increased pressures at these areas by 8% and 9.1% respectively. Between slow and fast speeds (70.2% increase), there was an increase in pressure by 17.4% and 33.3% respectively.

Similarly to walking, an increase in running speed from 11.2 km/h to 17.8 km/h (58.9% increase) has been reported to result in a significantly increased peak pressure beneath the whole foot of 15.1% (p < 0.01) in a small population of 11 adolescent runners (Fourchet et al., 2012). This relationship between increased running speed and plantar

pressures has also been observed in 20 female runners in a Taiwanese study (Ho et al., 2010). Increasing jogging speeds from 1.5 m/s to 2.0 m/s to 2.5 m/s resulted in significant increases in peak pressures (p < 0.05) in the lateral, central and medial forefoot (medial forefoot was significant only between 1.5 and 2.5 m/s). From 1.5 m/s to 2.0 m/s (33.3% speed increase), pressures at the lateral, central and medial forefoot increased by 9.6%, 9.3% and 6.2% respectively; from 2.0 m/s to 2.5 m/s (25% speed increase), the percentage increases were 7.9%, 9.0% and 4.7% respectively; and from 1.5 m/s to 2.5 m/s (66.7% speed increase), the percentage increases were 18.3%, 19.1% and 11.2% respectively (Ho et al., 2010).

Other notable observations include a reported increase in peak plantar pressures beneath the forefoot after long-distance running. Bisiaux and Moretto (2008) observed a 10% increase (p < 0.05) beneath the second to fourth metatarsal heads using in-shoe pressure sensors after a 30 minute, intensive run. Nagel et al. (2008) also reported a significant increase in barefoot pressures beneath the second (12.6%), and third to fifth metatarsal heads (16.1%) (p < 0.001). Both studies reported a non-significant decrease in plantar pressure beneath the first metatarsal head and a significant decrease beneath the hallux (p < 0.05 and p < 0.001 respectively). Nagel et al. (2008) also reported a significant decrease in pressure (p < 0.001) beneath the lesser toes which suggests that muscle fatigue resulted in an offloading of pressure from the toes to the metatarsal heads which explains the increased peak pressures in this region. A subsequent study which measured peak forces beneath shod feet on pressure plates, before and after a 20km running race, reported very similar results, noting lateral force distribution during the push-off phase (Willems et al., 2012).

Overall, the studies in this section clearly highlight the relationship between type and intensity of exercise and magnitude of peak plantar pressures. Furthermore, these increased pressures coupled with an increased number of steps per day, will lead to an accumulation of load which may have a profound impact on plantar skin.

A.6.4 Bony prominences

There are various foot deformities which may affect pressure beneath the forefoot. One such deformity is hallux valgus. Plank (1995) and Galica et al. (2013) found that pressure beneath the lateral metatarsal heads were significantly lower in hallux valgus subjects than normal subjects with the latter study reporting a decrease of 4.4% (p < 0.05) compared with controls. Galica et al. (2013) also found a significant increase (4.1%. p < 0.05) in
pressure beneath the lesser toes while a decrease of 3.0% (p < 0.05) beneath the hallux was noted. Bryant et al. (1999), however, found that hallux valgus produced significantly higher peak pressures beneath the first, second and third metatarsal heads than control subjects with percentage differences of 43.8%, 57.6% and 53.4% respectively (p = 0.000 at each site). They also observed a significant increase beneath the lesser toes (35.3%, p =0.008) and a non-significant increase beneath the hallux. Increased mean pressure beneath the first metatarsal head was observed by Martinez-Nova et al. (2010) who observed an increase of 7.9% (p = 0.019). These authors also noted a significant increase in pressure beneath the hallux of 124.2% (p = 0.001). The differences in pressure beneath the hallux observed by Martinez-Nova et al. (2010) and Galica et al. (2013) could be due to the fact that the former authors only included mild cases of hallux valgus whereas the latter authors also included severe cases which may have resulted in offloading as a result of discomfort (Galica et al., 2013). Differing stiffness properties of the joints in the first metatarsal and hallux, and deviation of the hallux could also possibly have contributed to differing pressure profiles. Collectively, these studies show higher forefoot loading medially than laterally and highlight how hallux valgus can change the pattern of pressure distribution in the foot.

While hallux valgus is well publicised, other anatomical conditions are known to affect plantar pressures. Some of these include hallux limitus, which significantly increases pressures beneath the hallux and lesser toes due to increased stiffness at the interphalangeal and metatarsophalangeal joint (Bryant et al., 1999, Zammit et al., 2008); Pes cavus, which significantly increases pressure beneath the metatarsal heads due to medial arch elevation, resulting in reduction in contact area (Burns et al., 2005, Gravante et al., 2005, Fernandez-Seguin et al., 2014); and acute Charcot arthropathy (Armstrong and Lavery, 1998) which is associated with increased pressures beneath the whole forefoot, and is possibly a precursor to destruction of the midfoot bones and joints.

A.6.5 Disease

There are certain diseases which are associated with increased plantar pressures, but by far the most publicised included diabetes and rheumatoid arthritis. Diabetic peripheral neuropathy has been linked with increased forefoot plantar pressures (Fernando et al., 2013) and in turn, high plantar pressures are associated with ulceration (Boulton et al., 1983, Frykberg et al., 1998) making them an important area of discussion.

In the diabetes literature, the two major factors affecting pressures includes tissue thickness and stiffness; and a reduction in joint mobility. Mueller et al. (2003) found that the greatest predictor of increased pressure beneath the forefoot in individuals with diabetes was hammertoe deformity, but they also reported soft tissue thickness to be important. Hammer toe deformities have been associated with increased peak plantar pressures in diabetes, with peak pressures reported to be 71.1% (p < 0.001) higher in the forefoot (Bus et al., 2005). The significance of the pressure increase also relates to the level of deformity with a significant negative correlation between toe angle and peak pressure (r = -0.74) (Bus et al., 2005). The elevated pressures are likely to be due to displacement and thinning of the fat pad beneath the metatarsal heads due to the hyperextension of the metatarsophalangeal joint (Bus et al., 2004a). Plantar soft tissue thickness beneath the metatarsal heads has been found to have an inverse relationship with peak plantar pressures (r ranges from 0.430 to 0.605 for second to fifth metatarsal heads. P < 0.0001) meaning that in cases where the tissues are thinner, the peak pressures tend to be higher (Young et al., 1995, Abouaesha et al., 2001). This may be due to thinner tissues which stiffen faster upon loading, causing the peak pressures to increase.

One other area relating to tissue stiffness, independent from the plantar fat pad and soft tissues, is Achilles tendinopathy, a common problem in diabetes resulting in a shortened, stiffened Achilles tendon, caused by dense packing of collagen fibrils with altered morphology in its tissue (Grant et al., 1997). This results in ankle equinus, a severe reduction in dosiflexion of the ankle joint which has been reported to significantly increase pressures by 7.6% (p = 0.007) in the forefoot, probably due to the foot being pulled into plantar flexion (Lavery et al., 2002). Additionally supporting this theory, Achilles tendon lengthening surgery has been shown to promote healing in forefoot ulceration (Holstein et al., 2004) due to a reduction in peak pressures at the forefoot by 26.7% (p < 0.001) (Armstrong et al., 1999).

Limited joint mobility, particularly of the subtalar and first metatarsophalangeal joints are also associated with high plantar pressure and ulceration in diabetes. Increases in peak pressure of up to 121.1% (p < 0.001) between individuals with no forefoot problems and those with limited subtalar joint mobility have been reported, and this may further increase up to 152.6% (p < 0.001) with the presence of neuropathy (Fernando et al., 1991). A more recent study conducted by Viswanathan et al. (2003) found individuals with neuropathy had significantly impaired joint mobility at the subtalar and first metatarsophalangeal joint (p < 0.001), resulting in increases in forefoot peak plantar

pressures compared with those without neuropathy (30.8%. p < 0.001). The reduction in joint mobility was further reduced (p < 0.001) and peak pressures dramatically increased by 58.0% in those with a history of ulcers, compared to individuals without neuropathy (Viswanathan et al., 2003). The pressure increases are probably due to the increased stiffness of the joint which results in less of the load being dissipated (Mueller et al., 1989). Increases in thickness of the plantar fascia beneath the whole foot has been shown to be inversely correlated to metatarsophalangeal joint mobility (r = -0.53) and directly related to increased vertical forces beneath the metatarsal heads (r = 0.52), probably due to making the foot more rigid (D'Ambrogi et al., 2003). Stiffness of plantar soft tissues in the forefoot has been found to be significantly increased in individuals with diabetic peripheral neuropathy which could lead to increased pressures (Klaesner et al., 2002, Sun et al., 2011).

The volume of literature surrounding rheumatoid arthritis and forefoot pressures is less than that of diabetes. The major foot deformities associated with rheumatoid arthritis include hallux valgus, metatarsal head depression, hammer or claw toes, tendocalcaneal bursitis or subplantar spur formation (Dimonte and Light, 1983) plus reduced range of motion of the ankle and subtalar joints (Locke et al., 1986). Peak forces under the hallux, lateral three toes and first metatarsal head in severe rheumatoid arthritis, cases with heel valgus and reduced joint motion, are significantly decreased (p < 0.05) with reductions ranging from 28.3% (first metatarsal) to 100% (second toe), compared to a normal population (Sharma et al., 1979). A subsequent study found similar results with 37.2% force reduction beneath the toes and 16.1% reduction beneath the first and second metatarsals (Simkin, 1981). Both studies found no significant difference in the loading of the lateral metatarsal heads, indeed the former study reported an 18.6% increase. The authors of both studies suggest discomfort and changes in the midtarsal joints results in loads being shifted laterally (Sharma et al., 1979, Simkin, 1981). Woodburn and Helliwell (1996), however, reported a significant decrease in pressure over the lateral metatarsal heads in rheumatoid patients with a valgus heel compared with normal rheumatoid and control groups (pressure reductions ranging from 28.1% to 52.1%. p = 0.001), probably as a result of medial distribution of pressures as a result of the valgus (everted) rearfoot. Tastekin et al. (2009) found no significant difference in dynamic pressure between heel valgus and normal rheumatoid patients, possibly because of pain interfering with gait.

In a study comparing rheumatoid arthritis patients with a healthy population, Otter et al. (2004) found that there were no significant differences in forefoot pressures, but the

rheumatoid group did show a significant increase in pressure time integral and decrease in force-time integral compared with the control group. The authors were unsure of the mechanism for this but suggested that patients may have had a slower gait cycle than the controls, possibly due to avoiding pain (Otter et al., 2004). There is also evidence of pressures being related to damage to the metatarsal joints. One study observed that joint damage had a linear relationship with peak plantar pressures (van der Leeden et al., 2006).

This section has shown that diseases, particularly diabetes and rheumatoid arthritis may have a profound effect on plantar pressures. These diseases have multi-tissue effects: pathological changes may result in tendons and soft tissues becoming stiffer (Grant et al., 1997), stiff tendons may cause joint deformity, causing hyperextension and as a result thinning plantar soft tissues by stretching them (Bus et al., 2004a). Thickened plantar fascia, which is related to altered joint function, may affect how load is distributed beneath the foot (D'Ambrogi et al., 2003). As a result of deformities and associated pain, individuals may reduce walking speed and load different sites of the foot, which may in turn also cause problems (Sharma et al., 1979). These factors all have a significant influence on pressures beneath the foot.

A.6.6 Plantar pressure and callus

There are few studies in the literature which demonstrate the hypothesised link between foot biomechanics and callus, but three of particular relevance do exist (Bevans and Bowker, 1999, Abouaesha et al., 2001, Menz et al., 2007). Bevans and Bowker (1999) compared lower limb biomechanical characteristics of three groups of individuals. The first (n = 22) contained subjects with diabetes presenting with forefoot callus; the second (n = 20) contained subjects with diabetes without plantar callus, and group 3 (n = 17)contained subjects free from diabetes presenting with forefoot callus. They found that both groups with callus (including individuals with diabetes, and individuals without) presented with a significantly higher total of structural anomalies than the group without callus (96 and 81 versus 65 for groups 1 and 3 versus group 2 respectively). The anomalies more prevalent in the callus groups included forefoot invertus, forefoot evertus, and equinus. These are static structural deformities that reflect differences in the alignment of the forefoot to the rearfoot, and limited ankle dorsiflexion. In the group without callus, the mean position of ankle dorsiflexion with the knee extended and the mean relaxed and maximally dorsiflexed hallux positions were significantly greater, up to 3.0% (p = 0.03), 16.5% (p = 0.013) and 12.8% (p = 0.015) for each parameter respectively; and the mean eversion of the subtalar joint was significantly less than the two callus groups with angle reductions reaching 57.9% (p = 0.001). The callus group without diabetes had a significantly greater angle of rearfoot varus than the two groups (callus and non-callus) with diabetes. The angle of ankle dorsiflexion with the knee bent at 90°, the range of motion of the first metatarsal joint and the angle of the subtalar joint in the relaxed stance position were similar in all three groups. From a discriminant function analysis, they found that the strongest predictor of callus development was subtalar eversion (or pronation) (p = < 0.001), but the decreased dorsiflexion with knee extension was also a predictor. No information on the position of calluses on the forefoot was given, so it is unclear whether certain types of biomechanical problems related to formation of calluses in certain areas, such as whether eversion might result in medial forefoot callus.

The significantly limited range of ankle dorsiflexion in the callus groups is likely to lead to increased plantar pressure due to compensatory changes in gait. The significantly limited range of hallux dorsiflexion in relaxed and maximally dorsiflexed positions in the callus groups may increase loading beneath the interphalangeal joint (Bevans and Bowker, 1999) and perhaps affect the way load is dissipated during gait, as with individuals with limited joint motion in diabetes (Mueller et al., 1989). The presence of diabetes seems not to be a factor because there were no significant differences in these parameters between the diabetes group with callus and the non-diabetes group with callus, while differences were shown between the callus and non-callused group (who had diabetes). Furthermore, the diabetes group with callus did not have a significant difference in blood levels of glycated haemoglobin compared to the diabetes group without callus which suggests that the differences in joint angle data is biomechanical, not biochemical in cause (Bevans and Bowker, 1999). Linking these results to the previous subsections, increased pressure could result from limited mobility of the subtalar and ankle joints (Mueller et al., 1989, Fernando et al., 1991, Viswanathan et al., 2003) particularly in individuals with diabetes where insensitivity caused by neuropathy is present.

Abouaesha et al. (2001) assessed plantar pressure, skin thickness and callus in individuals with diabetes (n = 157). They found that subjects with forefoot callus had a significantly increased pressure under each metatarsal head with pressure increases ranging from 21.7% (left foot, fourth metatarsal) to 51.8% (right foot, third metatarsal) compared to control subjects (p < 0.05). They also found these subjects had significantly reduced plantar tissue thicknesses under the second to fifth metatarsal heads (p < 0.05). This study and others discussed previously have also highlighted the inverse relationship

between tissue thickness and plantar pressures, particularly in cases where the metatarsophalangeal joint is hyperextended, causing displacement and thinning of plantar soft tissues (Young et al., 1995, Bus et al., 2004a). These may stiffen faster upon loading, causing increased pressure.

Menz et al. (2007) recruited 292 participants from a retirement home and screened them for callosities, which affected 52% of their sample (n = 151). Using a MatScan[®] pressure mat, they assessed barefoot walking and found that significantly higher plantar pressures existed under calluses overlying the hallux (13.8% increase; p = 0.007), second metatarsal head (10.4% increase; p = 0.001) and third to fifth metatarsal heads (14%) increase; p = 0.009), but no significant differences beneath the first metatarsal head or lesser toes. From foot deformity assessments, they found that calluses were most likely to occur in females with moderate to severe hallux valgus, toe deformities and foot pain; with hallux valgus and toe deformities being significantly associated with callus (p < 0.001 and 0.016 respectively). This data relates to the previous discussed studies in hallux valgus and hammer toe deformities which have reported significantly increased pressures beneath the metatarsals (Bryant et al., 1999, Bus et al., 2005, Martinez-Nova et al., 2010). This study is very important in assessing the link between bony deformities, which have been shown to be associated with increased plantar pressures, and the presence of callus. It could be argued that callus builds in areas where there are increased pressures as a result of bony deformities, but this is not clear cut in the literature. The studies by Bevans and Bowker (1999), Abouaesha et al. (2001) and Menz et al. (2007) present compelling evidence that suggests structural abnormalities such as toe deformities, which may result in stretched, thinner soft tissues (Bus et al., 2004a, Abouaesha et al., 2001), or compensatory changes in gait (Bevans and Bowker, 1999), lead to increased peak plantar pressures, a hypothesised precursor to callus development. However, evidence supporting the contrary – callus being a cause, and not a result, of increased pressures, has also been presented in the literature.

Table A.5, shows several other studies have reported significantly higher (p < 0.05) peak plantar pressures under feet with callus compared with control groups (Potter and Potter, 2000b, Pataky et al., 2002, Duffin et al., 2003). However, it is unclear from the literature whether increased pressure beneath callus is a predisposing factor or a result of changes in skin thickness and characteristics. There appears to be conflicting evidence in the literature assessing the effect of callus debridement on pressure (Table A.6). For example, Pitei et al. (1999) and Pataky et al. (2002), both found that removal of callus in patients with diabetes significantly reduced plantar pressure (p < 0.014 and 0.001

respectively) which might indicate that callus causes elevated pressures. However, Potter and Potter (2000b) (pressures not shown in paper) and Woodburn et al. (2000) found no significant difference after removal of callus which might suggest that increased plantar pressure was not the outcome of the callus but rather a feature of the callus site prior to its development. In contrast to the studies investigating callus debridement in diabetes, Woodburn et al. (2000) reported a 17% mean increase (p > 0.05) in peak pressure after callus debridement. This was explained by the fact that there was a decrease in contact area over the metatarsal heads after callus removal (thus increasing pressures) and a statistically significant reduction in pain (p = 0.01) might have increased walking speed and pressures (based on evidence of decreased contact time post debridement). In a later study on rheumatoid arthritis patients (n = 18), Davys et al. (2005) found a non-significant decrease in pressure after callus debridement which would again suggest that callus debridement in this group of patients, while possibly beneficial for pain relief, does not impact on pressure reduction and that callus is not the cause of elevated plantar pressures, rather increased pressure is a pre-existing feature of the site.

	Potter and Potter (2000b)	Pataky et al. (2002)	Duffin et al. (2003)	
	Healthy	With diabetes	With diabetes	Healthy
Callus	472.9 kPa	314 kPa	43 N/cm ²	46.5 N/cm ²
Control	355.1 kPa	128 kPa	39 N/cm ²	39 N/cm ²
% increase	33.17	145.31	10.26	19.23

 Table A.5 Other studies reporting significant pressure increases beneath calluses

Pressure changes calculated by ((diabetes group – control group) / control group) x 100.

	Pitei et al. (1999)					
	Group	Group	Group	Woodburn et	Pataky et al.	Davys et al.
	0	А	В	al. (2000)	(2002)	(2005)
Before						
debridement	374.8	351.7	241	241	340	828
After debridement	251*	240.5*	176.2*	285	141*	817
% change	-33.0	-31.6	-26.9	18.26	-58.53	-1.33

 Table A.6. Peak pressure changes following callus debridement (kPa)

* = Significant change in pressure. Pressure changes calculated by ((post treatment – pre treatment) / pre treatment) x 100. Pitei et al. (1999): Group 0 - patients presenting with callus for first time, with no history of ulceration; Group A – history of ulceration, requiring callus debridement every 6 - 8 weeks; Group B – history of ulceration, requiring callus debridement every 3 - 4 weeks.

One possible factor causing the different outcomes in these studies might be differences in inclusion criteria. Pitei et al. (1999) treated patients with diabetes with neuropathy and a history of ulcers (n = 24), Pataky et al. (2002) treated 13 patients with diabetes without neuropathy, any skin problems or bony prominences (n = 13), Potter and Potter (2000b) recruited only healthy individuals without the presence of disease (n = 15; total of 36 calluses) who had callus removed every two to three months. They also reported not including subjects with bony abnormalities, those who had never had callus removed previously, or had regular callus removals frequently (every three to four weeks). Woodburn et al. (2000) recruited rheumatoid arthritis patients (n = 14). The results presented by Pitei et al. (1999) are particularly interesting because individuals with neuropathy and a history of ulceration are significantly more prone to biomechanical problems than those without a history of ulceration (Fernando et al., 1991, Viswanathan et al., 2003). The fact that callus removal resulted in comparable reductions in pressures in their two groups with a history of ulceration and the group with no history of ulceration, suggests that if biomechanical problems and increased pressures were a factor in their ulceration, they were not a factor in their callus development. The pressures beneath the calluses would remain relatively unchanged. In these four studies, it could be argued that the presence of diabetes is a factor contributing to the results, but neuropathy and structural deformities associated with the condition are not factors due to the fact that Pataky et al. (2002) did not recruit subjects with these characteristics.

The differing pressure measurement systems may have also played some part. Pitei et al. (1999) used an F-Scan insole and Pataky et al. (2002) used Force Sensing Resistors placed under each metatarsal head and these could have been more sensitive to changes in pressure after callus removal compared with the barefoot pressure mats employed by Potter and Potter (2000b), Woodburn et al. (2000). Davys et al. (2005) also used a pressure mat in their study. The variances in the participants of each study in relation to underlying disease pathology make it difficult to pool the results. As discussed previously, diabetes and rheumatoid arthritis have different features when compared to feet without underlying disease. Alterations in gait due to neuropathy and the anatomical changes discussed previously may have played a part in these different results.

Overall, the literature seems to point towards anatomical variations contributing to increased pressures leading to callus development, but this is not absolute because there are studies which have found that debriding callus also reduces the pressures (Pitei et al., 1999, Pataky et al., 2002) which would suggest that the presence of callus increases

pressure too. The only way to determine whether pressure is a cause of callus would be to conduct studies on normal plantar skin to determine whether pressure causes these types of skin changes. Either way, pressure studies are limited in the fact that the pressure measures only the compression element of the pressure which acts perpendicular to the foot – sensor interface. This does not provide any information about horizontal shear which may also have an impact on callus development.

A.7 Plantar shear pressure

Plantar pressure data in the literature, refers to forces applied perpendicular to the load sensor surface, and this ignores forces applied in medio-lateral and antero-posterior shear directions. Compared with plantar pressure studies that report vertical pressures, the volume of literature addressing specifically plantar shear pressure is scant; therefore how the anterior and posterior ground reaction shear forces are distributed across the forefoot as shear pressure is not clear. It follows that the role of shear in callus development, or increases in shear forces with callus, are poorly understood. However, there are some helpful papers which provide some insight into this area. In a sample of healthy males (n = 10) walking in conventional leather shoes, Pollard et al. (1983) found, using individual sensors placed under the metatarsal heads, hallux and heel, that the highest plantar shear force (values not available) occurred at the metatarsal heads; an anterior force occurred over the central and medial heads and a posterior force occurred over the lateral heads. In barefoot walking, the force was increased at all sites except the hallux. During propulsion, the gross forces applied to the forefoot are in the same direction so it is an interesting observation that shear occurred in two directions. They also found in a study of patients with diabetic neuropathy (n = 6), using the same instrumentation, that foot ulceration occurs at the site of maximum shear, the same site as maximum pressure (values not available) and this occurred with both barefoot and shod feet (Pollard and Le Quesne, 1983). Subsequently, the same research group (Tappin and Robertson, 1991) found in a sample of healthy individuals (n = 20) that the peak compression and peak shear forces in the first, fourth and fifth metatarsal heads occurred at the same time, which may have an effect of occluding skin perfusion. These authors used a semiconductor field coil with a centre tap placed on a 16mm diameter plate and a magnet placed on a second plate. The two plates were connected via silicon rubber allowing displacement to be measured (Pollard et al., 1983). However, the system only allowed unidirectional measurement in the

orientation of the grooves within the plates, so this could have led to some discrepancies in force magnitudes observed. It may have caused some medio-lateral shear to be recorded as antero-posterior shear as the mechanics of the sensor could have forced movement in an antero-posterior direction. The force tracings in these studies are graphically represented (in kg) but no numerical data is presented so the compression/shear ratio cannot be calculated.

Hosein and Lord (2000), using a similar system to Pollard et al. (1983) found in healthy individuals (n = 8) that peak plantar shear pressures occurred more laterally, under the third and fourth metatarsal heads (86.5 kPa and 71 kPa respectively), than peak vertical pressure, which occurred under the second and third heads (both 228 kPa). The compression to shear ratios (in kPa) were 5.8, 7.4, 2.6, and 2.1 per one kPa of shear stress under the first four metatarsal heads respectively. This range of ratios, if true, would suggest that shear pressures are not easily predicted or inferred from vertical pressure data. The same authors subsequently found that in cases of diabetes (n = 6), the peak shear pressures occurred more medially (72.7 kPa under the first metatarsal head) and peak compression pressure more medial (273 kPa under the second head) than the healthy group they reported in the previous study (Hosein and Lord, 2000). The medial metatarsal heads were found to be the most common area for ulceration corresponding to the site of peak shear pressure (Lord and Hosein, 2000). Like the previous authors, Perry et al. (2002) found in 12 individuals with diabetic neuropathy that peak compression pressure occurred in the medial metatarsal heads (189 kPa) while peak shear occurred in the lateral metatarsal heads (33 kPa). Furthermore, when the authors analysed the combined effects of compression and shear, they found that these forces occurred at the same site in 50% of individuals, but occurred at different times. Peak compression occurred at an average of 0.186s before shear in these individuals, but it is unclear what percentage of stance phase this was (Perry et al., 2002). Similarly, Yavuz et al. (2007) found that in 60% of their sample of patients with diabetic neuropathy (n = 10) that the sites of peak compression and shear pressures were different. Subsequently they observed that generally the magnitude of peak shear pressure was significantly greater at 83.3 kPa vs 62.3 kPa (33.7% difference; p = 0.014) in patients with diabetes (n = 15) than normal control subjects (n = 20) (Yavuz et al., 2008). The peak compression pressures were also higher in the diabetes group at 614.2 kPa vs 497.5 kPa (23.1% difference) but not significantly. However, neither study disclosed the specific locations of these shear forces in detail.

One study (Stucke et al., 2012) has specifically investigated whether the differences in the location of peak compression and shear pressures could be due to the movement of tissue during loading. The authors studied the gait of 11 healthy individuals, and from their data the forefoot peak pressure ratio was 1 kPa of shear for 14 kPa compression. They reported that compression and shear peak pressures occurred in different locations and at different times with peak compression occurring before peak antero-posterior shear in 57.6% of cases. They also observed, in 67% of cases, that the peak compression pressure was located between two peak shear pressure points. They concluded that 'since the peak pressure is located directly in between both of the peak shear values and the forces are moving in opposite directions about the peak pressure location radial ''spreading'' occurs' (Stucke et al., 2012 p.621). Perry et al. (2002) observed similar results in tissue activity where they found that skin was stretched to a greater degree than it was bunched in the forefoot.

One study has investigated shearing forces present under callused regions of the foot (Mori et al., 2012). These authors measured plantar shear force in patients with diabetes with (n = 9) and without (n = 41) forefoot callus. After normalising the data to bodyweight, they found a significant increase (31.6%; p = 0.03) in peak shear force in subjects with callus which, like the pressure studies presented in section A.6.6, may have preceded or succeeded callus development. The presence of callus may have increased the shear value as there is more tissue available to be deformed with the introduction of horizontal forces.

The varying results across the literature may be attributed to the fact that measurement of shear forces is technically challenging and each research group used different instrumentation and data collection techniques. Pollard et al. (1983) and Tappin and Robertson (1991) measured shear and compression forces in barefoot walking with separate compression and shear transducers. As their transducers were attached to the plantar skin and were 2.3mm thick, this could have caused increased loading and affected the loading at the foot/floor interface, and thus their measures of compression and shear could be directly influenced by the presence of the sensor. The device used by Lord and Hosein (2000) and Hosein and Lord (2000) was an in-shoe measurement device that could measure antero-posterior and medio-lateral shear pressures in separate gait cycles but they did not measure this at the same time as compression pressure (which was measured in a separate gait cycle using F-Scan insoles). Perry et al. (2002), Yavuz et al. (2007), Yavuz et al. (2008) and Stucke et al. (2012) used platforms designed to simultaneously collect shear

and compression data in barefoot walking. Perry et al. (2002) only analysed data collected from the initiation stage of gait and used the two-step method, whereas Mori et al. (2012) used an insole capable of measuring shear force and compression pressure at the same time. While all these studies are technically different, the results give a good insight into likely magnitudes, compression to shear ratio, and time of peak compression and shear forces.

In reality, peak shear pressure to compression pressure ratios vary greatly across the studies, with pressure ranging from 2.7 to 14 times greater than shear (mean across studies is 6.9 times greater). The far greater magnitude of compression than shear pressure concurs with ground reaction forces, which tend to be six times greater in the vertical direction than in the horizontal direction (Richards, 2008). Additionally, peak compression and shear pressures tend to occur at different sites. In the callus literature, calluses have been noted to occur at the areas of peak compression pressure in the forefoot (Pitei et al., 1999, Potter and Potter, 2000b, Pataky et al., 2002). Because callus and peak compression pressure are linked, it is pertinent that this area be explored in greater depth. However, in order to do so, one must understand how the skin reacts when forces are applied to it.

A.8 How the epidermis adapts to external forces

While the evidence linking external loading to plantar callus development is scant, some authors have investigated the effects of external insults on human skin structure in-vivo. Several have looked at the skin's response to frictional stimulus. The first of these, Rubin (1949), tested the effect of applying shear and compression by rubbing anterior thigh skin for 10 minutes with a tongue depressor and lubricant on 5 adults daily for 30 days. Biopsies showed that the thickness of the stratum corneum was increased by an average of 36% compared to the control site (the corresponding area on the opposite leg). Another study by Goldblum and Piper (1954) used a scratching machine to administer tolerable scratches (with 75 grams of weight) using a lucite 'fingernail' (1/8 inch diameter) for one hour a day on the backs of four male patients. Biopsies of the intervention skin showed evidence of hyperkeratosis which was not present in control skin samples (thickness not mentioned). However, each of these subjects had a history of inflammatory skin disorders which may have made them more prone to skin thickening.

While there is evidence of the skin's response to shear, even small load stimuli have been shown to elicit a response. Pinkus (1952) and Brophy and Lobitz (1959)

performed tape stripping experiments on volar skin on the forearm of a 45 year old male and skin on the back of a 34 year old male respectively. Pinkus (1952) tape stripped the skin repetitively until the stratum corneum was removed (the skin appeared red, shiny and dry). Brophy and Lobitz (1959) tape stripped 5 skin sites on the subject's back, and then repeated stripping in intervals of four, eight, 12, 24 and 48 hours later at each respective skin site. Biopsies were then taken at five minutes, four, eight, 11, 16, 25, 28, 32, 39, 49 and 73 hours after injury at each skin site. Pinkus (1952) found an increase in diameters of the most superficial, medium and deep layers of the epidermis peaked and plateaued between 48 and 72 hours where increases of 69.1%, 93.8% and 103.5% respectively, at 72 the hour timepoint. Brophy and Lobitz (1959) observed that cell mitotic activity increased after the first tape strip, but after subsequent strips at each site, the mitotic activity mimicked that of the skin samples taken from the first strip. This suggests that the skin showed a greater response to the first tape strip, and that repeated strips did not alter the skin's inflammatory response to the original strip (Brophy and Lobitz, 1959). These studies begin to show how mild skin irritation is enough to drive physiological changes.

In-vivo animal studies (Mackenzie, 1974a, Mackenzie, 1974b, Goldstein and Sanders, 1998, Sanders et al., 2002) have further shown how skin adapts to friction, shear and compression. The main theme which has emerged is that under moderate levels of mechanical load, the skin mitotic levels increase and epidermal thickening occurs, while under a more severe mechanical stress, ulceration also occurs.

Mackenzie (1974a) and Mackenzie (1974b) studied the effect of friction on the epidermis of mouse ears. They administered friction using 10 circulations of a rotating brush, which rotated at a speed of 40 - 50rpm with a force of 8 - 9 grams (0.08 - 0.09 N), to three groups of mice ranging from one friction application to daily application for 35 days. A subsequent group received a greater number of circulations (x30) from the rotating brush between one and 10 days. It was found that mitotic activity initially decreased within the first 24 hours after a single friction application, but increased again to reach a peak at 48 hours, a similar finding to Pinkus (1952). In the mice who received daily friction application period. For mice euthanized within 24 hours of the final daily friction application for seven, 14, 28 and 35 days of friction application, mitosis was 261.0%, 462.8%, 278.4% and 183.6% times higher than controls respectively (Mackenzie, 1974b). Furthermore, it was observed that the ear epidermis of mice that had received friction for seven days or longer contained between 49.0 - 82.3% more cells (p < 0.05) and the

stratum corneum was reportedly thicker than controls (measurements not available) (Figure A.7). There was no significant difference noted in thickness between the epidermis specimens that had received friction for seven days or longer which means that this hypertrophy must have occurred within the first seven days. In the group (n = 10) receiving 30 circulations, ulceration was noted and the skin thickening response was generally much greater than the group receiving less friction. In all cases, the stratum corneum was thickened and the strata basale and spinosum were increased in size, likely due to increased demand for cell production (Mackenzie, 1974a, Mackenzie, 1974b). The results of these studies suggest that 35 days is ample time for the skin to show a sustained response to a small mechanical stimulus at a force of 0.08 - 0.09 Newtons.



Figure A.7 – Specimens of mouse ear epidermis A = control; B = specimen after 7 days of daily friction application. Scale is 50 µm (Mackenzie, 1974a).

Goldstein and Sanders (1998), applied different combinations of compression and shear to the thigh skin of infant pigs (n = 8) using a load application device. The aims of the experiment were to test the effects of different combinations of compression and shear force application in an acute manner, whereby high loads were applied at 10 minute

intervals over 40 minutes on just one occasion (magnitudes not disclosed); and chronic manner, whereby the loads were applied for 40 minutes per day for three weeks. The 'chronic' load combinations included 5N compression with 2N shear, 6N compression with 1N shear and 3N compression with 1N shear; the magnitudes were lower than those used in the 'acute' experiments. Following the load application period the skin was histologically studied. The 'acute' load group showed that increased shear force magnitudes resulted in faster breakdown of the skin with abrasion and blister formation occurring. On skin where less shear force magnitude was applied, only skin redness was evident. For the chronic group, no significant changes were noted but there was some increase of epidermal volume and decrease in dermal volume which might suggest the beginning of a callus-like skin response. Again, this study demonstrates how skin may begin to change after several weeks of regular, moderate load application.

The dermis' resistance to mechanical stress is reportedly due to the morphology of collagen fibrils which has previously been linked to mechanical load bearing in evolutionary literature (reviewed by Sanders et al., 1995). Excessively high pressures over a prolonged period of time will lead to tissue breakdown and this is accelerated by increased age, smoking habits and moisture of the skin (Sanders et al., 1995). With regards to adaptation, collagen fibril diameters have been found to increase in tendon and skin as a result of increased compressive and shear forces. This in turn allows greater force tolerance (reviewed by Wang and Sanders, 2003). Thus external forces may lead to structural changes in the dermis as well as the epidermis.

In summary, there is a good body of evidence linking changes in skin structure to external forces placed upon it. The main theme which has emerged is that skin, in areas not designed for being loaded, adapts to mechanical loads up to a certain threshold of acute or accumulative load application, which if exceeded, is likely to cause skin breakdown and subsequent ulceration. Adaptation seems to occur under relatively small loads applied over a short period of time i.e. above seven days (Mackenzie, 1974b). It is a plausible but untested hypothesis that prolonged moderate loads under the metatarsal heads will alter chemical and physical properties of plantar skin and lead to hyperkeratosis. As the above studies have shown, increased loads may cause a skin response in as short a period as seven days. The separate contributions of compression and shear to any skin response are not known.

A.9 Gaps in the research into callus

Firstly, it is important to draw attention to the fact that corns have very seldom been studied, and barely feature in the literature. While callus and corns may often be considered together clinically, there is a lack of evidence to suggest they are similar in their biochemistry or in their aetiology. The vast majority of the literature published to date has focussed on callus; therefore callus will be the subject of the remainder of this thesis. Table A.7 shows the important gaps in the literature. Perhaps the most important omission from the literature is how calluses differ from normal plantar skin in terms of their form and functional properties. Whilst there is some understanding of how calluses develop and their prevalence, there are no studies investigating the biophysical characteristics of these lesions. In fact, there is little information which outlines the characteristics of normal plantar skin (such as profiles of hydration and mechanical properties) to provide a reference point against which abnormal skin can be measured. It is important to understand how normal and callused plantar skin differs because these differences could provide important insights into aetiology and treatments. Therefore, quantifying the characteristics of normal and callused plantar skin is an important first step.

Research gap	Importance
There are no studies which investigate the	It is important to understand how normal and callused plantar
biophysical properties of callus	skin differs because this could provide information into
There are no studies which investigate the	aetiology, inform treatment approaches and improve the
biophysical properties of normal plantar	investigation of treatment efficacy.
skin	
It is not known how normal and callused	
plantar skin differ in terms of biophysical	
properties	
The precise relationship between external	It is important to understand what mechanical factors cause
loading and plantar callus development has	hyperkeratosis in plantar skin because clinicians can then use
never been scientifically investigated.	this information to aid in callus prevention and treatment,
	particularly in at-risk groups such as rheumatoid arthritis and
	diabetes.

Table A.7 - Gaps in the research into callus

Characterising plantar callus first requires that suitable tools are available for characterising foot skin. Instruments for characterising skin are available but these have not previously been used on the foot. The thick epidermis of plantar skin and hyperkeratotic lesions which make the skin particularly hard may make measurement difficult as the devices available are primarily intended for use on soft skin, such as that on the face. Therefore, it is important to ensure that these devices will give reliable measurements on plantar skin before attempting to use them for characterisation of normal and abnormal foot skin, such as callus.

There is also a possible link between callus development and external mechanical loads. It has been shown in the literature that high peak plantar pressures are associated with callused regions on the foot. Furthermore, it has been shown on other areas of the body and in animal studies that hyperkeratosis occurs as a result of compression and shear pressures which mean that it is highly likely that calluses develop as a result of these pressures. These studies highlight an area of importance; skin taken outside its physiological range of load tolerance, in terms of both magnitude and duration of load, causes it to change. Plantar pressure studies tend to only state the magnitude of pressure beneath the callused skin and do not express the accumulative load beneath this skin site. This is an important omission from the literature – increased duration of increased pressure will also have an accumulative effect, so to fully understand the relationship between pressure and callus, this needs to be taken into account. Further investigation into understanding the relationship between pressure and callus is essential for prevention and treatment of these lesions.

Part B. Skin measurement devices

B.1 Introduction to skin measurement parameters

In the context of the gaps in the literature, it is important to characterise the biophysical properties of plantar skin and how these properties are affected in cases of callus. The previous section highlighted that mechanical properties, trans-epidermal water loss (TEWL) and hydration are interrelated properties and may be useful measurement parameters to characterise plantar callus. The structural changes which occur in callus would most likely affect these properties and it is thus hypothesised that a relationship exists between callus development and regression, and skin properties.

However, there are other potentially useful parameters. The skin's surface topography could be useful as there is a noticeable change in appearance from normal to callused skin. The relationship between topography and biophysical properties would be useful in characterising the skin, especially in the context of clinical evaluations of skin, where observation may be the only tool available. Cross sectional anatomy and thickness measurement of callus would also be useful for similar reasons but would also potentially allow analysis of how the lesions affect structures deep to the epidermis, such as the dermis and subcutaneous tissues. These measures may also assist the evaluation of intervention efficacy and also provide patients with a visual indication of improvement.

To investigate the biophysical properties of plantar skin, non-invasive approaches are preferred. Invasive biopsies have the benefit of being able to visualise and biochemically test the tissues but pose safety and ethical issues. They are far less practical when trying to characterise changing properties of lesions in-vivo, as repeated biopsies would need to be taken from the same skin sites over the test period. It is also important that techniques used will be able to characterise differences in both normal and callused skin, and therefore have sufficient sensitivity to changes in skin. It is important to note that while devices might prove reliable on some skin sites, because plantar skin is different in anatomy, it is likely to have different biophysical properties than non-plantar skin sites. Therefore reliability must be assessed separately on plantar skin to ensure that any devices used are fit for purpose in this area.

This section of the literature review will explore the different measurement modalities which can be used to measure the skin properties highlighted above including skin surface hydration, skin mechanical properties, skin imaging and skin surface analysis. Table B.1 shows details of the devices of interest and alternative modalities which will be covered in the following section.

Device name	Skin property	Measurement	Validated/	Alternative
	measured	principle (units)	on plantar	measurement
			skin?	modalities
Corneometer [®] **	Hydration.	Skin surface	Yes / no	Conductance, NMR
		capacitance (aµ).		spectrometry, TTT.
Tewameter [®] *	Skin barrier	TEWL $(g/h/m^2)$.	Yes / no	None.
	function.			
Cutometer [®] *	Mechanical	Negative pressure	Yes/ yes	Torsion ballistometry,
	properties:	application (mm).		indentation, ultrasound,
	viscoelastic and			digital image speckle
	stiffness parameters			correlation (direct
	under negative			measures), RRT
	pressure.			(surrogate measures)
Reviscometer [®]	Mechanical	RRT (aµ).	Yes / no	Suction cup method,
**	properties: direction			torsion ballistometry,
	and density of			indentation, ultrasound,
	collagen and elastin			digital image speckle
	fibres within the			correlation (all direct
D' (skin.		X 7 /	measures).
Diagnostic	Cross sectional	Reflection and	Yes / no	High frequency
ultrasound (low	anatomy/ deptn.	scattering of		diagnostic ultrasound,
frequency).		ultrasonic waves (B-		OC1, MRI.
Dormatoscopo	Skin surface	Direct imaging of	No / no	None
Dermatoscope	topography: visual	skin surface	10/10	None.
	skin surface	Skill Sulface.		
	imaging			
Visioscan®	Skin surface	IIV light emitting	Yes/no	None
(ISTOSCUIT	topography: UV	camera and	1057110	i tone.
	skin surface imaging	specialised software		
	with measures of	analysing pixel and		
	texture parameters.	grey levels (aµ).		
Visioline®	Skin surface	Shadow profilometry.	No / no	Quantimet (automated
	profilometry:	I ST		replica measurement
	negative replica			system).
	analysis of wrinkles.			

 Table B.1 – List of measurement devices and alternative measurement modalities

*Direct measure of skin property. **Surrogate measure of skin property (for non-imaging devices).

NMR – Nuclear magnetic resonance; TTT – Transient thermal transfer; TEWL – Trans-epidermal water loss; RRT – Resonance running time; OCT – Optical coherence tomography; MRI – Magnetic resonance imaging.

B.2 Skin surface hydration

The Corneometer[®] (Courage and Khazaka, Köln, Germany) (Table B.1) measures the stratum corneum hydration based on the principle of capacitance. The Corneometer[®] comprises two electrodes which are covered in a dielectric material and act as a capacitor. When a material is brought between the two electrodes and voltage is introduced, the amount of charge stored by the capacitor is the capacitance (displayed as an arbitrary unit). The capacitance is influenced by the hydration of the biological material placed in contact with the electrodes. An increase in hydration will lead to an increase in capacitance and vice versa. (Barel and Clarys, 2006, Courage-Khazaka, 2009a).

The Corneometer[®] has been used in many studies involving different skin sites on the body (Holm et al., 2006, Kim et al., 2002, Matsumoto et al., 2007, Sator et al., 2003, Eberlein-König et al., 2000) but in only two studies evaluating the efficacy of foot treatment (Garrigue et al., 2011, Papanas et al., 2011). The first of these by Garrigue et al. (2011) used Corneometry to compare hydration of the skin of 54 patients with diabetes after using Pedimed[®] moisturiser and a placebo treatment. The second, by Papanas et al. (2011), used the device to evaluate the efficacy of Neuropad Repair Foam[®] on the plantar skin of 20 patients with diabetes. Garrigue et al. (2011) only measured the dorsal aspect of the foot despite the fact that the treatment was designed to prevent ulcers on areas of hyperkeratosis and fissures which are generally found on the plantar aspect. Papanas et al. (2011) followed a similar protocol but measured plantar skin on an area free from hyperkeratosis, the rationale for this is unclear. Both studies found a significant increase in skin hydration post-treatment. Garrigue et al. (2011) observed a 48.9% and 57.3% increase in hydration after 14 and 28 days (p = 0.0002 and < 0.0001 respectively), and Papanas et al. (2011) observed an 8.9% and a 20.6% increase after seven and 14 days respectively (p < 0.001). These results show that the Corneometer[®] is sensitive to skin changes in the foot - changes which are shown to occur over a short period of time. However there is no data yet available for callused skin.

In terms of validation, there is sufficient evidence to support the accuracy and sensitivity of the Corneometer[®] for use in vitro (Barel and Clarys, 1997, Fluhr et al., 1999a). These studies report high correlations (r > 0.8) between the capacitance values and volume of water within filter paper, solutions with different dielectric constants and between Corneometer[®] and other hydration measurement instruments (Barel and Clarys, 1997, Fluhr et al., 1999a). Accuracy tests have also been conducted on forearm skin comparing Corneometer[®] measurements against the Skicon[®] which uses a conductance measurement method (and can therefore be considered a direct measurement of hydration) (Clarys et al., 1999). The two devices correlated highly (r = 0.89) and the intra-rater reliability of the Corneometer[®] was acceptable with a coefficient of variation (CV) ranging from 9-30% (Clarys et al., 1999). A similar comparative study by Fluhr et al (Fluhr et al., 1999b) found similar results in a population of 20 participants using seven skin sites. The Corneometer[®] and Skicon[®] (as well as the Nova DPM impedence and DermaLab capacitance devices) correlated highly and the CV of the Corneometer® ranged from 14-36.8% (Fluhr et al., 1999b). No validation work using the Corneometer[®] on the foot has yet been published and there has been no inter-rater or inter-day reliability data reported.

Aside from capacitance and conductance methods of measuring skin hydration, there are several other available systems (Table B.1). Nuclear magnetic resonance (NMR) spectrometry provides a direct measurement of epidermal hydration. The technique involves the use of a magnetic field to study the proton content of water (this parameter is termed 'resonance of hydrogen'). Transient thermal transfer (TTT) involves the measurement of heat transfer from the body by sending a thermal pulse through the epidermis (Girard et al., 2000). These two measurement techniques have been compared in a study with the Corneometer[®]. Girard et al. (2000) investigated different moisturising products on 12 subjects and found good intra-rater reliability with all three devices (Corneometer[®] CV < 7%; TTT CV < 4% and 2% on different depths; NMR CV < 0.3%). These results were echoed in tests over a three day period using the Corneometer[®] and TTT. The NMR and TTT methods are able to measure different layers of the epidermis and superficial dermis whereas the Corneometer[®] only measures the stratum corneum which could contribute to the higher variance as the moisture content of this layer fluctuates, particularly with application of topical agents (Girard et al., 2000).

While the NMR and TTT could be of particular value in foot dermatology research, the Corneometer[®] is more readily available, far cheaper and easy to use; and has been shown to provide reliable data in several studies. Combined with other instruments measuring different biophysical parameters, it could be of value in the characterisation of plantar skin.

B.3 Trans-epidermal water loss (TEWL)

The Tewameter[®] TM300 (Table B.1) belongs to a group of electronic instruments called evaporimeters which are used to quantify TEWL through the stratum corneum; a measure of skin barrier function (Elkeeb et al., 2010). The Tewameter® TM300 probe consists of a hollow, cylindrical head (dimensions = 1 x 2cm) which is open at both ends and is thus termed an open chamber evaporimeter. Within this cylindrical chamber lies one pair of sensors which measure moisture and temperature. The location of these sensors allows the moisture gradient between them to be measured. The unit of measurement is diffusion flow rate which is expressed in g/h/m² and derives from Fick's law of Diffusion (Courage-Khazaka, 2012).

In order to produce valid measurements, the evaporimeter method must be proven to be a valid measure of skin barrier function, and the different types of devices available must correlate with each other. Elkeeb et al. (2010) used TEWL measurement on titrated skin in vitro using three evaporimeters (Tewameter[®], Vapometer[™] and AquaFlux). They found that TEWL values correlated well with thickness of skin samples (AquaFlux $r^2 = -0.44$, p = 0.01), two of the three device measurements correlated with the flux of titrated water (Tewameter $r^2 = 0.5$, p = 0.0. AquaFlux $r^2 = 0.034$, p0.04) and all the devices could measure skin barrier function over time. Steiner et al. (2011) found that the Tewameter[®] open and Tewameter[®] closed devices, and the Tewameter[®] open and Vapometer devices showed high levels of agreement (r = 0.98 and 0.7 respectively). The Tewameter[®] has been shown to be more sensitive in the low to ranges of TEWL between 15 – 40 g/h/m² and less sensitive above 50 g/h/m² (Steiner et al., 2011).

However, not all studies have shown positive results. Chilcott et al. (2002) investigated the relationship between skin permeability using titrated water and TEWL in vitro. They found no significant correlation existed between TEWL and human epidermis (p = 0.72) or porcine skin (p = 0.068) with titrated water, or tape stripping and needle punctures on porcine skin (p = 0.64 and 0.13 respectively) (Chilcott et al., 2002). It is important to note, however, that this study evaluated skin damage in vitro, which may not have provided conditions where skin barrier function could be adequately measured (Levin and Maibach, 2005).

Although measurements from closed and open chamber evaporimeters have been shown to correlate, it is worth explaining the merits of each. One study concluded that a closed chamber device generally does not need to be used in a controlled climate as the chamber shields the measurement site from any air turbulence and can be used in a variety of different orientations whereas the open chamber device shows greater differences during these tests (Tagami et al., 2002). These authors sought to analyse the difference between a closed and open chamber evaporimeter device (Nikkiso-YSI and DermaLab respectively). Like the studies mentioned above, there was a good correlation between the two devices ($r^2 = 0.92$, p = 0.0001).

Overall, the evaporimeter method of measuring TEWL has been shown to be a valid measure of barrier function. The Tewameter[®] correlates with other evaporimeter devices which indicate that it accurately performs its task of measuring TEWL. This device has potential in the characterisation of plantar skin and could be particularly valuable when comparing normal skin to callused skin which may have altered barrier function due to possible changes in intercellular lipid profile.

B.4 Measurement of the mechanical properties of the skin

The Cutometer[®] (Courage and Khazaka, Köln, Germany) (Table B.1) measures the viscoelastic properties of the skin under negative pressure application (Courage-Khazaka, 2010a, Barel et al., 2006). This deformation of the skin is measured (in mm) by an optical light system (Figure B.1) and the data is displayed as a strain-time or stress-strain curve from which many mechanical parameters can be calculated (Courage-Khazaka, 2010a, Barel et al., 2006). The Cutometer[®] has been used to investigate the effects of aging (Ryu et al., 2008, Sungyeon et al., 2007, Krueger et al., 2011), the effectiveness of different skin grafts (Nguyen et al., 2010, Rahmanian-Schwarz et al., 2011, Sín et al., 2010); and in conditions such as connective tissue disorders (Catala-Petavy et al., 2009), infection (Dobrev, 1998), psoriasis (Dobrev, 2000) and Raynaud's phenomenon (Dobrev, 2007). The Cutometer[®] has been tested for reliability on scar tissue by two studies (Fong et al., 1997, Draaijers et al., 2004). Fong et al (Fong et al., 1997) found from measurements taken on 12 scar sites that the intraclass correlation coefficient (ICC) value showed good reliability (r = 0.776) between three observers using the maximum distension parameter. Draaijers et al (Draaijers et al., 2004) found the reliability to be better on normal skin (n = 20) than scars (n = 49) with normal skin showing high ICC values for all parameters (> (0.75) and scars showing a range of low to high values (0.35 - 0.93) for different parameters with the lowest being obtained from the viscoelastic (UV) parameter. The intra-rater reliability was analysed using CV and this was found to be between 16.1-33.8% for normal skin, and 22.5-36.0% for scar tissue. The authors explained that the lower reliability readings for scars might be because the tissue is stiffer than normal skin, thus giving lower values for each parameter. These, in turn, will be lower compared to the resolution of the device (Draaijers et al., 2004).



Figure B.1 - Measurement principle of the Cutometer (CK, 2010a p.14)

The Cutometer® has also been tested for use on foot skin by Hashmi and Malone-Lee (2007). The authors analysed pressure data for square wave formations and corrected the data to fit the square wave curves to assess device error on three sites of the foot in 20 people. The error was found to be between 1.3 and 3.3%. The stress-strain relationship was assessed by plotting skin displacement against pressure and the relationship between these was found to be linear (r = 0.51 - 0.96) in agreement with Hooke's law which states that stress applied to a material is proportional to the resulting strain (Vincent, 1982). The intra-rater reliability was found to be low (CV = 0.3 - 0.6%). The second stage of the study analysed viscoelastic parameters of the skin on different sites of the foot of 87 people and found that plantar metatarsal skin exhibited the lowest series elastic element, viscoelasticity and plasticity compared to dorsal and medial arch skin (Hashmi and Malone-Lee, 2007). In a study by the same authors (Hashmi et al., 2006), when normal, healthy plantar skin (n = 87) was compared with the skin of individuals with diabetes (n = 87) 103), the series elastic element on retraction and plasticity of the skin was found to be significantly greater in those with diabetes while the viscoelastic properties were not significantly different. These differences could potentially be due to increases in thickness and stiffness of soft tissues in the feet of individuals with diabetes (Duffin et al., 2002,

Chao et al., 2011). Tests on the inter-day reliability of the Cutometer[®] have not been carried out to date.

The duration of negative pressure application (on-time) and duration of skin relaxation after cessation of negative pressure (off-time) can be set by the researcher and influence the elastic (the length of time the skin takes to return to its original state) and plastic (the degree of deformation in shape that occurs) deformation of the skin (Sín et al., 2010, Courage-Khazaka, 2010a). However, the durations of on- and off-times could be a potential source of variation. Having repetitive measures on the same site cause skin creep, meaning that more deformation occurs and the way in which it retracts becomes altered due to being measured in a non-normal state (Dobrev, 2005). Similarly, it is likely that having a single long on-time will also increase skin creep. To obtain accurate information about mechanical properties, an optimal on-time and off-time needs to be set. For the foot, the skin is thicker and less-elastic (Hashmi and Malone-Lee, 2007) which means a greater on-time is required if skin mechanics are to be accurately measured. Hashmi and Malone-Lee (2007) used 60 seconds as on-times and off-times in their work on pedal skin, to achieve this accuracy. This would allow ample time for enough skin to be drawn into the device to minimise any reliability issues, such as were described on scar tissue by Draaijers et al. (2004)

Other, less widely reported mechanical property measuring modalities (Table B.1) include torsion ballistometry (Jemec et al, 2001), indentation (Delalleau et al, 2006; Pailler-Mattei et al, 2008; Zahouani et al, 2009), ultrasound (Diridollou et al, 1998), Frictiometry (Neto et al., 2013) and digital image speckle correlation (Staloff et al, 2008). Few of these techniques have been compared, but those studies which are available are now discussed

Murray and Wickett (1997) found in their study on the mechanical properties of human calf skin that Cutometer[®] data and Dermal Torque Meter[®] (another suction cup device) data did not correlate well with each other, or with the Dermal Phase Meter[®] (a skin capacitance device). This was deemed to be due to the different types of forces applied to the skin (Murray and Wickett, 1997). Jemec et al. (2001) compared a suction cup device (DermaFlex) with a ballistometer (Dia-Stron Torsion Ballistometer) on palmar, and volar and hairy forearm skin, and found a low to moderate correlation between results of the two methods (r = 0.315 - 0.540). Again this was most likely due to difference in measurement techniques: the DermaFlex draws skin through an aperture while the ballistometer analyses skin recovery after tapping it with a known force (Jemec et al.,

2001). A later study (Woo et al., 2014) compared Ballistometer parameters to those of Cutometer on forearm, cheek and forehead skin. They found good correlations (r > 0.07) between Ballistometer parameters and three of the Cutometer parameters based on distensibility on only cheek skin. Other sites did not show good correlations between devices, again due to differences in measurement techniques (Woo et al., 2014). Pedersen et al. (2003) compared two suction cup devices (DermaLab[®] and DermaFlex[®]) and found moderate correlations (r = 0.383 - 0.437) in data obtained from volar forearm skin. The authors considered design difference between devices to be the main issue as the devices use Young's Modulus (stress to strain ratio) in their measurements in slightly different ways. The DermaFlex[®] measures the distance of distension under stress while the DermaLab[®] relies on elasticity constant and material thickness values in addition to measuring deformation of the skin (Pedersen et al., 2003). Neto et al. (2013) compared the Reviscometer[®] to the Cutometer[®] and a friction based skin mechanics measurement device, the Frictiometer[®] on the abdomen of 34 female volunteers. There was no correlation between the Frictiometer[®] and Reviscometer[®]. Weak to moderate correlations were found between Frictiometer[®] and Cutometer[®] parameters, and Cutometer[®] parameters and Reviscometer[®]. Again, the authors cited difference in methodology as a reason for poor correlations. The literature suggests that these devices can clearly identify differences between skin sites, but the lack of agreement between some of the devices aiming to measure similar parameters is an issue. There is a lack of standardisation in the literature of a single 'best' technique, making it difficult to compare data.

The suction cup method has been the most extensively studied. The advantage of the suction cup based Cutometer[®] is that an 8mm diameter aperture probe is available which makes it ideal for determining the mechanical properties of the full skin thickness (including the dermis which influences the skin's mechanical properties), and more rigid callused skin, which might otherwise result in unreliable measures using a small aperture probe. Having a wider aperture will allow for greater vertical deformation of the skin (Barel and Clarys, 1995). Ballistometry involves tapping a very small area of skin to assess how it responds (Jemec et al., 2001) but as plantar skin is thick and quite stiff, it may not be able to create a suitable deflection of the skin. Indentation involves the application of a vertical force to the skin causing an indent which is a more accurate measure of softness and water content (Manny-Aframian and Dikstein, 1995) which again might not be suitable for assessing very rigid plantar skin.

The Reviscometer[®] (Courage and Khazaka, Köln, Germany) (Table B.1) is a relatively new device which purports to measure the elasticity of skin. It is based on the principle of acoustic shear wave propagation whereby an acoustic shockwave is sent between two sensors (approximately 2mm apart) through the skin (Figure B.2). The time which the wave takes to pass from one sensor to the other (resonance running time or RRT) is a surrogate measure of skin stiffness and is displayed as an arbitrary unit (Courage-Khazaka, 2005).



Figure B.2 – Diagram of the Reviscometer® probe (Courage-Khazaka, 2005).

The device has been used in many published studies including tests of skin treatments (Paye et al., 2007, Uhoda et al., 2002) and the assessment of skin variations with age, BMI, body site and gender (Ohshima et al., 2011, Xin et al., 2010, Hermanns-Lê et al., 2001). Two validation studies have been undertaken (Barel et al., 2005, Verhaegen et al., 2010). The first of these (Barel et al., 2005) showed an inverse linear relationship between RRT and polymer stiffness. The RRT showed a significant negative correlation with Young's Modulus from data collected from the polymers using a Cutometer[®]. The second study by Verhaegen et al. (2010) sought to assess the reliability of the instrument on normal skin (of the forearm, upper arm and abdomen) and scar tissue. Two investigators used the Reviscometer[®] to take multidirectional measurements from normal skin (n = 50) and scar tissue (n = 50). An intraclass correlation coefficient (ICC) analysis

revealed that the inter-rater reliability was >0.79 on normal skin and >0.86 on scar tissue for all outcome parameters (mean RRT, mean amplitude and mean ratio). The intra-rater reliability was >0.66 for normal skin and >0.75 for scar tissue. This relatively high reliability might be partly because the measurement probe was left in the same position for both investigators. The reliability may have been reduced had each investigator placed the measurement probe on the skin before taking their sets of measurements, as would happen in the clinical setting. No studies have been published detailing the use of the Reviscometer[®] on pedal skin or on inter-day reliability; however this initial work on reliability appears generally promising.

In areas with a thicker stratum corneum such as plantar skin, there could be an element of error as there is the possibility of the thickness acting as a barrier to shear waves. However, a study by Vexler et al. (1999) used a device similar to the Cutometer[®] on agar gels of varying thicknesses over silicon rubbers of varying stiffness and found that their standard deviations were less than 3%. However, the same study found that the thinner the agar, the more influence the silicon rubber below had on the measurements (Vexler et al., 1999) which might mean that measurements of the dermal elasticity might be more accurate when the epidermis is thinner or less dense. In an unpublished study (Van Engelen et al., 2008) the shear wave penetration depth by the Reviscometer[®] was found to be 0.7mm which means that there could be potential issues with measurement on skin sites with a very thick epidermis.

There is some disagreement in the literature about what the RRT (shear wave) measurement values actually mean. Some argue that RRT measures the density and stiffness of the skin (Hermanns-Lê et al., 2001, Koehler et al., 2009). Others argue that RRT may also indicate of degree of alignment - i.e. a lower mean RRT on one site compared to another may indicate that the fibres run more parallel (Verhaegen et al., 2010). However, fibre orientation may have no bearing on skin stiffness and therefore give false indications of differences in stiffness. Some argue that RRT is indicative of direction of skin tension (Uhoda et al., 2002, Quatresooz et al., 2006, Ohshima et al., 2011) while others argue that it reflects the direction of elastic fibres (Dang et al., 2005). All of the studies took multidirectional measurements by rotating the probe between repeated measures. The mean elasticity is equal to the mean value of all measurement orientations so is in effect a composite measure, reflecting variations in stiffness in various rotations. The direction can be identified by determining the maximum (tension) and minimum (laxity and firmness) RRT values (Hermanns-Lê et al., 2001). Anisotropy, the direction of

the skin's tensile strength, can be determined by calculating the mean amplitude or mean ratio (these, in effect, have been found to correspond to the same variable, but the mean amplitude has been shown to be a more reliable measure) (Verhaegen et al., 2010). It can also be calculated from the coefficient of variance (Hermanns-Lê et al., 2001, Uhoda et al., 2002).

It is surprising that many studies using the Reviscometer[®] have been conducted but none have explicitly investigated what the device actually measures. A future validity study would certainly be of benefit and clarify the device's true value. If the device measures what it claims to, it could have potential in assessing mechanical properties in the skin of patients with diseases such as rheumatoid arthritis and diabetes where the skin's integrity is compromised. In these cases, a device which applies external force such as the Cutometer[®] would be contraindicated and the Reviscometer[®] might provide a useful alternative.

There are many devices available which measure the mechanical properties of the skin directly and as a surrogate measure. The Cutometer[®] is the most common device occurring in the literature and has been validated for use on normal skin, scar tissue and pedal skin. The Reviscometer[®] has been validated on polymers, normal skin and scar tissue. Both of these devices show promise in the results of the available studies and would likely be of benefit for characterising pedal skin provided they can deliver reliable results. Because they have different measurement methods, they could be of significant worth in gaining a more robust mechanical profile of the skin

B.5 Imaging of the skin

Ultrasound (Table B.1) is the use of high frequency sound waves to visualise anatomical structures within the body. The principle is reflection and scattering of the waves off anatomical structures of variable densities, and different tissues of variable densities within a structure. The transducer creates a pulse and the echoes from the reflection and scattering of the sound waves are received by the transducer. These are then processed by the device and an image is created based on the strength of the echoes (Kremkau, 2006). The cross-sectional image, characteristic of diagnostic ultrasound, is seen with B-mode scanning whereby many A-mode scan pulses (the display which represents the amplitudes of a single line of ultrasound echoes) are processed into a greyscale image (Serup et al., 1995).

The use of diagnostic ultrasound in dermatology is not a new concept. Generally in skin research, high frequency ultrasound (above 20MHz) is used as it gives a high level of clarity of both the epidermis and dermis (general use of ultrasound in dermatology is reviewed by Fornage (1995), Schmid-Wendtner and Dill-Muller (2008), Jasaitiene et al. (2011) and Kleinerman et al. (2012)).

There have been many studies that have validated ultrasound for skin thickness measurements by comparing malignant skin lesion images with histological specimens (Fornage et al., 1993, Lassau et al., 1999, Bessoud et al., 2003, Pellacani and Seidenari, 2003, Lassau et al., 2006, Bobadilla et al., 2008, Machet et al., 2009, Vilana et al., 2009). For low frequency ultrasound between 7 and 15 MHz, the correlation between images and specimens is above 0.9 (Lassau et al., 2006, Bobadilla et al., 2008, Vilana et al., 2009), and for high frequency ultrasound at 20MHz, the correlation is above 0.89 (Fornage et al., 1993, Lassau et al., 1999, Bessoud et al., 2003, Pellacani and Seidenari, 2003, Machet et al., 2009). These studies are evidence of the potential accuracy of both low and high frequency ultrasound in skin thickness measurement. Particular advantages of this modality are that it is readily available and poses no risk to patients.

Plantar skin ultrasound studies are scarce and have focussed on the epidermal thickness of patients with diabetes. Duffin et al. (2002) found no significant difference in plantar skin thickness between healthy controls and individuals with type 1 diabetes. However, this study used a 7.5MHz transducer which may not have been sensitive enough to detect minute differences. Hashmi et al. (2006), using 20MHz, found plantar skin to be significantly thicker in individuals with type 2 diabetes than the healthy control group (p = 0.017). Using a 55MHz transducer, Chao et al. (2011) found plantar skin on average to be 6% thicker in patients with diabetes compared to controls.

Despite the good quality images that high frequency ultrasound can deliver, the challenge increases with the presence of hyperkeratosis. Low frequency ultrasound can be used to detect structures deep within the dermis (Jasaitiene et al., 2011) and has been used to assess the mechanical properties of deeper plantar soft tissues (Bygrave and Betts, 1992, Cavanagh, 1999, Hsu et al., 2005). Nishide et al. (2009) used low frequency ultrasound (10MHz) to assess inflammation caused by callused tissue as a predictor of ulcer development in patients with diabetes (Nishide et al., 2009). However, this study did not assess the lesion itself, only the oedema present in the subcutaneous tissue. Potter and Potter (2000a) used A-mode ultrasound to study the speed of callus re-growth. From their pilot work, they determined that A-mode scanning gave reliable results (CV < 4%) for

measuring callus thickness. However, this mode of scanning is wrought with potential error. Because there is no image production in A-mode scanning, repeatable positioning of the transducer is very difficult, the sonographs have to be carefully interpreted, and accompanying anatomical features that may affect data may go completely unnoticed (Potter and Potter, 2000a).

In the context for the research conducted for this thesis, high frequency ultrasound would likely be little value in measuring calluses because thickness and density of the stratum corneum would obscure the image. Keratotic material can cause shadowing and strong reflection obscuring an image (Serup et al., 1995, Schmid-Wendtner and Dill-Muller, 2008). A-mode ultrasound scanning would not be practical due to the lack of an image and difficulty with repeatable probe positioning. An example of a low 7.5MHz frequency image of the plantar metatarsal area (PMA) is shown in Figure B.3.



Figure B.3 – Low frequency ultrasound image of tissue overlying the first metatarsal head. The black stripes near the top of the image reflect the type of obscurity that results from measuring plantar skin. The ultrasound is unable to produce a clear image of the skin due to the amount of keratotic material present in plantar skin.

Other methods for measuring the thickness of skin are available (Table B.1). The most recent innovation is optical coherence tomography (OCT), which uses optical light reflections to produce a high resolution cross-sectional image (Gambichler et al., 2005). Several studies have used the device in the assessment of skin thickness. The OCT measurement technique provides epidermal thickness data which correlates to that obtained by histology, although due to the removal and processing of histological specimens, the explicit values are different (Gambichler et al., 2006, Gambichler et al., 2007, Silver et al., 2012). However, identification of the dermal-epidermal junction (DEJ) is necessary to provide accurate measurements (Gambichler et al., 2006, Josse et al.,

2011). Although no studies measuring plantar skin thickness with this device have been published, Michelson Diagnostics (2012) have provided the author with images of plantar skin which clearly show the layers of the epidermis and DEJ. Figure B.4 shows an OCT image of plantar heel skin. This method of imaging shows promise in dermatology and would be a useful commodity for this research project. However due to the very high cost, it is not readily available.



Figure B.4 – OCT image of plantar heel skin. Here, various layers and landmarks within the skin can be clearly observed.

Magnetic resonance imaging (MRI) is another option for the study of skin thickness. This technique has been deemed to be acceptable in measurement of skin thickness in healthy individuals (Sans et al., 2011). It can also detect changes in the behaviour of water within the skin layers (Richard et al., 1991, Richard et al., 1993, Mirrashed and Sharp, 2004a) and can be useful in assessing states of altered hydration such as in lymphodema (Idy-Peretti et al., 1998). Research into the hydration of skin using moisturisers has also been undertaken and has found MRI to be a useful modality (Mirrashed and Sharp, 2004b).

MRI could potentially be a very effective method in the assessment of plantar skin in the proposed research. However it is expensive, largely inaccessible to researchers outside of radiology, requires specialist training to use, is impractical for recruitment purposes and is also potentially hazardous due to its powerful magnetic field which would pose a danger to subjects or researchers with metal implants or jewellery.

Overall, there are several useful modalities that could be useful for imaging of the skin. MRI and OCT would certainly provide excellent quality images but are unfortunately very expensive and not readily available. Diagnostic ultrasound has been widely used in the characterisation of skin carcinomas and has been validated inr this area. Low frequency

ultrasound is likely to be the best choice for imaging callused skin but may lack image clarity.

B.6 Measurement of skin topography

The measurement of skin topographical properties would be a useful adjunct alongside other measurements, such as hydration and mechanical properties, as there is a noticeable visible change from normal skin to callused skin which likely reflects changes in the microstructural and mechanical properties occurring. There are several instruments available which measure the skin's surface topography. As topography is a wide subject area, the devices available measure different parameters of the skin's surface.

Dermoscopy (Table B.1) is a technique where direct imaging of the skin's surface is achieved (Figure B.5). A wealth of research has been conducted using this modality in the field of melanoma diagnosis (reviewed by Braun et al. (2009)). Several papers have reported the use of dermoscopy in the assessment of pedal skin lesions. Bae et al. (2009) highlighted the usefulness of dermoscopy in differentiating between plantar warts, calluses and corns by assessing the topographical characteristics under magnification. Saida et al. (2004), Miyazaki et al. (2005), and Altamura et al. (2006) have used dermoscopy as a tool for differentiating between benign and malignant melanocytic lesions on pedal skin through analysing the specific topographical patterns. The value of using dermoscopy to assess skin topography is clear from the research papers, and this could have potential value in assessing the skin surface characteristics of plantar callus.



Figure B.5 - Callus photograph (A) and corresponding dermatoscope image (B) (Bae et al., 2009, p.222).

The Visioscan[®] (Courage and Khazaka, Köln, Germany) (Table B.1) is a device which uses a UV light emitting camera to capture images of the surface of the skin producing a high definition image of the stratum corneum surface (Figure B.6). The images are then processed using specialised software (SELS software) based on chosen parameters of skin surface measurement (Courage-Khazaka, 2009b, Tronnier, 1999). The device has been used in various studies analysing skin treatments (Pena Ferreira et al., 2010, Berardesca et al., 2006, Choi et al., 2012), evaluation of the scalp (Pierard et al., 2012, Pierard-Franchimont et al., 2011, Xhauflaire-Uhoda et al., 2010), and the effects of photoaging (Petit et al., 2003, Quatresooz et al., 2011). One study has assessed the reliability of the device (Kottner et al., 2013). Three measures were taken from four volar forearm sites on 12 participants. The ICC was calculated to assess intra-site reliability and it was found that the Visioscan® SELS and roughness parameters all gave highly reliable results (ICC > 0.95). However, these favourable results may have been due to the fact that each set of images was taken by a single investigator (it is not stated whether investigators were changed between participants), all of which were taken without removing the device away from the skin. Also forearm skin is generally homogenous whereas studies of callus involve normal and callused areas. No research with this device has been undertaken on the foot and there are no studies which have investigated the device's inter-rater or interday reliability.



Figure B.6 - Visioscan[®] image of callused skin.

The Visioline[®] (Courage and Khazaka, Köln, Germany) (Table B.1) is a device which is designed specifically for shadow profilometry of wrinkles, undulations in the skin's surface. A silicone negative replica is taken of the skin surface and then placed under an oblique light source. A shadow is formed by the replica where the skin surface shows depressions and wrinkles allowing for the difference in height between high and low areas to be measured (Courage-Khazaka, 2010b). Shadow profilometry is not a new concept, and studies have been undertaken exploring skin furrows multidirectionally on silicone skin replicas using automated measurement systems (Corcuff et al., 1983, Corcuff et al., 1984, Corcuff et al., 1987, Corcuff et al., 1991). So far however, the Visioline[®], a system whereby replicas are oriented and measured manually, has been the topic of very few published research papers and its relationship to other measures of topography have not been reported. Two papers identified by the author include mouse model studies: one assessing the effects of ingestion of collagen tripeptide on photo aging and skin barrier function (Pyun et al., 2012) and the other assessing the effects of cultured fibroblasts on wrinkled murine skin (Jeong et al., 2015). In both cases the device was used for skin replica analysis. The proposed advantages over Visioscan[®] might include the fact that an image of a high quality replica can give exceptional detail of the skin surface contours (Figure B.7) including those caused by striations and sweat ducts.



Figure B.7 - Visioline[®] replica of a callus strip.

Unlike other devices measuring biophysical properties of the skin discussed previously, the validation literature surrounding devices measuring skin surface topography is scant. Dermoscopy has been shown to be useful in visually characterising malignant lesions while the Visioscan[®] has been validated on forearm skin. While the Visioline[®] has no supporting literature available yet, it may also be of use for visual analysis of skin topographical characteristics. Because it uses shadow profilometry, it may be useful for assessing alternative skin characteristics to the dermoscopy such as skin striation or sweat ducts which might be affected in callused skin.

B.7 Gaps in skin measurement research

This section has highlighted several key areas which need to be addressed. (1) With the exception of the Cutometer[®], few measures of skin's biophysical properties have been used and validated on the foot (Corneometer[®] has been used in efficacy studies involving pedal skin (Garrigue et al., 2011, Papanas et al., 2011) but has not been tested for reliability in this area). (2) There is also an overall lack of literature which supports the reliability of these devices in general. While some studies have looked at inter- and intra-rater reliability within a single session using CV and ICC, there has not yet been a study published which has looked at inter-day reliability. (3) The relationship between skin hydration, TEWL, mechanical properties and topography has never previously been explored in plantar skin research.
Conclusion

This review has enabled a substantial acquisition of knowledge in the area of plantar skin physiology, biochemical and biomechanical skin properties, and devices available to measure plantar skin properties. This has enabled the main research problems related to understanding of plantar callus to be identified.

One main conclusion is that there is a lack of research characterising normal and hyperkeratotic plantar foot skin. There is also a lack of literature linking external mechanical factors (compressive and shear pressures) to plantar callus development. An improved scientific knowledge of normal, callused skin and the pathway to callus development would be useful in the evaluation and design of interventions to prevent and treat these common, troublesome skin lesions, and could be fundamental to clinical practice.

In order to characterise the skin, skin hydration, TEWL, mechanical properties, thickness and topography should be considered. These are expected to be interdependent properties. For each skin property, a number of measurement techniques are available. This critical review supports the use of the following instruments: the Corneometer[®], the Cutometer[®], the Reviscometer[®], diagnostic ultrasound, dermoscopy, the Visioline[®] and the Visioscan[®] in that what they measure has face validity as a minimum and some show good reliability for non-foot applications. However, before being used to characterise plantar skin, they need to be validated for use in this area of the body.

Therefore, the purpose of the first study in this thesis will be to assess the interrater and inter-day reliability of these devices which will inform whether these devices are appropriate for plantar skin research (Chapter 3). This will then inform the instruments which are used to characterise normal and callused plantar skin in the second study of this thesis (Chapter 4). Once a robust knowledge of normal and callus skin properties has been established, further work will seek to investigate whether it is feasible to induce callus-like skin changes to normal skin through mechanical loading (Chapters 5 and 6), and changes in callused skin as a result of pressure reduction (Chapter 7).

Chapter 3: Inter-rater and inter-day reliability of non-invasive instruments measuring the biophysical properties of foot skin

3.1 Introduction

In Chapter 2, the author highlighted the need to assess the reliability of available biophysical skin measurement tools for use on the foot. If the devices are deemed to provide reliable data, they can then be used in further research to investigate the biophysical characteristics of normal and callused plantar skin which will be useful in the clinical and research settings. Reliability must be evident between days for one investigator using a measurement device (inter-day) and also between two or more investigators using the same device in the same session (inter-rater). These factors must be investigated because there is a lack of literature that outlines the reliability of devices required to characterise normal and callused foot skin. Also, because plantar skin is weight bearing, it is thicker and stiffer than normal skin, and may have completely different biophysical properties as a result, so its measurement using devices that are designed for other areas of the body might be difficult. Therefore, fitness for purpose and proof of reliability is essential if these devices are to be used in plantar skin research.

In this study, three specific biophysical properties of the skin were selected for measurement which would provide comprehensive information of plantar skin. These properties included skin surface hydration, skin stiffness, and skin surface profilometry. The instruments of choice were: Corneometer[®] (skin surface hydration), Cutometer[®] (distensibility) and Visioscan[®] (skin surface roughness). Aside from the Cutometer[®], none of these devices have been validated for use on pedal skin.

Five additional devices that were reviewed in the previous chapter were excluded from the study. Diagnostic ultrasound was excluded as preliminary tests with the device revealed that images of callus lesions were not clear enough to measure. Dermoscopy was excluded on the grounds that high quality quantitative data could be collected with Visioscan[®]. The Tewameter[®] was excluded as in-vivo and in-vitro preliminary tests showed the device to be too sensitive to movement to use. It was also impractical for use on the foot as the device is designed to be held horizontally at a perpendicular angle to the skin, which would make foot positioning difficult and many surfaces on the foot are not flat. The Reviscometer[®] was excluded because it was found that the data collected on hard, callused skin was invalid and produced outlier data. As the Reviscometer[®] provides only a

surrogate measure of skin mechanical properties, the Cutometer[®] was preferred due to its direct measurement of distensibility. The Visioline[®] was excluded due to the fact that while striation and sweat duct topography might be interesting skin measures, they are time consuming and difficult to measure, prone to human error and are as not important for understanding the properties of callus as the other measures.

3.2 Aims and objectives

The aims of this study were to test the inter-rater and inter-day reliability of the Corneometer[®], Cutometer[®] and Visioscan[®] to assess whether they could be of use in subsequent research studies. The objectives included collecting biophysical data from each device on normal and callused skin on a small sample of participants with two investigators over two days, and analysing agreement between raters and between days.

3.3 Materials and methods

Ethical approval for this study was granted by the University of Salford's College Research Ethics Panel (application number HSCR12/09).

3.3.1 Subjects

Participants were recruited via a poster campaign on the University campus including the Podiatry Clinic and an automated email was sent to staff and students. Participants who had a plantar forefoot callus were recruited provided they did not meet any of the following exclusion criteria which could potentially affect the skin's properties: compromised cardiovascular or neurological status, connective tissue disorders (such as lupus erythematosus), diabetes, autoimmune disorders (such as rheumatoid arthritis), peripheral vascular disease, or wounds/ ulcers of the legs and feet, and eczema, psoriasis or other dry skin disorders affecting the plantar skin.

The foot was assessed by the author of this thesis and confirmed by a HCPC registered podiatrist. Pulses from the posterior tibial and dorsalis pedis arteries were assessed by palpation. Presence of neuropathy was assessed by vibration perception using a 128MHz tuning fork and 10g monofilament test on the hallux, first and fifth metatarsal heads. If these tests, skin assessment and medical history were satisfied, the volunteers were recruited to participate in the study.

3.3.2 Data collection

Data was collected on two consecutive days. At each session, the foot was allowed to acclimatise for approximately 15 minutes before measurements were taken. Temperature and humidity during each session was monitored but not controlled. The participant lay on a plinth with the plantar surface of the foot facing the investigators.

The skin sites measured included: centre of callus, edge of callus, adjacent skin (the distance of the lesion's diameter medial or lateral from the centre of the lesion), and two normal weight bearing skin sites including (1) non-callused skin overlying the plantar metatarsal area (PMA), and (2) adjacent skin overlying the base of the fifth metatarsal (figure 3.1). The callus centre was chosen because of its location at the centre of the lesion, and could thus be considered the most 'callused' area. The callus edge was chosen because of its potential importance in later longitudinal studies; a change in callus properties post-treatment may potentially become apparent faster at an area of 'less callused' skin, such as the boundary of the lesion itself. Adjacent skin located close to the callus was chosen because of interventions. Normal weight-bearing (PMA) and semi weight-bearing (fifth metatarsal base) skin sites were chosen to compare with callused skin and to give a good profile of normal skin properties.



Figure 3.1 - The measurement sites on the callus include the centre, edge, and skin directly adjacent to the callus; and skin overlying the plantar metatarsal area (PMA) and base of the 5th metatarsal (adapted from Hashmi and Malone-Lee, 2007 p.253).

On day one, the skin sites to be measured were marked with a water soluble ball point pen. The centre and edge of the callus were identified and the width was measured using a ruler. Generally there is a difference in skin texture and hardness between callused and non-callused skin so the boundary between these areas, the callus edge, can be visualised and palpated. The adjacent skin was positioned 200% of the radius length away from the callus centre. The fourth metatarsal was used as a control site, unless the callus lay over this site, in which case the first metatarsal head was used. The fifth metatarsal base of the same foot was used as a second control site. Both control sites were identified through palpation and marked. The distance of the radii of the measurement probes were marked over the centre of each skin measurement site to allow accurate probe placement.

For each skin site and device, the first investigator would take measurements followed by the second investigator. Each investigator was blinded to their own measurements during testing to eliminate measurement bias. On the second day, the investigator order was reversed. Measurements were taken based on manufacturers' instructions but adjusted for the needs of plantar skin testing as follows. For each investigator, 10 measurements were taken per skin site using the Corneometer[®] and one image was taken per site for the Visioscan[®]. For the Cutometer[®], 500mbar of negative

pressure was applied. The on-time and off-time were set to 30 seconds each. One cycle of on-time and off-time was used on each skin site by both investigators.

Single data values for hydration are produced by the Corneometer[®] and a mean was taken from the 10 measurement values obtained from each skin site. The Cutometer[®] produces a range of different values based on stiffness, elasticity and plasticity. The Uf parameter, which measures maximum skin distensibility (a surrogate measure for stiffness), was chosen based on the hypothesis that callused skin is stiffer than normal plantar skin. Two variables, *variance* and *homogeneity*, were obtained from the Visioscan[®] images. These were chosen based on the hypothesis that callused skin is rougher than normal skin and therefore more variable and less homogenous in appearance. The homogeneity parameter is a measure of how homogenous the skin is and measures the combinations of grey levels in the image. The more frequent the grey level combination appears in the image, the lower the homogeneity and the rougher the skin (Courage-Khazaka, 2009b).

3.3.3 Statistical analysis

To compare inter-rater and inter-day reliability, the ability of an instrument to distinguish between individuals (de Vet et al., 2006), the intraclass correlation coefficient test was used (Shrout and Fleiss, 1979). This test calculates reliability by dividing the within subject variance (σ_p^2) by the within subject variance plus measurement error: the systematic variance between raters (σ_{pt}^2) and the variance not explained by test subjects or raters ($\sigma_{Residual}^2$) (de Vet et al., 2006). Systematic variance between raters were included in the measurement error because these are important to take into account when devices are intended to be used by different investigators (de Vet et al., 2006).

Intraclass correlation coefficient

$$ICC = \frac{\sigma_p^2}{\sigma_p^2 + \sigma_{pt}^2 + \sigma_{Residual}^2}$$

(de Vet et al., 2006, p.1036)

This test was chosen over Kappa because it is designed for continuous data whereas Kappa is designed for categorical data (Landis and Koch, 1977). Based on reliability statistic criteria for the Kappa equation, agreement ratings were assessed according to the following: slight (r = 0.0 - 0.2), fair (r = 0.21 - 0.4), moderate (r = 0.41 - 0.6), substantial (r = 0.61 - 0.8), and high (r = 0.81 - 1.0) (Landis and Koch, 1977, p.165).

To assess the inter-rater and inter-day agreement, the standard error of measurement (SEM) and smallest detectable change (SDC, also known as smallest detectable difference, minimum detectable change or minimum important difference) (de Vet et al., 2006) were calculated. The SEM gives the measurement error by calculating the square root of the error variance. This is done by dividing the standard deviations of the mean differences by the square root of two. The square root of two is used because change is calculated by the difference between two measurement values (de Vet et al., 2006).

Standard error of measurement

$$SEM = \frac{SD_{Differences}}{\sqrt{2}}$$
(de Vet et al., 2006, p.1037)

The SEM is useful alone when there is a clear understanding of the data, and of what magnitude of difference is important. However, when this is unclear, the SDC can be calculated and conceptualises the SEM value. The SDC gives the value which a change must exceed in order to be considered real and not due to error - it gives a value which can be considered a guide for detecting clinically relevant changes in data (de Vet et al., 2006). It is calculated by multiplying the SEM by the square root of two and 1.96, and produces the same value at Bland and Altman's Limits of Agreement.

Smallest detectable change

 $SDC = 1.96 \times \sqrt{2} \times SEM$ (de Vet et al., 2006, p.1038)

Descriptive data including means, 95% confident intervals (CI) and percentage differences between datasets were also calculated. Percentage difference was calculated by

subtracting value 2 from value 1, dividing by the value 1 and multiplying by 100. In this study, Day 2 and Investigator 2's device readings were taken as value 2 for inter-day and inter-rater differences.

Percentage difference

% difference =
$$\frac{Value \ 2 - Value \ 1}{Value \ 1} \times 100$$

Statistical analysis was undertaken on SPSS 20.0 and Microsoft Office Excel 2010.

3.4 Results

The following section outlines the result of the study. 'Inter-rater' reliability refers to reliability assessed between the two investigators on day 1, then again on day 2. 'Interday' reliability refers to reliability for each investigator between the two days. Inter-day reliability may also be referred to as intra-rater reliability because it assesses two sets of results of one rater, but the term 'inter-day' was deemed more appropriate because measurement sessions were carried out one day apart. Eight healthy adults (with a total of eight calluses) gave informed consent to take part in the study. Of the eight calluses, three were consistent with Merriman's grade 1 'no specific callus plaque, but diffuse or pinch callus, or present in narrow bands' and five were consistent with grade 2 'circumscribed, punctuate oval or circular, well-defined thickening of keratinized tissue' (Springett and Merriman, 1995, p.207).

3.4.1 Corneometer[®]

The inter-rater and inter-day reliability for the Corneometer[®] data is shown in table 3.1 - 3.2. ICC values were high across most skin sites (> 0.8) for both inter-rater and inter-day reliability with a few exceptions. For the callus centre, the inter-day reliability was moderate (r = 0.44) for Investigator 1; and fair (r = 0.25) for Investigator 2. The SEM and SDC values ranged from 0.73 - 2.28au and 1.9 - 6.32au respectively between investigators. Between days, the SEM and SDC values ranged from 1.36 - 3.69au and 3.76 - 10.23au respectively. The percentage differences between investigators were variable. The highest percentage differences on Day 1 were found at the callus edge and skin overlying the fifth metatarsal base (14.9% and 7.8% respectively). On Day 2, the highest

differences were at the callus edge and normal PMA skin sites (13.9% and 11.2% respectively). The largest percentage differences for Investigator 1 were at the callus centre and adjacent skin sites (28.5% and 13.6% respectively). For Investigator 2, the largest differences were at the callus centre and normal skin overlying the PMA (21.9% and 10.2% respectively).

Skin site	Mean (95% CI)	n (95% CI) Mean (95% CI) %		ICC	SEM	SDC
	Investigator 1	Investigator 2	difference			
	I	Day 1				
Callus centre	4.39 (1.48 - 7.31)	4.53 (1.39 - 7.67)	3.1	0.8	1.56	4.32
Callus edge	7.92 (1.18 - 14.67)	6.74 (1.77 - 11.71)	14.9	0.87	2.28	6.32
Skin adjacent to callus	11.99 (2.69 - 21.29)	11.39 (2.65 - 20.13)	5.0	0.98	1.36	3.77
Normal PMA skin	13.57 (5.72 - 21.42)	14.05 (5.67 - 22.43)	3.6	1.0	0.89	2.47
Skin overlying 5th	15.33 (9.34 - 21.32)	14.14 (9.49 - 18.79)	7.8	0.93	1.36	3.77
metatarsal base						
	l	Day 2		•		
Callus centre	5.65 (3.12 - 8.17)	5.52 (3.25 - 7.80)	2.2	0.94	0.68	1.9
Callus edge	7.10 (2.53 - 11.66)	6.11 (2.31 - 9.90)	13.9	0.92	1.21	3.36
Skin adjacent to callus	10.36 (3.84 - 16.89)	10.61 (5.25 - 15.97)	2.4	0.93	1.82	5.04
Normal PMA skin	13.92 (6.95 - 20.89)	15.49 (6.53 - 24.44)	11.2	0.94	1.98	5.49
Skin overlying 5th	14.66 (9.33 - 19.98)	15.35 (9.24 - 21.46)	4.7	0.98	0.73	2.02
metatarsal base						

Table 3.1 – Corneometer[®] inter-rater reliability

Mean, 95% CI, SEM and SDC represent hydration values in arbitrary units.

Skin site	Mean (95% CI)	Mean (95% CI)	%	ICC	SEM	SDC
	Day 1	Day 2	difference			
	Investig	gator 1				
Callus centre	4.39 (1.48 - 7.31)	5.65 (3.12 - 8.17)	28.5	0.44	2.2	6.09
Callus edge	7.92 (1.18 - 14.67)	7.10 (2.53 - 11.66)	10.4	0.85	2.51	6.97
Skin adjacent to callus	11.99 (2.69 -	10.36 (3.84 -	13.6	0.9	2.63	7.28
	21.29)	16.89)				
Normal PMA skin	13.57 (5.72 -	13.92 (6.95 -	2.6	0.95	1.87	5.18
	21.42)	20.89)				
Skin overlying 5th	15.33 (9.34 -	14.66 (9.33 -	4.4	0.94	1.53	4.25
metatarsal base	21.32)	19.98)				
	Investig	gator 2				
Callus centre	4.53 (1.39 - 7.67)	5.52 (3.25 - 7.80)	21.9	0.25	2.59	7.18
Callus edge	6.74 (1.77 - 11.71)	6.11 (2.31 - 9.90) 9.4		0.87	1.76	4.88
Skin adjacent to callus	11.39 (2.65 -	10.61 (5.25 -	6.8	0.8	3.69	10.23
	20.13)	15.97)				
Normal PMA skin	14.05 (5.67 -	15.49 (6.53 -	10.2	0.97	1.36	3.76
	22.43)	24.44)				
Skin overlying 5th	14.14 (9.49 -	15.35 (9.24 - 8.6		0.89	1.86	5.15
metatarsal base	18.79)	21.46)				

Table 3.2 – Corneometer[®] inter-day reliability

Mean, 95% CI, SEM and SDC represent hydration values in arbitrary units.

3.4.2 Cutometer[®]

The inter-rater and inter-day reliability for the Cutometer[®] data is shown in Tables 3.3 - 3.4. The ICC values on callused skin were high (r = 0.68 - 1.0) for both inter-rater and inter-day reliability, aside from inter-rater reliability for adjacent skin on Day 2 which was moderate (r = 0.42). For both inter-rater and inter-day reliability tests, the ICC values ranged from fair to moderate for PMA skin (r = 0.27 - 0.54) and slight to substantial for the 5th metatarsal base (r = 0.1 - 0.7). The SEM and SDC values ranged from 0.01 - 0.26mm and 0.04 - 0.71mm respectively between investigators. Between days, the SEM and SDC values ranged from 0.05 - 0.28mm and 0.14 - 0.78mm respectively. The percentage differences between investigators were varied. On Day 1, the callus centre had the greatest percentage difference (37.8%) followed by the normal PMA and skin overlying the 5th metatarsal base which were also large (15.1% and 16.6% respectively). For Day 2, the largest differences included callus edge and skin overlying the 5th metatarsal base (20.8%, and 20.9% respectively). Between days, for Investigator 1 the

largest percentage differences included callus centre and edge (36.6% and 18.7% respectively). For Investigator 2, the largest percentage differences were at the skin adjacent to callus and the skin overlying the base of the fifth metatarsal (6.2% and 8.5% respectively).

Skin site	Mean (95% CI)	Mean (95% CI)	%	ICC	SEM	SDC
	Investigator 1 Investigator 2		difference			
		Day 1		•		
Callus centre	0.83 (0.30 - 1.36)	0.52 (-0.01 - 1.04)	37.8	0.71	0.26	0.71
Callus edge	0.65 (0.10 - 1.21)	0.65 (0.17 - 1.13)	0.3	0.91	0.18	0.51
Skin adjacent to	1.00 (0.66 - 1.34)	1.03 (0.84 - 1.23)	3.1	0.77	0.15	0.43
callus						
Normal PMA skin	1.24 (1.08 - 1.39)	1.05 (0.90 - 1.20)	15.1	0.53	0.06	0.17
Skin overlying 5th	1.03 (0.81 - 1.26)	0.86 (0.74 - 0.98)	16.6	0.51	0.11	0.31
metatarsal base						
		Day 2				
Callus centre	0.52 (0.11 - 0.94)	0.51 (0.09 - 0.93)	3.1	1.0	0.01	0.04
Callus edge	0.77 (0.42 - 1.13)	0.61 (0.19 - 1.03)	20.8	0.87	0.12	0.52
Skin adjacent to	1.03 (0.76 - 1.30)	0.97 (0.81 - 1.12)	6.1	0.42	0.19	0.52
callus						
Normal PMA skin	1.16 (1.01 - 1.31)	1.05 (0.96 - 1.14)	9.6	0.54	0.07	0.21
Skin overlying 5th	1.00 (0.89 - 1.10)	0.79 (0.74 - 0.83)	20.9	0.1	0.07	0.2
metatarsal base						

Table 3.3 – Cutometer[®] inter-rater reliability

Mean, 95% CI, SEM and SDC represent maximum distension values in mm.

Skin site	Mean (95% CI) Mean (95% CI)		%	ICC	SEM	SDC	
	Day 1	Day 2	difference				
Investigator 1							
Callus centre	0.83 (0.30 - 1.36)	0.52 (0.11 - 0.94)	36.6	0.63	0.28	0.78	
Callus edge	0.65 (0.10 - 1.21)	0.77 (0.42 - 1.13)	18.7	0.74	0.26	0.73	
Skin adjacent to callus	1.00 (0.66 - 1.34)	1.03 (0.76 - 1.30)	2.9	0.87	0.13	0.36	
Normal PMA skin	1.24 (1.08 - 1.39)	1.16 (1.01 - 1.31)	6.2	0.27	0.14	0.39	
Skin overlying 5th	1.03 (0.81 - 1.26)	1.00 (0.89 - 1.10)	3.5	0.7	0.11	0.3	
metatarsal base							
	Investig	gator 2	•	1	1		
Callus centre	0.52 (-0.01 - 1.04)	0.51 (0.09 - 0.93)	1.4	0.95	0.12	0.34	
Callus edge	0.65 (0.17 - 1.13)	0.61 (0.19 - 1.03)	5.8	0.99	0.05	0.14	
Skin adjacent to callus	1.03 (0.84 - 1.23)	0.97 (0.81 - 1.12)	6.2	0.68	0.11	0.3	
Normal PMA skin	1.05 (0.90 - 1.20)	1.05 (0.96 - 1.14)	0.2	0.37	0.11	0.31	
Skin overlying 5th	0.86 (0.74 - 0.98)	0.79 (0.74 - 0.83)	8.5	0.26	0.08	0.23	
metatarsal base							

Table 3.4 – Cutometer[®] inter-day reliability

Mean, 95% CI, SEM and SDC represent maximum distension values in mm.

3.4.3 Visioscan[®]

The inter-rater and inter-day reliability for the Visioscan[®] data is shown in Tables 3.5 - 3.6 (homogeneity) and 3.7 - 3.8 (variance). Across both parameters, the inter-rater and interday reliability ranged from moderate to high (r = 0.52 - 0.91). For the homogeneity parameter, the SEM and SDC values ranged from 0.02 - 0.06au and 0.06 - 0.16aurespectively between investigators and 0.04 - 0.08au and 0.1 - 0.23au respectively between days. For the variance parameter, the SEM and SDC values ranged from 0.4 - 0.93au and 1.11 - 2.58au respectively between investigators and 0.49 - 0.81au and 1.37 - 2.24aurespectively between days. For homogeneity, the largest percentage differences between investigators was at skin overlying the base of the fifth metatarsal on Day 1 (3.1%difference) and over the PMA on Day 2 (1.5% difference). Between days, the highest percentage difference for both Investigator 1 and Investigator 2 was at the normal PMA skin site (1.3% and 2.9% respectively). For variance data, the largest difference between investigators was for normal PMA skin on Day 1 (12.2%) and on Day 2, the callus centre (4.1%). The largest inter-day percentage difference for Investigator 1 was at the normal PMA skin site (3.6%) and for Investigator 2, the callus centre (12.6%).

Skin site	Mean (95% CI) Investigator 1	Mean (95% CI) Investigator 2	% difference	ICC	SEM	SDC
	•	Day 1		1	1	
Callus centre	1.36 (1.23 - 1.49)	1.39 (1.29 - 1.49)	1.9	0.83	0.05	0.14
Normal PMA skin	1.45 (1.34 - 1.56)	1.49 (1.39 - 1.59)	2.8	0.91	0.02	0.06
Skin overlying 5th metatarsal base	1.42 (1.34 - 1.49)	1.46 (1.39 - 1.53)	3.1	0.54	0.05	0.14
	•	Day 2	·			
Callus centre	1.36 (1.26 - 1.45)	1.34 (1.25 - 1.44)	1.0	0.75	0.05	0.15
Normal PMA skin	1.47 (1.38 - 1.56)	1.45 (1.33 - 1.57)	1.5	0.76	0.06	0.16
Skin overlying 5th metatarsal base	1.40 (1.33 - 1.48)	1.42 (1.34 - 1.51)	1.3	0.87	0.03	0.09

Table 3.5 – Visioscan[®] homogeneity inter-rater reliability

Mean, 95% CI, SEM and SDC represent homogeneity values in au.

Table 3.6 – Visioscan[®] homogeneity inter-day reliability

Skin site	Mean (95% CI) Day 1	Mean (95% CI) Day 2	% difference	ICC	SEM	SDC
Investigator 1						
Callus centre	1.36 (1.23 - 1.49)	1.36 (1.26 - 1.45)	0.5	0.87	0.05	0.13
Normal PMA skin	1.45 (1.34 - 1.56)	1.47 (1.38 - 1.56)	1.3	0.68	0.06	0.17
Skin overlying 5th metatarsal base	1.42 (1.34 - 1.49)	1.40 (1.33 - 1.48)	0.8	0.7	0.05	0.13
	Investig	gator 2				
Callus centre	1.39 (1.29 - 1.49)	1.34 (1.25 - 1.44)	3.2	0.82	0.04	0.1
Normal PMA skin	1.49 (1.39 - 1.59)	1.45 (1.33 - 1.57)	2.9	0.54	0.08	0.23
Skin overlying 5th metatarsal base	1.46 (1.39 - 1.53)	1.42 (1.34 - 1.51)	2.5	0.52	0.06	0.16

Mean, 95% CI, SEM and SDC represent homogeneity values in au.

Skin site	Mean (95% CI) Investigator 1	Mean (95% CI) Investigator 2	% difference	ICC	SEM	SDC
	•	Day 1	·			
Callus centre	6.43 (4.41 - 8.45)	5.97 (4.56 - 7.38)	7.2	0.76	0.93	2.58
Normal PMA skin	5.10 (3.44 - 6.76)	4.48 (3.11 - 5.84)	12.2	0.88	0.41	1.14
Skin overlying 5th metatarsal base	5.68 (4.47 - 6.89)	5.15 (4.09 - 6.22)	9.3	0.66	0.69	1.91
		Day 2				
Callus centre	6.46 (5.02 - 7.90)	6.72 (5.29 - 8.14)	4.1	0.8	0.72	1.99
Normal PMA skin	4.91 (3.45 - 6.37)	4.96 (3.49 - 6.42)	0.9	0.99	0.4	1.11
Skin overlying 5th metatarsal base	5.84 (4.68 - 7.00)	5.68 (4.31 - 7.04)	2.8	0.91	0.41	1.14

 Table 3.7 – Visioscan[®] variance inter-rater reliability

Mean, 95% CI, SEM and SDC represent variance values in au.

Table 3.8 – Visioscan[®] variance inter-day reliability

Skin site	Mean (95% CI) Day 1	Mean (95% CI) Day 2	% difference	ICC	SEM	SDC
	Investi	gator 1				
Callus centre	6.43 (4.41 - 8.45)	6.46 (5.02 - 7.90)	0.4	0.86	0.76	2.12
Normal PMA skin	5.10 (3.44 - 6.76)	4.91 (3.45 - 6.37)	3.6	0.86	0.67	1.85
Skin overlying 5th metatarsal base	5.68 (4.47 - 6.89)	5.84 (4.68 - 7.00)	2.7	0.86	0.49	1.37
Investigator 2						
Callus centre	5.97 (4.56 - 7.38)	6.72 (5.29 - 8.14)	12.6	0.79	0.55	1.52
Normal PMA skin	4.48 (3.11 - 5.84)	4.96 (3.49 - 6.42)	10.8	0.79	0.68	1.87
Skin overlying 5th metatarsal base	5.15 (4.09 - 6.22)	5.68 (4.31 - 7.04)	10.1	0.61	0.81	2.24

Mean, 95% CI, SEM and SDC represent variance values in au.

3.4.4 Order of measurement observations

On observation of the mean values (Tables 3.1 - 3.8), no patterns were identified in terms of the effect of the order of measurement (Investigator 1 vs Investigator 2) except for the Cutometer[®] where Investigator 1 consistently obtained higher values in general over both days. Investigator 1 collected data first on day 1 and second on day 2.

3.5 Discussion

Generally, the inter-rater and inter-day reliability has been shown to be good, though variable, across the different instruments. The values concur with the available reliability literature (Fong et al., 1997, Draaijers et al., 2004, Kottner et al., 2013). For example Fong et al. (1997) reported Cutometer[®] inter-rater reliability scores (r = 0.78) on scar tissue and these are similar to ICC values found on callused skin sites in this study (generally high r > 0.7 in 83.3% of cases).

For the Corneometer[®], the inter-rater reliability was good while the inter-day reliability was low for the centre of callus. There are several possible reasons why this might be the case. Firstly, because callused skin is particularly hard and protrudes from the foot, this could potentially affect the amount of pressure applied by the probe. Hard, protruding calluses may have caused greater deformation of the springs in the probe head leading to higher measurement values. Indeed, Clarys et al. (2011) found that in very dry skin, there was an increase of over 40% in capacitance reading when the operator applied a higher manual pressure, which would deform the springs in the probe head more than a lighter pressure. At the driest skin (12.6 au) there was an increase of 42% when applying high pressure. Preliminary work done by the author of this thesis found a mean increase of 24.4% when deliberately applying high pressure over different plantar skin sites on a single subject. Thus, variations in the pressure applied by the operators may have had some influence on measurements and thus affecting the reliability. These factors might explain the low inter-day reliability, but it does not explain why the inter-rater reliability was still high. Perhaps callused skin is more prone to hydration changes between days that the other sites - hydration is more likely to vary between days than in a single session which might explain why inter-rater reliability was higher than inter-day reliability. The low measurement values compared to the other skin sites are at the lower end of the device range which might also affect results. Inter-day variation in skin hydration could also have contributed to poor inter-day reliability, whereas hydration is less likely to consistently vary in a single session and inter-rater reliability was higher. Given this fact, it might be that having each investigator measure twice on the same day would have given higher reliability scores, instead of having measures between days. However, since the instruments are intended to be used longitudinally, it was appropriate to test reliability between days, despite the fact that it could have given intermittently lower reliability scores.

The percentage differences between investigators and between days were variable for the Corneometer[®] with the highest inter-rater difference being observed for the callus edge on Day 1 (14.9%) and the highest inter-day difference observed on callus centre for Investigator 1 (28.5%). However, on Day 1, the inter-rater differences between PMA and skin overlying the fifth metatarsal base were 3.6% and 7.8% respectively, and for Investigator 1, 2.6% and 4.4% respectively. Across the entire dataset, 90% of the differences were less than 15%. The mean percentage difference between callus and control sites across both investigators and days (65.5%) was larger than the differences between investigators and days by a considerable margin.

The Cutometer's[®] inter-rater and inter-day reliability for callused skin was generally moderate to good and ICC values on callused skin sites reflect the literature on normal and scar tissue (Fong et al., 1997, Draaijers et al., 2004). However, inter-rater and inter-day reliability was more variable for the adjacent and control skin sites. For noncallused PMA skin, there was moderate inter-rater agreement but poor inter-day agreement. This again might suggest natural skin variation over a day-to-day period. The fifth metatarsal base was even more variable with poorer inter-rater agreement on the second day, and poorer inter-day agreement for Investigator 2. For these skin sites, Investigator 1 had slightly higher readings than Investigator 2 which could be a result of differences in probe application pressure. One paper, published after the study in this chapter was completed, has highlighted that applying a heavy load to the probe can significantly affect the measurements (P < 0.005) compared to applying a light or no load (Bonaparte et al., 2013). The authors recommended applying minimal force to the probe when holding it to the skin. However, it must be noted that due to the nature of the skin sites in this study and the fact that the probe is positioned perpendicular to the foot and parallel to the ground, the application of some pressure is necessary to keep a tight seal between the probe and skin, and to prevent movement of the probe. Holding the probe so the spring loaded head was in line with the outer rim, but without pressing excessively hard, was observed to give a smoother distension/ time curve than applying too little pressure – where fluctuations in skin distension readings were evident during a pressure application cycle due to minute movements. The spring loaded head is intended to ensure constant probe pressure (Courage-Khazaka, 2010a) and given that both investigators in the study had experience of handling the probe and applying constant pressure between measurements, any effect of probe pressure was likely to be minimal.

The high reliability values on callused compared to normal skin sites with the 8mm diameter Cutometer[®] probe suggests it might be better suited to stiffer skin than normal skin, such as that overlying the PMA and base of the fifth metatarsal. The fact that skin overlying the PMA and fifth metatarsal base is more convex in nature compared to callused skin might also have played a part in the initial position of the skin inside the probe which could have led to discrepancies in reliability. Methodologically, there are unlikely to be any aspects of the protocol that could be modified to counteract this effect. Using a smaller aperture measurement probe might have given more reliable results on control site skin, but at the expense of less accurate data obtained from callused skin (assuming a smaller aperture becomes challenging on stiffer skin). The positioning of the probe was as accurate as possible and the probe was repositioned between each measurement which has been shown to be more reliable than leaving the probe attached to the skin between measurements, which could lead to other factors, such as distortion of the skin, or skin occlusion, which will affect the measurements (Bonaparte and Chung, 2014).

The percentage differences between investigators and between days were variable for the Cutometer[®] with the highest inter-rater and inter-day difference being observed on the callus centre for Day 1 (37.8%) and Investigator 1 (36.6%) respectively. However, the percentage differences between days for the PMA and base of fifth metatarsal were 15.1% and 16.6% respectively, and for Investigator 1, 6.5% and 3.2% respectively. Across the entire dataset, 80% of differences between investigators and between days were below 20%. The mean difference between callus and control sites across both investigators and sessions was 41.6%, which is much larger than most of the inter-rater and inter-day differences.

The reliability of the Visioscan[®] is generally moderate to good for callused skin sites. While the ICC values were lower than those reported by Kottner et al. (2013), the probe was manually positioned by each investigator before imaging and was not fixed in position. The device is very sensitive and any slight change in probe position can affect the values. While a guidance mark was placed at each skin site, the probe would still have to be aligned by each investigator before taking the measurement, and between days. This phenomenon and variations in other skin properties, which may have affected the skin's surface profile, together could account for the moderate agreement. For the homogeneity parameter, the percentage differences between investigators and between days were very small ranging from 0.5% to 3.2%. These differences are smaller than the mean percentage difference between callus and control sites across investigators and days which was 5.8%.

For the variance parameter, the largest percentage difference between investigators was seen for normal PMA skin on Day 1 (12.2%). The largest percentage difference between days was Investigator 2's callus centre data (12.6%). On day 1, the differences were 9.3% for the fifth metatarsal base, and for Investigator 2, 10.1%. However, across the entire dataset for this device, 66.67% of differences were below 10%; and 50% of differences were below 5%. Compared to the mean percentage difference between callus and control sites across investigators and days (23%), the inter-rater and inter-day differences are very small.

Across all the devices, the SEM and SDC values were variable with callused sites generally showing less agreement than normal sites. The SDC, which is calculated from the SEM, suggests in some cases, particularly skin hydration and stiffness at callus sites, that quite large changes in data might be necessary to identify a change that can be deemed as 'real' and not as a result of measurement error. However, this should not be an issue so long as control sites, which showed marked differences from callused skin in this study are used as a comparison (on average the control sites were 65.6%, 41.6%, 5.6% and 24.2% different from callused skin for hydration, distensibility, homogeneity and variance respectively). Thus, variation between days in the measurements for Investigator 1 and 2 are lower than the differences one might expect to see between callus and non-callused sites. The level of agreement in absolute units of measurement may appear quite low in some cases, but this must be taken into context with the other results. The agreement results may be a product of the fact that plantar skin properties can deviate in a short space of time (i.e. Day 1 to Day 2) and the devices are sensitive to these skin changes as well as the variation in the investigators' operating skills. This suggests use of control sites is important in future research designs and that change in skin properties compared to changes at a control site is a key approach to data analysis. Given that reliability and explicit differences in mean scores are acceptable and show that differences between skin sites that are greater than differences between investigators and days, the SDC scores are not concerning in that, for the intended studies of foot skin, the measurement approaches appear fit for purpose.

This study was undertaken as part of a larger project investigating the reliability of biophysical measurement devices which was published in the Journal of Foot and Ankle Research (Hashmi et al., 2015b). The results from this study also relate to corn and heel fissure skin which was not included in this chapter. The main difference between this study and the data presented in this chapter are the inclusion of the Reviscometer[®] probe. The

author wrote this chapter after the study had concluded and it was clear from the Reviscometer[®] data that it was greatly affected by hardness of the skin, and regularly showed error values. Because this phenomenon was evident from preliminary tests, before the study had commenced, the decision was made to omit the results. A further difference between the results of this chapter and the published study are that the Visioscan[®] parameters are different. The homogeneity and variance parameters were chosen by the author of this thesis because they reflected what the author felt were the most important aspects to measure, and anecdotally reflected what was observed during testing. These parameters were also consistently reliable, which is of importance for future testing.

3.6 Conclusions

This chapter has demonstrated development of a greater understanding of research techniques in terms of design and methodology for skin measurement techniques, and in data handling and statistical analysis. Furthermore, it has allowed evaluation of the skin instruments thus enabling the author to make judgements on their limitations and potential use in further research.

The results of the study have shown good inter-rater and inter-day reliability of the Corneometer[®], Cutometer[®] and Visioscan[®]. The ICC values were generally good and comparable to the literature and the percentage differences in mean values between investigators and days were smaller than differences between normal and callused skin sites. This means that variations between investigators and days should not mask the expected differences in skin properties between sites.

The SDC data suggests that control site data will be important when collecting data over a longitudinal time period as normal variations in skin properties over time could mask skin changes caused by a potential intervention or other independent variables (such as an increase or decrease in load applied to the foot site). The inter-rater reliability could have been affected by variation in pressure applied by each investigator and probe positioning, and skin measurement tools remain sensitive to operator judgement. The inter-day reliability across all the devices could have been affected by variations in the skin's properties and probe positioning, and thus reflect more than simply variation due to operator technique.

Chapter 4: The biophysical characteristics of normal and callused plantar skin

4.1 Introduction

Chapter 2 highlighted that there was a gap in the research literature regarding the biophysical properties of normal and callused plantar skin, and the relationship between pressure and callus development. Therefore, various skin properties were highlighted which would be useful to measure to characterise the skin including hydration, mechanical properties, TEWL, skin surface topography and cross sectional thickness. Chapter 3 investigated the inter-rater and inter-day reliability of the measurement devices that characterised the hydration, mechanical properties and topography of normal and callused plantar skin. The study reported on the extent to which reliable data collection was possible with the Corneometer[®] (hydration), Cutometer[®] (distensibility) and Visioscan[®] (skin surface topography parameters: homogeneity and variance), and thus the limits within which any study of differences between normal and callused skin sites would be possible.

Knowledge of the normal characteristics of plantar skin is essential if one is to quantify what an abnormal change in skin is and evaluate the effectiveness of a treatment to return skin to a normal state. Also, understanding how an abnormal skin site is different may point at treatment strategies more/less likely to work, because approaches that target a specific difference between sites (e.g. in hydration) could be prioritised.

In addition, to investigate what causes callus to develop, or regress, a researcher may seek to manipulate an independent variable (e.g. external load) and observe change in an area of skin, either away or towards normal values. Thus, quantifying normal and abnormal skin properties helps define the boundaries for studies such as this. Work of this sort is described in subsequent chapters. The study outlined in this chapter was concerned with collecting biophysical data from normal and callused plantar skin on the forefoot which could be used to characterise their biophysical properties. The hydration, mechanical and topographical properties of these two skin states can be characterised to develop a biophysical profile.

4.2 Aims and objectives

The aim of this study was to characterise normal and callused plantar skin. The objectives were to measure the hydration, stiffness and topographical properties of these skin sites in a large sample of individuals and use this data to create a biophysical profile.

4.3 Materials and methods

Ethical approval for this study was granted by the University of Salford's College Research Ethics Panel (application number HSCR12/55).

4.3.1 Subjects

Participants were recruited via a poster campaign on the University campus including the Podiatry Clinic and an advert in a local newspaper. Participants who had a plantar forefoot callus were recruited provided they did not display any of the following exclusion criteria which could potentially affect the skin's properties: compromised cardiovascular or neurological status, connective tissue disorders (such as lupus erythematosus), diabetes, autoimmune disorders (such as rheumatoid arthritis), peripheral vascular disease, or wounds/ ulcers of the legs and feet, and eczema, psoriasis or other dry skin disorders affecting the plantar skin.

The foot was assessed by a podiatrist registered with the HCPC. Pulses from the posterior tibial and dorsalis pedis arteries were assessed by palpation. Presence of neuropathy was assessed by vibration perception using a 128MHz tuning fork and 10g monofilament test on the hallux, first and fifth metatarsal heads. If these tests, skin assessment and medical history were satisfied, the volunteers were recruited to participate in the study.

4.3.2 Data collection

Data was collected in a single session. The foot was allowed to acclimatise for approximately 15 minutes before measurements were taken. Temperature and humidity were monitored but not controlled.

The skin sites measured included: centre of callus, edge of callus, adjacent skin (the distance of the lesion's diameter medial or lateral from the centre of the lesion), and two normal weight bearing skin sites including (1) non-callused skin overlying the plantar metatarsal area (PMA), and (2) adjacent skin overlying the base of the fifth metatarsal

(figure 4.1). The callus centre was chosen because of its location at the centre of the lesion, and could thus be considered the most 'callused' area. The callus edge was chosen because of its potential importance in later longitudinal studies; a change in callus properties post-treatment may potentially become apparent faster at an area of 'less callused' skin, such as the boundary of the lesion itself. Adjacent skin located close to the callus was chosen because of the potential for skin that is normal in appearance to still display callus-like properties due to its close proximity to the lesion, which could thus aid in evaluation of interventions. Normal weight-bearing (PMA) and semi weight-bearing (fifth metatarsal base) skin sites were chosen to compare with callused skin and to give a good profile of normal skin properties.



Figure 4.1 - The measurement sites on the callus include the centre, edge, and skin directly adjacent to the callus; and skin overlying the plantar metatarsal area (PMA) and base of the 5th metatarsal (adapted from Hashmi and Malone-Lee, 2007 p.253).

The skin sites to be measured were marked with a ball point pen. The centre and edge of the callus was marked and the width of the lesion recorded. Generally there is a difference in skin texture and hardness between callused and non-callused skin so the boundary between these areas, the callus edge, can be visualised and palpated. The adjacent skin was positioned 200% of the radius length away from the callus centre. The fourth metatarsal was used as a control site, unless the callus lay over this site, in which

case the first metatarsal head was used. The fifth metatarsal base of the same foot was used as a second control site. Both control sites were identified through palpation and marked. The distance of the radii of the measurement probes were marked over the centre of each skin measurement site to allow accurate probe placement.

Measurements were taken based on manufacturers' instructions but adjusted for the needs of plantar skin testing as follows. 10 measurements were taken per skin site using the Corneometer[®] and one image was taken per site for the Visioscan[®]. For the Cutometer[®], 500mbar of negative pressure was applied. The on-time and off-time were set to 30 seconds each. One cycle of on-time and off-time was used on each skin site.

Single data values for hydration are produced by the Corneometer[®] and a mean was taken from the 10 measurement values obtained from each skin site. The Cutometer[®] produces a range of different values based on stiffness, elasticity and plasticity. Skin distensibility (a surrogate measure of stiffness) was chosen based on the hypothesis that callused skin is stiffer than normal plantar skin. Two variables, *variance* and *homogeneity*, were obtained from the Visioscan[®] images. These were chosen based on the hypothesis that callused skin is rougher than normal skin and is therefore more variable and less homogenous in appearance. The homogeneity parameter is a measure of how homogenous the skin is and measures the combinations of grey levels in the image. The more frequent the grey level combination appears in the image, the lower the homogeneity and the rougher the skin (Courage-Khazaka, 2009b). The variance parameter is a measure of how variable the grey level pixel values are within the image and is higher the rougher the skin is (Courage-Khazaka, 2009b).

4.3.3 Statistical analysis

Statistical analysis was undertaken on SPSS 20.0 and Microsoft Office Excel 2010. Means and 95% confidence intervals were calculated for each skin site. Percentage differences were also calculated to show explicit differences between skin sites.

Percentage difference

$$\% difference = \frac{Skin site 1 - Skin site 2}{Skin site 2} \times 100$$

To assess whether these differences were statistically significant, the repeated measures Analysis of Variance (ANOVA) with post-hoc Bonferroni adjustment, or Friedman's ANOVA with post-hoc Wilcoxon Signed-Ranks Test were used for parametric and non-parametric data distributions respectively. Differences were tested between all skin sites including callus centre, callus edge, skin adjacent to callus, normal PMA skin and skin overlying the fifth metatarsal base.

The repeated measures ANOVA relies on the F-statistic, a value used to assess significance of variance between means, which is calculated by dividing the residual mean of squares by the model mean of squares. To calculate the mean of squares, one must first calculate the within subject sum of squares (SS_W) by multiplying the degrees of freedom by the within subject variance, and the model sum of squares (SS_M) by calculating the difference between the mean of each group and the overall mean. These are then squared, multiplied by the number of subjects and then the values of each group are summed together. Once these have been calculated, the residual sum of squares (SS_R) can be found by subtracting the model from the within subject sum of squares. The mean squares are calculated by the sum of squares divided by the degrees of freedom (df). The model (the variation explained by the model) and the residual (the variation not explained by the model) mean of squares is found by dividing the sums of squares by the degrees of freedom. The F-ratio is found by dividing the model by the residual mean squares. The F-ratio is determined which highlights differences between the groups (Field, 2009).

Repeated measures ANOVA

$$SS_W = \sum_{i=1}^n (x_i - \bar{x}_i)^2$$

$$SS_M = \sum n_i \, (\bar{x}_i - \bar{x}_{grand})^2$$

$$SS_R = SS_W - SS_M$$

 $df_R = df_W - df_M$

$$MS_{M} = \frac{SS_{M}}{df_{M}}$$
$$MS_{R} = \frac{SS_{R}}{df_{R}}$$
$$F = \frac{MS_{M}}{MS_{R}}$$

(Field, 2009, pp.465 - 468)

The Friedman's ANOVA is the non-parametric equivalent of the repeated measures ANOVA and is based on ranks as opposed to actual scores. The data for each condition for each participant is ranked. The ranks of each condition are then summated (R_i) and the Friedman statistic (F_r) is calculated as below. The degrees of freedom and the significance of the test statistic is then given (Field, 2009).

Friedman's ANOVA

$$F_r = \left[\frac{12}{Nk \ (k+1)} \sum_{i=1}^k R_i^2\right] - 3N \ (k+1)$$

(Field, 2009, p.574)

To assess the correlation between skin measurement parameters at each skin site, the Spearman correlation coefficient was used. This statistical test was chosen over Pearson's correlation coefficient because it allows for non-normally distributed data (Field, 2009) which was a feature with different datasets in this study. It is calculated by summing all of the squared differences in ranks of data, multiplied by six and then divided by the number of samples multiplied by the number of samples squared minus 1. This value is then subtracted from 1 and the resulting value is the correlation coefficient (or r value) (Lund and Lund, 2013).

Spearman's rank order correlation

$$r = 1 - \frac{6 \sum d_i^2}{n (n^2 - 1)}$$

(Lund and Lund, 2013)

4.4 Results

In total 51 healthy adults with a total of 61 calluses were enrolled onto the study. Out of the sample, 80% were female. The participants had a mean height of 164cm (\pm 13); weight of 74.9kg (\pm 15.9); and BMI of 27.9 (\pm 5.3). Of these calluses, 16 were consistent with Merriman's grade 1 'no specific callus plaque, but diffuse or pinch callus, or present in narrow bands' and 45 were consistent with grade 2 'circumscribed, punctuate oval or circular, well-defined thickening of keratinized tissue' (Springett and Merriman, 1995, p.207). The following sections detail the skin results by parameter measured.

4.4.1 Skin hydration



Skin Hydration

Figure 4.2 – Mean and 95% CI hydration data for each of the five measurement sites.

Skin at the centre of the callus was 17.8% less hydrated than skin of the callus edge (p = 0.02); 57.0% less hydrated than skin adjacent to callus (p = 0.000); 62.4% less hydrated than skin overlying the PMA (p = 0.000) and 66.7% less hydrated than skin overlying the fifth metatarsal base (p = 0.000). Skin of the callus edge was 47.7% less hydrated than skin adjacent to callus (p = 0.000); 54.2% less hydrated than skin overlying the PMA (p = 0.000) and 59.4% less hydrated than skin overlying the fifth metatarsal base (p = 0.000). Skin adjacent to callus was 12.4% less hydrated than skin overlying the PMA (p = 0.085) and 22.4% less hydrated than skin overlying the fifth metatarsal base (p = 0.062). Skin overlying the PMA was 11.5% less hydrated than skin overlying the fifth metatarsal base (p = 0.451).

4.4.2 – Skin distensibility



Skin distensibility

Figure 4.3 – Mean and 95% CI distensibility data for each of the five measurement sites.

Skin at the centre of the callus was 28.8% less distensible than skin of the callus edge (p = 0.000); 48.7% less distensible than skin adjacent to callus (p = 0.000); 53.3% less distensible than skin overlying the PMA (p = 0.000) and 47.0% less distensible than skin overlying the fifth metatarsal base (p = 0.000). Skin of the callus edge was 27.9% less distensible than skin adjacent to callus (p = 0.000); 34.4% less distensible than skin overlying the PMA (p = 0.000) and 25.6% less distensible than skin overlying the fifth metatarsal base (p = 0.000) and 25.6% less distensible than skin overlying the fifth metatarsal base (p = 0.000). Skin adjacent to callus was 9.1% less distensible than skin overlying the PMA (p = 0.006) and 3.2% more distensible than skin overlying the fifth metatarsal base (p = 1.0). Skin overlying the PMA was 13.4% more distensible than skin overlying the fifth metatarsal base (p = 0.000).

4.4.3 Skin Topography – homogeneity



Figure 4.4 – Mean and 95% CI homogeneity data for each of the four measurement sites.

Skin at the centre of the callus was 0.4% less homogenous than skin of the callus edge (p = 1.0); 7.5% less homogenous than normal skin overlying the PMA (p = 0.000); and 6.3% less homogenous than skin overlying the base of the fifth metatarsal (p = 0.000). Skin at the callus edge was 7.1% less homogenous than normal skin overlying the PMA (p = 0.000) and 5.9% less homogenous than skin overlying the base of the fifth metatarsal (p = 0.001). Normal skin overlying the PMA was 4.9% more homogenous than skin overlying the base of the fifth metatarsal (p = 0.001).

4.4.4 Skin Topography – variance



Figure 4.5 – Mean and 95% CI variance data for each of the four measurement sites.

Skin at the centre of the callus was 3.7% more variable than skin of the callus edge (p = 1.0); 29.6% more variable than normal skin overlying the PMA (p = 0.000); and 19.6% more variable than skin overlying the base of the fifth metatarsal (p = 0.001). Skin at the callus edge was 24.9% more variable than normal skin overlying the PMA (p = 0.000) and 15.3% more variable than skin overlying the base of the fifth metatarsal (p = 0.01). Normal skin overlying the PMA was 13.4% less variable than skin overlying the base of the fifth metatarsal (p = 0.01). Normal skin overlying the PMA was 13.4% less variable than skin overlying the base of the fifth metatarsal (p = 0.323).

4.4.5 Correlations between variables

Table 4.1 shows the correlation coefficient for the biophysical parameters at each skin site measured. Between hydration and stiffness parameters, there was a moderate significant correlation at the callus centre (r = 0.56) and edge (r = 0.55), and a weak significant correlation on skin adjacent to callus (r = 0.33). Between hydration and homogeneity, there was a moderate significant correlation on PMA skin (r = 0.69) and skin overlying the fifth metatarsal base (r = 0.59). There was a moderate significant negative correlation at these

two sites between hydration and variance parameters (r = -0.68 and -0.66 at PMA and fifth metatarsal base respectively). There was also a weak significant negative correlation between stiffness and variance at PMA skin sites (r = -33). At each skin site there was a very strong and significant negative correlation between homogeneity and variance parameters (r ranged between -0.91 to -0.96).

	Callus centre	Callus edge	Skin adjacent to callus	PMA	5 th met. base
Hydration v stiffness	0.56**	0.55**	0.33*	0.25	0.2
Hydration v homogeneity	0.2	0.24		0.69**	0.587**
Hydration v variance	- 0.12	-0.25		-0.68**	- 0.66**
Stiffness v homogeneity	0.18	0.18		0.3*	0.17
Stiffness v variance	-0.14	-0.26		-0.33*	-0.14
Homogeneity v variance	-0.91**	-0.94**		-0.96**	-0.96**

Table 4.1 Correlation coefficient between biophysical parameters at each skin site

Spearman's correlation: r value. * = significant correlation (p < 0.05); ** = significant correlation (p < 0.01).

4.4.6 Differences between callus grades

There was a significant difference in hydration (p = 0.045) at the callus centre between grades 1 and 2 callus (according to Merriman's classification system). There were no significant differences identified between grades for distensibility, homogeneity and variance parameters.

4.5 Discussion

The aim of this study was to define the characteristics of normal and callused plantar skin. The results showed that callus was significantly less hydrated, less distensible (stiffer), less homogenous in appearance, and more variable in appearance (rougher) than normal plantar skin sites. This is one of very few studies that have sought to characterise the biophysical properties of skin as a primary outcome. At the time of writing, there have been only two which tested differences between skin sites using similar measures. Marrakchi and Maibach (2007) studied differences in hydration between skin sites on the face, using the Corneometer[®] and observed that the skin of the neck had significantly higher hydration values than other skin sites. Ryu et al. (2008) measured many mechanical parameters of different skin sites using a Cutometer[®] and found significant differences, particularly in stiffness and elastic parameters, between the face and other skin sites. Other studies have also observed variations between skin sites, but did not test for significance as they were

not a primary outcome measure. Krueger et al. (2011) and Luebberding et al. (2014) observed large variations in mechanical properties between skin sites using the Cutometer[®]; and Firooz et al. (2012) observed variations in hydration and mechanical properties between skin sites, using the Corneometer[®] and Cutometer[®].

The study described in this thesis chapter has added to the work of previous authors by specifically investigating the biophysical properties of foot skin, which no others have attempted to characterise, and in particular, the differences between callus and normal skin. As a measure of skin hydration, two previous studies have used the Corneometer[®] on foot skin (Papanas et al., 2011, Garrigue et al., 2011), and one study has evaluated mechanical properties on the foot using the Cutometer[®] (Hashmi and Malone-Lee, 2007). In this study, the skin hydration values for normal skin are much lower than those reported by Papanas et al. (2011). This study found a mean hydration on normal skin overlying the PMA and the base of the fifth metatarsal of 9.96au and 11.96au respectively which is much lower than 26.55au on non-callused plantar skin reported by Papanas et al. (2011). However they used participants who had diabetes and they measured non-callused plantar skin on the 'centre' of the plantar aspect, which might have meant the plantar arch, which is not fully weight bearing. The results in this chapter might also suggest that individuals who have callus display generally drier plantar skin than those with healthy plantar skin.

Hydration results cannot be compared to those reported by Garrigue et al. (2011) as they only included participants with diabetes who had xerotic skin. It is also not appropriate to compare skin stiffness results with those obtained by Hashmi and Malone-Lee (2007), since they used a 2mm diameter probe and analysed different aspects of the time displacement curve to this study. There are no available studies which have previously measured skin surface topography on plantar skin to which the results here can be compared.

The observations in this study, that callused skin is less hydrated, stiffer, and rougher in appearance than normal non-callused skin sites, were expected given findings of previous studies showing marked histological differences between callused and normal skin. Thomas et al. (1985) found callused stratum corneum was 2-3 times thicker (p < 0.001) than normal plantar stratum corneum (349 versus 123 cell layers thick) and now this study has shown that histological differences are also related to biophysical differences between callus and control sites.

There are several reasons that could explain why callus is significantly less hydrated and stiffer than normal plantar skin. Through immunohistochemistry tests, Kim et al. (2010) attributed callus development to an increase in: Keratins K9 and K14; adhesion proteins CDSN, DSG1 and DSC1; the proteins involcurin, filaggrin, caspase 14, and CaSR. Collectively this would cause an increased rate of proliferation and increased cell cohesion which thus increases the rate of keratinisation (Kim et al., 2010). As a result of increased cell proliferation, the cells may not have enough time to fully differentiate (Thomas et al., 1985) leading to intercellular lipids becoming fragile (Harding et al., 2003). As a result, the skin's barrier function suffers (Wickett and Visscher, 2006) resulting in decreased hydration (and increased TEWL) (Baroni et al., 2012).

It is not clear whether the effect of altered lipid profiles directly affects the stiffness of the skin. In theory, lipids become more fragile due to lack of differentiation (Harding et al., 2003) which might cause weaker bonds between cells. However, as cell cohesion and proliferation is accelerated (Kim et al., 2010), this increased cell turnover might be enough to increase skin stiffness alone through the skin becoming thicker which might overrule any possible weakness caused by fragile lipids. It could be that fragile lipids may not have any effect on stiffness; they might only affect the epidermal barrier function and hydration. As the biochemical triggers involved in callus formation (Kim et al., 2010) probably increase lipid secretion and thus cohesion, this might be enough to increase stiffness even if they are fragile. Hydration may also have an effect on stiffness. The decreased hydration may have the effect of contributing to increased skin stiffness by possibly increasing cell cohesion through retention of corneodesmosomes (Warner et al., 1999, Bouwstra et al., 2003, Wu et al., 2006).

Increased skin roughness in callus, illustrated by decreased homogeneity and increased variance, are a visible effect of the processes above. The scaliness of the skin is likely to be a result of altered desquamation which has been observed by Thomas et al. (1985) who reported that in the most superficial aspect of callused stratum corneum, the rate of cell loss was increased suggesting less tightly bound keratinocytes. It could possibly be due to fragile intercellular lipids as discussed above. While it is not known for sure why this is the case, it does explain why the surface of callused skin can appear scaly in nature.

The devices used in this study (Corneometer[®], Cutometer[®] and Visioscan[®]) have been shown to clearly identify differences between skin sites at a statistically significant level. They have effectively described the properties of callused and normal plantar skin. Correlation analysis has revealed that on callused skin, hydration and distensibility (stiffness) parameters moderately correlate with each other showing that there when skin becomes less hydrated, it becomes stiffer. There were also moderate correlations between the roughness parameters and hydration at non-callused skin sites, but not at callused sites. This may suggest that when skin becomes callused, the roughness and hydration changes. While the changes are significant in both cases, they don't closely mimic each other in all individuals. The homogeneity and variance parameters showed a very strong negative correlation to one another which suggests that they are very closely related.

The results of this study could be far reaching in terms of providing targets for treatment evaluation. To the best of the author's knowledge, this has never been done before. But previous literature has used the same skin devices to evaluate treatments. Two studies have been published which have used hydration as a parameter for evaluating treatment. Garrigue et al. (2011) and Papanas et al. (2011) used the same skin hydration device (Corneometer[®]) to evaluate the efficacy of skin moisturising products in xerotic skin. The studies reported hydration increases of up to 57.3% post-treatment, showing that skin hydration changes to treatments can be detected on the foot using the device. There are no available studies which have evaluated foot treatments, using the 8mm (or other diameters) Cutometer[®] probe, but treatments on other areas of the body have been evaluated effectively using this method. One such example by Josse et al. (2009) used the Cutometer[®] to assess mechanical properties in arm skin treated with a topical corticosteroid cream. The device was able to detect a 30% pre- and post-treatment change in stiffness. Like the Cutometer[®] device, the Visioscan[®] has never been used to evaluate foot skin treatments, but it has been used on other sites. The best example involved the evaluation of different moisturising agents applied to the face. The device was able to detect statistically significant differences over the 28 day study period (mean values and % differences not available) in pre- and post-treatment roughness parameters (Pena Ferreira et al., 2010).

The results presented in this chapter can be used as a guide for what magnitude of skin change, as a result of treatment or intervention, can be expected or aimed for using the skin devices. For example, if one were to aim to improve callused skin hydration to the point where it were 'less callused', a target of 17.8%, which would improve the central callused skin to same hydration as the callus edge, might be appropriate as a short term goal. Likewise, an increase of 62.4% hydration, the increase required to equal the hydration of normal PMA skin, might be a reasonable long-term target. As with the skin hydration, the distensibility and skin roughness data might be used in the same way, with differences between skin sites acting as a guide. Using these devices together gives an all-

round profile of the skin and can therefore allow clinicians and researchers to draw inferences about the data. For instance, if callused skin were to show characteristics like normal PMA skin after an intervention, one could suggest that the intervention has had the effect of reversing some of the callus biochemical changes. For example the proliferation of cells has sufficiently slowed down, leading to less thickened skin, and allowing more time for keratinocyte differentiation, leading to improved lipid profile and an improved barrier function; thus greater hydration (Thomas et al., 1985, Kim et al., 2010, Baroni et al., 2012).

In reality, however, the situation is not as simple as this. Basic treatments such as moisturisers tend to treat the symptom -i.e. stiffness, dryness, roughness, which may lead to discomfort – as opposed to the cause of the callus. The superficial skin will be reduced using a scalpel, and perhaps softened and hydrated using a moisturiser. While there may be some temporary effect on the biochemical process of hyperkeratosis, the underlying mechanism is likely to remain. If the treatment stops, it is likely that callused skin will return, as previous research has documented (Potter, 2000). This mechanism is hypothesised to be plantar pressure and the evidence links to this (Bevans and Bowker, 1999, Abouaesha et al., 2001, Menz et al., 2007). The following chapters of this thesis will aim to use the new knowledge gained from the study in this chapter to try to change the properties of the skin biomechanically. The next study will aim to increase loads beneath normal plantar skin, and use the skin measurement devices to evaluate whether or not the normal skin becomes 'callus-like', i.e. whether it shows signs of reduced hydration, distensibility, and homogeneity, and increased variance, which would point to hyperkeratosis. The final study will use the devices to determine whether reducing pressure beneath existing calluses leads to skin properties similar to, or beginning to change in the direction of, normal PMA skin. If the intended skin changes in these two studies occur, it can be argued that alterations in the biomechanics of the skin have had an effect on the biochemical process of hyperkeratosis. Conversely, if no changes occur, the opposite can be argued.

The main limitation to this skin characterisation study is that means of visualising an actual cross-section of the skin were not attempted. Ideally, histological analysis of skin at each of the measurement sites could have been used to greater link the biophysical properties to the cellular differences between callused and normal skin. However, this would not have been well tolerated by participants because it would have involved skin biopsy. This could introduce pain, the risk of infection, and accidental injury to the participant or investigator, and consequently result in a large reduction in participant numbers. Also, because this research is intended to be applied to the clinical and research settings in analysing the effects of skin treatments and interventions, performing biopsies at follow up appointments would be inappropriate. Pain and inflammation which might result from skin punctures, will almost certainly have an effect on gait (Mickle et al., 2010), and thus pressure distribution – an important area of investigation in the context of this thesis. Inflammation could also inadvertently affect the skin properties. Thus, while a limitation to this study, withholding the use of invasive measures was an appropriate decision.

One other limitation of this study is that the majority of the sample (80%) was female, which could potentially have affected the data. Previous, large sample studies have shown gender differences on non-plantar skin sites such as the face and forearm. Differences have been noted in TEWL and skin hydration between genders. Young men have been shown to have more hydrated skin than young women, and a lower TEWL. However, the TEWL of both genders becomes more similar in fifth decade of life, and men's skin loses hydration and women's skin remains similar or increases in hydration depending on skin site (Luebberding et al., 2013). Similar differences were found in mechanical properties, with women shown to have significantly stiffer but more elastic skin than men up until the fourth decade, where the differences begin to diminish with increased age (Luebberding et al., 2014). However, the agreement between studies is not consistent with some showing no gender-related differences in hydration or mechanical properties (Man et al., 2009, Firooz et al., 2012). Gender differences in other parameters such as sebum level and PH have also been reported but vary between studies (Man et al., 2009, Luebberding et al., 2013). Nevertheless, the available research suggests that gender differences do exist, so the possibility of having a population with 80% females might affect the results. Future plantar skin characterisation studies might benefit from gendermatching their participants to minimise gender bias affecting data.

A control group of individuals who are age- and gender-matched without callus may have been beneficial to include in this study and would have allowed comparisons of the general skin quality between individuals with and without callus. This could give valuable information about individuals who might be at risk of callus development, based on their skin properties, and would be an interesting area to research.

This study was undertaken as part of a larger portfolio of work investigating the biophysical properties of callus, corns and heel fissures, and the characterisation data was
thus used to evaluate treatments to these lesions. At the time of writing, the characterisation portion of the project has now been published in the Journal of Foot and Ankle Research (Hashmi et al., 2015a). Two other studies, one evaluating topical and podiatry treatments for callus (Hashmi et al., 2015, under review) and heel fissures (Hashmi et al., 2015, under review) have been submitted for publication and are both currently under review. These studies showed large increases in hydration (increases ranged from 46% - 152%) and distensibility (increases ranged from 15% - 100%) in callused skin sites between baseline and post-treatment were possible in podiatry and topical treatment groups. There were no significant changes in the control site across the study. Such a large difference suggests that the instruments used in this study are effective in tracking callused skin longitudinally through periods of intervention, and the techniques will be employed to track skin through biomechanical changes in the remaining studies in this thesis.

4.6 Conclusion

This chapter demonstrates development of new knowledge through an original research study in an area that has never been investigated – the characterisation of foot skin, and namely plantar callus. This is despite management of normal and callused skin sites being fundamental to long standing areas of podiatry practice across the world. The importance of the data presented here is therefore potentially very high. Using the same skin measurement techniques evaluated in the previous chapter, this chapter has, for the first time, reported the biophysical properties of callused skin in relation to normal plantar skin. Callused skin is less hydrated, stiffer and rougher in appearance than normal plantar skin and reflects the biochemical changes associated with callus. This information could potentially be useful in the clinical setting, where treatment approaches can be quantitatively evaluated and the ability to return callused skin to normal can be judged. It could also be useful in assessing skin which might be becoming 'callused' as a result of exogenous insults, such as a poorly fitting shoe or bony prominences. In the clinical setting and research, this information will be very valuable in evaluating skin treatments. In the context of this thesis, it is valuable as the effect of applying load to normal skin can now be quantitatively evaluated using the properties of callused skin as a target for effects of increasing/decreasing load.

Chapter 5: Development of a device to apply external loads to plantar skin

5.1 Introduction

The previous chapter (Chapter 4) has demonstrated that callus is significantly less hydrated, less distensible (stiffer), less homogenous and more variable (rougher) than normal plantar skin. Now that a biophysical profile of plantar skin has been constructed, the role of external load in the development of callus can now be investigated. Chapter 2 identified that there is a hypothesised association between callus and external loads applied to skin, but this link has never been explicitly studied systematically. The specific unresolved issue is whether increased plantar pressure causes callus; research on other skin sites suggests that increasing external load leads to callus creation. Despite the fact callus is more common on the foot than any other area of the body, all prior work has been on sites other than the foot. The purpose of the next two chapters is therefore to analyse the effect on plantar skin properties of increases in external load. To achieve this in controlled laboratory conditions, i.e. apply increased loads systematically and in a safe manner, a device was required to apply loads on living subjects. The outline idea was to apply invivo like pressures to a single site under the foot, in repetitive cycles over a period of several weeks. This chapter will describe the development of a device to apply loads to the plantar skin, focussing particularly on the target loads to be applied and the design and repeatability of the device. The subsequent chapter describes the application of this device in an experimental study.

5.2 Determining the forces and pressures to be applied by the loading device

A device was designed to apply external loads to plantar skin in a controlled way and mimic, as far as was possible, in-vivo and gait like loading conditions. As discussed in Chapter 2, during gait the foot is subject to compressive (vertical) loads and shear (horizontal) loads. The vertical loads reach a peak of around 1.2 times bodyweight while the shear loads generally reach 0.2 times bodyweight in both the posterior and anterior directions as a result of acceleration and deceleration (Richards, 2008). These loads are applied over the metatarsal heads resulting in the well reported plantar pressure patterns in the literature. Ideally the device for this study must be able to apply pressures similar to those experienced in gait in terms of three key parameters (1) magnitude of pressure, (2)

timing of pressure application, and (3) direction of pressure application. In order to do this, adequate force must be applied over a specific contact area at an appropriate rate to replicate in-vivo vertical and shear plantar pressures. The device would ideally apply both compression and shear pressure so that the relationship between skin properties and these forces can be investigated. As was discussed in Chapter 2, however, shear forces and pressures are thought to be much smaller than vertical compression forces and pressures. Also, there is far better data on vertical pressures upon which to design a device to apply additional loads.

To estimate how much force might be necessary to apply pressures that are realistic, two studies that reported normal pressure over each of the metatarsal heads using a pressure mat (Novell Emed[®]) were selected to provide target vertical plantar pressure data (Bryant et al., 1999, Putti et al., 2008). These studies were selected due to the fact that they are the only studies identified that reported pressures over each of the metatarsal heads separately and did so in large groups of healthy participants. Bryant et al. (1999) and Putti et al. (2008) both selected healthy subjects (n = 30 and 53 respectively) without the presence of any foot deformities, previous surgery, pain or any factors that might affect their gait so their data was as representative of 'normal' as possible. Also, and in the context of the research questions being addressed, these studies reported barefoot pressures, and were deemed to be more helpful than in-shoe pressure data (such as Pedar[®]) since they are greater and therefore would set higher target pressures.

However, since plantar pressures are person specific and plantar skin likely preconditioned to the pressure normally applied, then the target pressures applied by the device should be adjusted on a participant by participant basis. This would be necessary to ensure that the pressures applied were suitably towards the upper limits (or outside) those normally experienced by the foot and the skin of the foot, and thus more likely to stimulate some change in the skin. The assumption here was that, to investigate whether callus-like changes in skin occur due to external loads, the loads applied would have to be in addition to those normally experienced and thus registered by the body as 'a material increase in pressures'. To facilitate this, short doses of very high pressure (i.e. well outside the normal range for the foot) should be applied.

Thus, to inform device design, knowledge of normal peak vertical plantar pressures under each metatarsal head and contact area are important. Normal peak pressures were obtained from the results of the two aforementioned studies and are displayed as N/m^2 (kPa) (Table 5.1). These pressures were then normalised to kg of bodyweight so that they

could be translated into target pressure for research participants (whose weight would be unknown) (Table 5.2). From the peak pressure data, an estimate of how much pressure might be applied can be calculated as follows:

Target pressure applied by the device = normalised plantar pressure \times kg bodyweight of participant.

For example, a target pressure to apply for a 75kg adult might be 406.86kPa under the second metatarsal head (5.42 multiplied by 75) and 360.86kPa under the third metatarsal head (4.81 multiplied by 75).

Table 5.1 – Normal pressures under each metatarsal head

	Pressure (kPa)					
	1 MPJ	2 MPJ	3 MPJ	4 MPJ	5 MPJ	Weight (kg)
Bryant et al, 1999	289.6	419.5	362.8	251.1	248.6	70.1
Putti et al, 2008	277	361	330	233	151	74.2

Table 5.2 – Normal plantar pressures per kilogram of bodyweight

	Pressure (kPa) per kg of bodyweight				
	1 MPJ	2 MPJ	3 MPJ	4 MPJ	5 MPJ
Bryant et al 1999	4.131241	5.984308	5.175464	3.582026	3.546362
Putti et al, 1999	3.733154	4.865229	4.447439	3.140162	2.03504
Mean	3.932197	5.424769	4.811451	3.361094	2.790701

To calculate an approximate shear pressure that might be experienced under the metatarsal heads, and in the absence of quality data of shear pressure, force plate data was extrapolated and used as a guide. Vertical forces applied to the forefoot are approximately 1.2 times bodyweight, and forefoot shear force is approximately 0.2 times body weight – so approximately one sixth of the vertical force (Richards, 2008). Assuming that the ratio between vertical pressure and shear pressure is the same as between vertical forces and shear forces (merits of this are discussed later), this 1/6 relationship can be used to provide a general approximation of shear pressures the device might apply:

$$Target shear pressure = \frac{target vertical pressure}{6}$$

So 406.86 kPa divided by six gives an approximate shear value of 66.72 kPa for the second metatarsal head of a 75kg individual.

To calculate a suitable contact size, work was undertaken to allow a crude measure of metatarsal head diameters, using Vernier calipers and ultrasound, finding that second metatarsal diameters might reach 14mm. A composite, circular contact pad of 15mm diameter (0.000177m²) was chosen as a starting point (although this could be changed later). This would allow enough surface area to compress skin overlying the metatarsal heads, without being so large as to cause some offloading by adjacent metatarsals. To calculate the force the device needed to apply, to achieve the target vertical and shear pressures, the contact area would be used as follows:

Force $(N) = target \ pressure \ (Pa) \times contact \ area \ (0.000177)$

The process above allowed participant-specific plantar and shear pressure to be determined at, or close to, the likely in-vivo levels. However, for the purposes of attempting to induce callus-like skin changes over a time frame that was realistic (i.e. did not over burden participants by requiring a study lasting several months), it was assumed that pressures applied should be above those normally experienced by the participants' feet. While callus is associated with approximately 35% increase in pressure, individuals will be subjected to these pressures during each step they take during the day. As it would only be feasible to apply additional loads at suitably high levels on a regular basis (e.g. daily) to participants for a short window of time (e.g. a few weeks) the initial specification for the load to be applied by the device would need to allow vertical pressures to be two to three times those experienced in vivo (under the second metatarsal head). For the example for a 75kg person the target vertical pressure (three times normal) might therefore be 1,220.58 kPa compression and 200.16 kPa shear. For a heavier individual, of say 90kg, this would be 1,464 kPa compression and 244.11 kPa of shear. Using a 15mm diameter contact pad, a compression force would be required of up to 215.6N and 258.6N for 75kg and 90kg individuals respectively. Likewise, shear force would need to be 35.4N and 43.1N respectively. Through this process a general specification for the device and components (i.e. actuators) delivering the load to the feet was derived.

5.3 Design of loading device

5.3.1 Components

A composite drawing of the device design is shown in Figure 5.1. Table 5.3 shows the device components. The contact pad (A) that contacts the forefoot is fixed to the top of the shear load cell (B). The shear load cell is attached to a slide (C) which moves superiorly and inferiorly along two vertical chrome steel shafts. The vertical force is transferred to the compression load cell (D) through the slide, whose inferior movement during loading compresses the load cell which in turn will measure this compression force. The slide attaches to a bracket (E) that is fixed to the shear actuator (F) via a mounting plate (G). The shear actuator sits on top of the compression actuator is fixed via fixing brackets (J) to a base plate (K).



Figure 5.1 – Anatomy of loading device. A: contact pad. B: shear load cell. C: slide. D: compression load cell. E: bracket. F: shear actuator. G: mounting plate. H: compression actuator. I: connecting plate. J: fixing brackets. K: base plate.

0 4	G 1		
Component	Supplier	Part number	Function
Vertical actuator.	SMC Pneumatics	MGPM32-	Vertical movement.
	(UK) Ltd.	25A.	
Horizontal actuator.	SMC Pneumatics	CXTM16-	Horizontal movement.
	(UK) Ltd.	25B.	
Compression load cell.	Applied	CDFM3-	0-500N measurement range.
_	Measurements.	500N.	
Shear load cell.	Applied	OBUG-20kg.	0 - 200N measurement range.
	Measurements.		
Load cell digitiser x 2.	Applied	DSC-USB.	Supplies power to the load cell and
_	Measurements.		digitises the millivolt signal from it.
			Allows user to display and log the force
			values.
Pressure regulators x 2.	SMC Pneumatics	AW20-	Regulation of air pressure to cylinders.
_	(UK) Ltd.	F02BCE.	
Solenoid valves x 2.	SMC Pneumatics	SY5120-	Control of actuator movements.
	(UK) Ltd.	6LOU-C6F-	
		Q.	
Limiter switch x 2	SMC Pneumatics	D-M9BL.	Detects actuator movements.
	(UK) Ltd.		
Base plate, support;	Ryder and Wallace	SGCN 0433-	Aluminium supporting structure for
mounting and sliding	Ltd.	13A to 21A.	device.
brackets; contact pad.			
8mm diameter x 500mm	Hepco Motion.	NIM08-500.	Chrome steel shafts – elastic modulus
long ground shaft			200 GPa.
			Used in slide for movement of shear
			load cell to transfer load to compression
			cell.
Computer control box.	Buswell Machine	N/A.	Control of loading sequences. Drives
	Electronics Ltd.		solenoid valves.

Table 5.3 – Details of main device components.

The target pressure, force and contact area data described above were used to assist in the development of the device components. These data were a starting point rather than a completed technical specification since there was no previous attempt to do this type of research documented, and thus some unseen challenges were expected in both the device and experimental protocol in its use. In order to deliver forces in both a vertical and horizontal direction, in a controlled and cyclical manner, the decision was made to use actuators powered by pneumatic cylinders. This was deemed a suitable and reliable alternative to using a manually driven device, as had been used in previous projects in the same department (Hashmi et al., 2013). The use of actuators would give a repeatable and adjustable means of applying forces to the foot at a safe magnitude. In order to have optimum control of the forces delivered to the foot via a probe impacting the plantar surface, pressure regulators were fitted to both pneumatic cylinders to allow adjustment of the amount of air pressure delivered to them, and therefore the force resulting from the resistance of the foot. A Jun-Air[®] Quiet Air 6-15 air compressor base unit was used to deliver pressure to the actuators. This system delivers a maximum of 8 bars (120 PSI) of pressure and has a 15 litre air tank which makes it ideal for delivering a large range of air pressure values. Velocity of the actuator movements was controlled by air flow restrictors located on the actuators.

It was necessary to have a method of measuring the amount of force delivered so that the air pressure to change the resultant load could be adjusted accurately for each study participant and target plantar pressures achieved with suitable accuracy and repeatability. The use of load cells that could be attached above the actuators and below the contact pad to measure the amount of force delivered by the actuators was deemed a suitable approach. The types of load cells required needed to reflect the types and range of the loads applied. A strain gauge button load cell with a maximum working value of 500N was chosen for compression measurement and a strain gauge single point load cell with a maximum working magnitude of 200N was chosen for shear measurement. Given the contact area of 0.00017m², these sensors could measure up to 2,830.86 kPa of compression and 1,132.34 kPa; 2.3 and 5.6 times the expected maximum compression and shear pressures applied respectively. The compression load cell measures the vertical load applied directly to the cell, while the shear cell, which was positioned longitudinally, measures the amount of horizontal force delivered causing a bend through the strain gauge within the cell. Having developed an outline concept of what the device was required to achieve, the author consulted engineering design support to agree a technical specification, parts list, assembly schedule, and commissioning process. This allowed conversion of device specifications into practicality.

5.3.2 Loading programme sequences

Once the device was built, the issue of how to deliver the loading cycles was advanced. As the device is powered by pneumatic cylinders, there needed to be a system of switches and pneumatic control valves to enable the device to perform movements by allowing and ceasing the air delivery to the pistons controlling the actuators. This could either be performed manually by the investigator, or a series of sequences for loading cycles could be programmed into an electronic computer. For the purposes of efficiency, the latter option was chosen. For the control computer, some basic control functions were needed including: a programme selection; a stop and reset command; a cycle count display; a pause and continue function; and a cycle step function. The sequences that were chosen were based on possible methods of load application that the investigator would use on the foot. The basic load programme sequences are shown below.

- 1. 'Compression' programme: pressure applied to move contact pad vertically to contact the forefoot, then apply pressure in reverse direction to retract the contact pad to start position (the number of vertical movements is selected manually before intervention).
- 2. 'Compression and shear' programme: pressure applied to move contact pad vertically to contact the forefoot; pressure applied to move contact pad forwards; pressure applied in reverse direction to move contact pad backwards (the number of forwards and backwards movements is selected manually, and total number of movements is completed before next step); then apply pressure to retract the contact pad downwards, with horizontal actuator set in starting (backwards) position.
- 3. 'Combined' programme: pressure applied to move contact pad vertically to contact the forefoot; pressure applied to move contact pad forwards; pressure applied to retract the contact pad downwards; pressure applied to move contact pad backwards to starting position (all movements are one repetition, the number of repetitions is preselected).
- 4. 'Gait simulation' programme: pressure applied to move contact pad vertically to contact the forefoot; pressure applied to move contact pad forwards; pressure applied in reverse direction to move contact pad backwards; then apply pressure to retract the contact pad downwards to starting position (all movements are one repetition, the number of repetitions is preselected).

The following paragraphs show a breakdown of the actual sequences for each loading cycle which were based on predefined displacement profiles that were programmed into the device computer. Figure 5.2 shows the positions of the limiter switches which are instrumental in control of the device movements that are explained below.



Figure 5.2 – Diagram showing limiter switch positions.

Start conditions:

- Each actuator is controlled by single acting spring return pneumatic 12V solenoid valve.
- Compression actuator retracted, Limit A closed and Limit B open.
- Shear actuator retracted, Limit D closed and Limit C open.
- Select Programme 1, 2, 3 or 4.
- Set number of cycles.
- Set delay on compression actuator in 'up' position.
- Set delay on compression actuator in 'down' position.
- Set delay on shear actuator in 'out' position.
- Set delay on compression actuator in 'in' position.

Compression programme sequence:

In this sequence, the skin is compressed in vertical motions where the contact pad contacts and leaves the skin for a set number of cycles.

- 1. Shear actuator locked at 'in' position and Limit D closed (Limit C open).
- 2. Energise solenoid valve controlling compression actuator.
- 3. Actuator moves up and Limit B closes (Limit A opens).

- 4. Programmed time delay.
- 5. De -energise solenoid valve controlling compression actuator.
- 6. Actuator moves down and Limit A closes (Limit B opens).
- 7. Programmed time delay.
- 8. Repeat steps 3 to 8 for required number of cycles.
- 9. Test completed, return to start state conditions.

Shear programme sequence:

- 1. In this sequence, the contact pad moves anteriorly and posteriorly on the skin for a set number of cycles without retracting.
- 2. Energise solenoid valve controlling compression actuator.
- 3. Actuator moves up and limit B closes (limit A opens).
- 4. Energise solenoid valve controlling shear actuator.
- 5. Shear actuator moves out and limit C closes (limit D opens).
- 6. Programmed time delay.
- 7. De-energise solenoid valve controlling shear actuator.
- 8. Actuator moves in and limit D closes (limit C opens).
- 9. Programmed time delay.
- 10. Repeat steps 3 to 8 for required number of cycles.
- 11. De-energise solenoid valve controlling compression actuator.
- 12. Actuator moves down and Limit A closes (limit B opens).
- 13. Test completed, return to start state conditions.

Combined programme sequence:

In this sequence, the contact pad compresses than moves anteriorly on the skin for a set number of cycles. The contact pad then retracts and returns to the start conditions at the beginning of each cycle.

- 1. Shear actuator locked at 'in' position and Limit D closed (Limit C open).
- 2. Energise solenoid valve controlling compression actuator.
- 3. Actuator moves up and limit B closes (limit A opens).
- 4. Programmed time delay.
- 5. Energise solenoid valve controlling shear actuator.
- 6. Shear actuator moves out and limit C closes (limit D opens).
- 7. Programmed time delay.

- 8. De-energise solenoid valve controlling compression actuator.
- 9. Actuator moves down and Limit A closes (limit B opens).
- 10. Programmed time delay.
- 11. De-energise solenoid valve controlling shear actuator.
- 12. Actuator moves in and Limit D closes (Limit C opens).
- 13. Programmed time delay.
- 14. Repeat steps 3 to 14 for required number of cycles.
- 15. Test completed, return to start state conditions.

Gait simulation programme sequence

In this sequence, the skin is compressed, the contact head moves anteriorly and posteriorly on the skin, and then retracts. This sequence is repeated for the desired number of cycles.

- 1. Shear actuator locked at 'in' position and limit D closed ((limit C open)
- 2. Energise solenoid valve controlling compression actuator
- 3. Actuator moves up and limit B closes (limit A opens)
- 4. Programmed time delay (U).
- 5. Energise solenoid valve controlling shear actuator.
- 6. Shear actuator moves out and limit C closes (limit D opens)
- 7. Programmed time delay (O).
- 8. De-energise solenoid valve controlling shear actuator.
- 9. Actuator moves in and limit D closes (limit C opens).
- 10. Programmed time delay (I).
- 11. De-energise solenoid valve controlling compression actuator.
- 12. Actuator moves down and limit A closes (limit B opens).
- 13. Programmed time delay (D).
- 14. Repeat steps 3 to 14 for required number of cycles.
- 15. Test completed, return to start state conditions.

To ensure participant safety, the device computer had a preset function which could stop movements of the actuators mid-cycle. The power to the device could also be cut which caused the device to instantly retract downwards from the test subject's foot.

5.3.3 Design of device housing

A stage constructed of 25mm thick plywood was built to house the loading device and prevent any contact with the moving parts which could potentially cause injury to the participant (figure 5.3). This construct also enables the subject to stand above the device which would then apply loads vertically to the plantar aspect of the forefoot. Railings were attached to the stage to assist the participant in stepping onto it and for maintaining balance while having their foot loaded in a standing position. Holes in the sides of the stage were cut to allow the investigator access to the device in order to adjust any settings; and to allow piping and cables to be connected to external devices including an air compressor, laptop computer, and loading cycle computer.



Figure 5.3 – Device housing showing holes in wall and aperture on platform to allow loading of forefoot.

5.3.4 Device modifications

Various minor modifications were made to the device to improve its function and eliminate issues identified through pilot work. The first issue identified was that during the shear programme cycle (and other programmes using the forwards/backwards motion) the compression load cell would measure double the force magnitude at the posterior (starting) position of the shear movement. Through testing of the load cells and actuator movements, this was deemed to be due to compliance occurring through the device which caused excess compression on the compression load cell during the backwards movement. The compliance of the device was very small but given the low deflection range of the load cell, this was found to be too large. In an attempt to eliminate this, the compression cell was moved to a position beneath the shear load cell to reduce the effect of bending on force measurements; and the device bearings were replaced with larger shafts and secured tightly to increase the device stiffness, eliminating any play (Figure 5.4). This reduced the deviation in compression forces by around 20% but this was deemed to be insufficient. A bending moment was still occurring so the next step was to increase the device stiffness further. This was achieved by replacing the bracket, mounting plate and slide components (figure 5.1 C, E and G) with two pieces of thicker metal and making the shafts thicker, which helped to further eliminate compression force discrepancies (figure 5.4).



Figure 5.4 – Replaced components.

In order for the device to be used in biomechanics research, it must produce repeatable and consistent loads at different force magnitudes. The accuracy and repeatability of the loads applied should be tested using materials (e.g. wood and EVA) rather than the foot in the first instance as it would avoid potential safety concerns such as overloading.

5.4 Compression and shear load tests

5.4.1 Rationale and aim

Before any work using the device was conducted, a thorough test of the compression and shear load cells was required to ensure that load measurements were accurate, and to understand if any relationship existed between the compression and shear force measurements when applying only compression loads to the device. The aim of these tests was to test the accuracy of the load cells with different combinations of loads being applied with the device in different positions.

5.4.2 Materials and methods

Four 5kg mass Olympic weights discs, a hanging base weight (1kg) with one 1kg disc and three 2kg discs to make a total of 8kg of mass, a 0.1kg plate and a pulley system designed and built by the author of this thesis were used in these tests. At each test, the data logging software measurements were zeroed before any loads were applied. Loads were then applied to the horizontal pulley system to test the shear load cell and atop the contact pad to test the compression load cell. A small 0.1kg disc was used on top of the contact pad to provide a platform for the weight discs for the vertical cell. The device was set up so that the position of the contact pad was as close to the central longitudinal axis as possible. This setup was used for the remainder of the project. Shear loads were applied in the posterior direction with incremental increases in load on both compression and shear cells. This was conducted with the shear actuator in its backward (or 'in') position, then with the shear actuator set in the forward (or 'out') position. This aimed to test whether the position of the device in the antero-posterior direction had any effect on the load cell measurements. These tests were then conducted but with shear being applied in the anterior direction. This aimed to test whether there were any major differences in load cell measurements when the loads were applied in the opposite direction. At each magnitude of horizontal mass applied (0kg, 2kg, 4kg, 6kg, 8kg), vertical masses were increased from 0 to 20kg in 5kg increments so there was a matrix of 25 combinations tested.

5.4.4 Results

The results of the load cell tests are presented below. Compression and shear load cell measurements were obtained for posterior directed shear in the 'in' and 'out' position (Figures 5.5 - 5.8) and for anterior directed shear in the 'in' and 'out' position (Figures 5.9 - 5.12).



Shear in posterior direction, device locked at 'in' position

Figure 5.5 - Compression cell measurements: posterior shear, device locked at 'in' position



Figure 5.6 – Shear cell measurements: posterior shear, device locked in 'in' position

Shear in posterior direction, device locked in 'out' position



Figure 5.7 - Compression cell measurements: posterior shear, device locked in 'out' position



Figure 5.8 – Shear cell measurements: posterior shear, device locked in 'in' position



Anterior shear, device locked at 'in' position

Figure 5.9 - Compression cell measurements: anterior shear, device locked at 'in' position



Figure 5.10 – Shear cell measurements: anterior shear, device locked at 'in' position



Anterior shear, device locked at 'out' position

Figure 5.11 - Compression cell measurements: anterior shear, device locked at 'out' position



Figure 5.12 – Shear cell measurements: anterior shear, device locked at 'out' position

5.4.5 Discussion

At each weight increment, the compression and shear cell measurements were very consistent. When shear is applied in both the posterior and anterior direction, there appeared to be minimal influence of vertical load increase on shear cells measurements. The compression load cell was more sensitive to increases in shear loads, particularly in the anterior direction. The compression cell measurements increased at 0kg of compression loads when the shear loads were applied. This is likely due to the load cell being offloaded slightly in the posterior direction and increased slightly in the anterior direction; a product of the device design where there may be a slight bending moment with the introduction of horizontal loads that affects the compression cell measurement. When shear loads were increased the compression cell measurements tended to show a reduction in measured compression force at 5kg, 10kg, 15kg, and 20kg, particularly when shear was applied in the anterior direction. The reasons for this are unclear. There appears to be no influence of the position of the device, whether in the 'in' or 'out' position, on the load cell measurements. This indicates that play in the shear actuator does not cause discrepancies in data. This means that the device can remain attached to the shear actuator.

The horizontal loads measured by the shear cell appear smaller than the actual load being applied to the pulley. The difference between the measured load and the actual load becomes consistently larger with increases in horizontal loads applied. This could be due to friction in the pulley wheels taking some of the load. This can be tested by removing the device from the housing and hanging weights directly from the device to see if the loads measured are more accurate. If not, it might indicate that the load cell requires calibrating.

5.5 Shear load cell tests

5.5.1 Rationale and aim

Based on the results from the previous tests, it was necessary to assess whether the shear load cells were measuring loads inaccurately, or whether the discrepancies between measured and actual loads were due to friction in the pulley system. The aim of these tests was to assess shear load cell accuracy with load applied directly to it.

5.5.2 Materials and methods

The top part of the device housing the compression and shear cells was removed from the actuator system and attached to a purpose built platform which allowed weights to be applied directly to the device in order to test the shear cell accuracy.

5.5.3 Results

The results are presented in Figure 5.13.



Figure 5.13 – Results of shear cell tests. Weights applied directly to load cell (blue and red lines) show accuracy of measurement compared to when measurements were obtained using pulley system (green and purple lines).

5.5.4 Discussion

It was found that the shear load cell was working correctly because the loads that were applied to the device were measured accurately. Figure 5.13 compares the force measurements in the anterior and posterior direction with those obtained using the pulley system at 0kg of compression force. It shows a systematic effect of the pulley on the force measurements obtained from the shear cell which suggests friction in the pulley system. In both the anterior and posterior shear directions, the loads applied when using the pulley system are 68% of the actual loads.

5.4.6 Conclusion

Both the compression and shear load cells are able to measure loads accurately and consistently. There is no major influence of compression load on shear measurements, but there is some influence of shear loads on compression measurements. The reason for this is unclear but it may mean that some correction for errors in data might be necessary if using the shear actuator. In terms of compression cycles, there will likely be some shear measured by the shear load cell when applied to foot as the metatarsal head is not flat, but

the results of these tests suggest that any compression and shear measurements obtained from plantar skin are likely to be accurate. However, this process also indicates the need for regular check calibrations to ensure that there are no problems with the load cells.

5.6 Compression tests

5.6.1 Aims and methods of compression tests

The first aim of compression tests were to gain an understanding of the relationship between Bars of pressure delivered to the compression actuator, and the resulting forces (Newtons) applied by the device at the contact pad. This would show whether a linear relationship between pressure in the cylinder and force exists, and thus provide information about how to control and adjust the device through the pressure gauge. This is important in being able to apply person-specific plantar pressures, requires that the forces applied by the device can be adjusted accordingly, and they are in part dependent upon the pressure in the cylinder. The second aim was to determine whether or not the device could produce consistent loads during repeated compression cycles. To achieve the aims, the air pressure to the compression actuator was set at two, three, four, five and six Bars. For each test, the device applied 20 cycles of compression with a 10 millisecond delay in the 'up' and 'down' positions. The tests were delivered on plywood, medium density EVA and skin overlying the second metatarsal head of one subject. The following sections show the results from these tests.

5.6.2 Compression test results on plywood

The results of the compression tests on plywood are displayed in Table 5.4 and Figures 5.14 - 5.23. Table 5.4 and Figure 5.14 show that there is a linear relationship between the pressure applied to the compression actuator and the resulting force applied to the plywood. The raw data graphs for each compression test (Figures 5.15 - 5.23) show that the device delivers consistent loads at different magnitudes on hard material. The withinsession coefficient of variation (CV) of the force peak magnitudes was shown to be very small ranging from 0.4 - 1.9%. Repeatability was also tested for the between-session mean magnitudes at 2, 3, 4, 5 and 6 Bars of pressure with CV data collected for three datasets. The air pressure was cut between each set of compressions so that the results would reflect how accurately the investigator could set the air pressure. The CV between sessions was found to be low ranging from 0.3 - 5.6%.

Pressure	delivered to	Resulting compression load	Vertical pressure (kPa) at contact
actuator		(N)	point
2 Bars		99.90	565.61
2.5 Bars		143.99	815.23
3 Bars		183.73	1,040.23
3.5 Bars		214.42	1,213.98
4 Bars		252.56	1,429.92
4.5 Bars		285.61	1,614.04
5 Bars		318.43	1,802.86
5.5 Bars		354.01	2,004.3
6 Bars		382.69	2,166.68

Table 5.4 – Results of compression tests on plywood



Figure 5.14 – Relationship between cylinder pressure and force delivered by the device to the plywood



Figure 5.15 – 2 Bars compression test data



Figure 5.16 – 2.5 Bars compression test data



Figure 5.17 – 3 Bars compression test data



Figure 5.18 – 3.5 Bars compression test data



Figure 5.19 – 4 Bars compression test data



Figure 5.20 – 4.5 Bars compression test data



Figure 5.21 – 5 Bars compression test data



Figure 5.22 – 5.5 Bars compression test data



Figure 5.23 – 6 Bars compression test data

5.6.3 Compression test results on medium density EVA material

The results for compression tests on medium density EVA are shown in Table 5.5 and Figures 5.24 – 5.33. Table 5.5 and figure 5.24 show that there is a linear relationship between the pressure applied to the compression actuator and the resulting force applied to medium density EVA. The raw data graphs for each compression test (Figures 5.25 - 5.33) show that the device delivers consistent loads at different magnitudes on EVA. The withinsession CV of the force peak magnitudes was shown to be very small ranging from 0.48 – 1.29%. Repeatability was also tested for the between-session mean magnitudes at 2, 3, 4, 5 and 6 Bars of pressure with CV data collected for three datasets. The air pressure was cut between each set of compressions so that the results would reflect how accurately the investigator could set the air pressure. The CV between sessions was found to be low ranging from 0.63 - 4.04%.

Table 5.5 – Results of compression tests on EVA

Pressure	delivered to	Resulting compression load	Vertical pressure (kPa) at contact
actuator		(N)	point
2 Bars		107.80	610.33
2.5 Bars		140.90	797.74
3 Bars		175.86	995.67
3.5 Bars		211.42	1,197
4 Bars		242.89	1,375.17
4.5 Bars		273.48	1,548.37
5 Bars		308.23	1,745.11
5.5 Bars		330.97	1,873.86
6 Bars		359.37	2,034.65



Figure 5.24 – Relationship between cylinder pressure and force delivered by the device to the EVA material



Figure 5.25 – EVA 2 Bars compression test results



Figure 5.26 – EVA 2.5 Bars compression test results



Figure 5.27 – EVA 3 Bars compression test results



Figure 5.28 – EVA 3.5 Bars compression test results



Figure 5.29 – EVA 4 Bars compression test results



Figure 5.30 – EVA 4.5 Bars compression test results



Figure 5.31 – EVA 5 Bars compression test results



Figure 5.32 – EVA 5.5 Bars compression test results



Figure 5.33 – EVA 6 Bars compression test results

5.5.4 Discussion

These compression tests have shown that there is a linear increase in compressive force when the air pressure delivered to the actuator is increased. With the current area of contact head (15mm diameter circular head = $0.000177m^2$) the device may be able to apply compression pressures of over seven times greater than what is normally experienced under the second metatarsal head of a 75kg adult. This means it is more than capable of the load requirements for applying four times normal loads to the skin. The

relationship between bars of pressure applied to the device and resulting Newtons of force can be used to control the loads applied to the skin. The CV values obtained from the force peaks show that the within-session and between-session repeatability is excellent for plywood and medium density EVA.

5.5.6 Conclusion

These tests on plywood and medium density EVA have shown the linear relationship between Bars of pressure delivered to the compression pressure gauge and the resulting Newtons of force that are applied to the material. This information can be subsequently used to estimate the amount of compressive force delivered when adjusting the pressure gauge on the device in future tests on the foot and thus to apply person-specific pressures. This satisfies the first aim of compression tests which were to gain an understanding of the relationship between Bars of pressure delivered to the compression actuator, and the resulting force applied.

The data has also shown that the loads applied at repeated cycles are consistent which satisfies the second aim, which was to determine whether or not the device could produce consistent loads during repeated compression cycles. Collectively, these tests show that there are no issues with how the compression actuator operates, therefore the device has been deemed fit for preliminary skin response to load tests. The amount of pressure applied to the materials was large with up to 2,166.68 kPa of pressure applied with the regulator set to 6 Bars on plywood (over a 15mm diameter circular contact area). While these tests demonstrate what the device is capable of, the pressures applied to plantar skin would sit in the lower range of the device's capabilities.

Chapter 6: A pilot study to investigate whether laboratory controlled plantar loading causes callus-like thickening in healthy skin overlying the plantar metatarsal area

6.1 Introduction

The review of the literature (Chapter 2) discussed the impact that external loads have on skin. In-vivo human (Rubin, 1949, Goldblum and Piper, 1954, Pinkus, 1952, Brophy and Lobitz, 1959) and animal studies (Mackenzie, 1974a, Mackenzie, 1974b, Goldstein and Sanders, 1998, Sanders et al., 2002) have shown how skin adapts to friction, shear and compression loads. Under small to moderate loads over a period of up to 35 days, the skin's mitotic levels increase, and increases in epidermal thickness has been documented. However, these studies have been cited in the callus literature, but in reality cannot be applied to plantar skin, due to the anatomical and functional differences plantar skin has to the areas tested in these papers (e.g. leg, back, mouse ears etc.). Numerous studies have reported increased vertical pressures under callus (Springett, 1993, Pitei et al., 1999, Potter and Potter, 2000b, Abouaesha et al., 2001, Pataky et al., 2002, Menz et al., 2007), which highlights that there is likely to be a relationship between loading of the skin and callus development. However, no work has been done to further address the relationship between loads and callus development. Furthermore, the available evidence is limited in that plantar pressure studies only state the magnitude of pressure beneath the callused skin and not the accumulative increase in load, which is necessary to further understand any link between callus and pressure. This is essential if an understanding of plantar callus, and thus treatment, is to progress. The work outlined in this chapter is the first to ever attempt to cause callus like skin changes using compression pressure application, and to express this in terms of accumulative weekly loads.

Having established that the loading device detailed in Chapter 5 is fit for purpose, protocols for its use in an investigation of whether external loads lead to responses in the skin required development. Whilst some assumptions were already made in chapter 5, questions include the magnitude of pressure to apply, the site over which to apply this pressure, the number of cycles (loading dose) and the number of days per week to apply the loads. This chapter aims to further investigate the relationship between increased plantar loading and callus development by undertaking a pilot study to test skin properties pre and post a period of increased plantar loading (delivered with the loading device
developed in chapter 5). A pilot study was required because load application on plantar skin has never been conducted in this manner before, so feasibility would have to be proven and safety concerns, particularly the potential for biochemical changes leading to callus growth (and the possibility of continued callus development post-study), would have to be determined. The aim of this study was not to create callus, but rather to alter normal PMA skin in the direction of callus using data in Chapter 4 as a guide. The author hypothesises that increases in pressure beneath the metatarsal heads causes a hyperkeratotic response akin to callus development, which is consistent with current opinion (Singh, 1996, Freeman, 2002, Helfand, 2003, Grouios, 2004). It is hypothesised that a decrease in skin hydration, a decrease in skin distensibility, a decrease in skin homogeneity, and an increase in skin variance parameters will result from increased loading at the PMA skin site in comparison to the control sites. If these changes occur, it could be attributed to a callus-like skin response.

6.2 Preliminary tests

A series of short preliminary tests were conducted to develop and test detailed aspects of the protocol to be used in the later pilot study of how increases in plantar pressure might affect plantar skin.

6.2.1 Variability of positioning foot in insole

Foot positioning was tested on four healthy volunteers. The aim was to check the repeatability of positioning a metatarsal head over the loading device contact pad. A SalfordinsoleTM Firm insole was fitted for each volunteer. This insole was chosen because it is made of a rigid plastic and has a well-defined heel cup, which is beneficial for positioning the foot repeatably. The second metatarsal head was identified through palpation and marked using black ink. Each volunteer was then asked to carefully stand on the insole heel first and then slowly place their forefoot down onto the insole, ensuring their foot position corresponded, as closely as possible, to the contours of the insole at the midfoot and heel. The ink mark from the metatarsal head was transferred to the corresponding location on the insole. Using a Stanley knife, a 40mm diameter circle was cut with the ink mark in the centre of the circle which was deemed large enough to allow adequate access for foot marking from beneath while participants stood on the platform. The volunteers were then asked to stand on the platform and the foot was secured to the

platform to limit dorsiflexion. The investigator then used a ball point pen to mark the edges of the cut-out in the insole on the volunteer's foot. This process was performed a total of three times for each participant with a different colour of pen used on each occasion. It was found that a mean of 1.3mm in variation existed in the antero-posterior position and 1.2mm in the medio-lateral position, measured using Vernier calipers. This small amount of positional variation was deemed acceptable for the main pilot study.

6.2.2 Adjustment of plantar pressures applied

Using two participants, a test was performed to observe how accurately loads could be delivered using the pneumatic cylinder pressure regulator controls and sandbags to constrain foot movement. The aim was to establish the ability to fine tune the precise plantar pressure values applied to the foot. Using previously measured in-shoe pressures, a target pressure was set for each participant and the corresponding forces calculated to reach these pressures. The initial load was applied in real-time using the air pressure regulator with the device computer set to deliver a single constant compression load. Once the force applied to the foot was at an acceptable magnitude, a 10 cycle compression dose (at approximately 1Hz) at this setting was applied and the load cell data logged. The process was repeated once more. The results showed that compression cycles could be applied with loads within 10% of the calculated target load.

6.2.3 Duration of pressures applied

Further testing in the same participants above aimed to determine the magnitude of pressure that could be applied without causing discomfort. Using subjective feedback from repeated tests, it was found that applying no more than three times normal pressure in compression cycles would be appropriate. This could be tolerated over a period of 20 minutes (1,200 compressions at 1Hz). However, because this duration was nearing the limit of tolerability, it was determined that any longer duration of load application over a period of weeks could risk unintended discomfort, so limiting the duration to 1,200 compressions was appropriate. Any changes to load application could be done on a participant by participant basis during the study as necessary.

6.3 Pilot study methodology

Following the preliminary tests, a pilot study was conducted to determine whether increased plantar pressures could achieve a skin response akin to callus development, and if so, whether this skin change continues after load application ceases. This is important to determine before any larger-scale study is carried out as there are no other reports of research like this being undertaken in plantar skin. This pilot study is presented as a series of case studies where load applied with the device is expressed within the context of estimated weekly loads at the skin site of interest. As stated in Chapter 2 and earlier in this chapter, expressing applied loads in the context of estimated weekly loads addresses an important omission from the literature, as pressure studies only state the pressure beneath the callused site and not the accumulative effect of this. A case study design allows a more in-depth, meticulous approach which is important because this type of work has never before been conducted. Ethical approval for this study was granted by the University of Salford's College Research Ethics Panel (application number HSCR14/37).

6.3.1 Participant selection

The intended inclusion criteria for this study were individuals over the age of 18 with skin prone to callus (i.e. with callus on some areas of the foot), but with enough non-callused skin over the metatarsal areas to allow a loading and control site, so at least two metatarsal heads free from callus. This inclusion criterion was chosen because having plantar skin prone to callus would provide a better chance of observing a callus-like skin response during the study.

6.3.2 In-shoe pressure measurement

In-shoe pressure data collection in the participants' own shoes using a Novel Pedar-X inshoe pressure system (Novel, Germany) was undertaken to aid in setting the target pressures for the loading device. Pedar insoles have a matrix of 99 sensors (which vary in size according to the size of insole). The in-shoe pressure measurements were obtained using the participants' own shoes that they wore most often. The participants were asked to walk along an eight metre walkway, four times at a self-selected walking speed, which was chosen as it would give the most realistic pressure profile for each participant. Timing gates were used to record speed to ensure consistency in the walking trials.

The pressure data was analysed using InShoe Pressure Analyser[©] version 1.0 (2012), a pressure analysis programme written in the Foot and Ankle Research department at the University of Salford. The code separates the walking trial into blocks of steps. The first and last three steps were then removed as they represent gait initiation and end. The code then takes the peaks from all the sensors in each mask and then means them, so there is a mean peak pressure for each step, and then it gives a mean peak pressure in each mask for the entire walking trial. The masks employed included the first metatarsal head, second to fourth metatarsal heads, fifth metatarsal head, hallux and heel and is based on the work of Bontrager et al. (1997) who used Harris Mat imprints and is therefore anatomically accurate (Figure 6.1). For this study, peak pressure data was collected from the mask overlying the second to fourth metatarsal heads. These heads lie very close together and are therefore difficult to separate using the Pedar[®] sensors, particularly as the area is covered by an array of 16 sensors (4 x 4) and cannot be evenly split into three regions. Because the loading site in this study was only ever the second or fourth metatarsal head, using the Pressure Analyser software, the author equally divided the second to fourth metatarsal head mask into two smaller masks of a 4 x 2 sensor array, and the section overlying the area of interest (the second or fourth metatarsal head) was used. The peak pressure was then used as a guide target compression pressure for the study. Pressures at two to three times normal were deemed suitable because they were tolerable and did not lead to excessive discomfort, which was found when using higher loads in preliminary tests. In addition to peak pressure, pressure time integral (PTI) using the non-zero mean from all the sensors within the mask, and the contact time were also obtained for use later in the study.



Figure 6.1 – Bontrager et al. (1997) Pedar[®] mask used in study using percentages of insole length and width (Chapman, 2014, p.26). Region borders are placed to the nearest whole sensor.

6.3.3 Positioning the foot in relation to the loading device

As described previously, having had the metatarsal head of choice marked with a pen, the participant was asked to carefully stand over the insole heel first and gradually loading the forefoot over the insole to transfer the pen mark to the corresponding location. To ensure the loading device accurately contacted the skin overlying the correct metatarsal head without contacting the insole at full extension, the head of the device which houses the contact pad was removed and used as a stencil. With the pen mark of the metatarsal head on the insole, the aperture where the pin of the contact pad is inserted was used as a guide to visualise the pen mark, then the outer edges of the device head were marked on the insole. This was then cut out using a Stanley knife leaving a 30mm diameter square aperture. This ensured that in full extension, the contact pad of the device would contact the desired area of skin each time the foot was placed on the insole but without making contact with the insole, which would absorb some of the load. The foot was secured to the platform using a sandbag strapped to the dorsal aspect of the foot to limit dorsiflexion and superior movement of the forefoot, as described previously. A short loading test was performed to determine the actual loading point on the foot so as to measure the skin in the

precise area being loaded. Using the imprint from the contact pad of the device, the position was measured and noted, and would serve as the site of interest for all of the subsequent skin tests.

6.3.4 Measurement of skin biophysical properties

During the pilot study, a minimum timeframe of six weeks was chosen. This was deemed appropriate as it would potentially allow at least two plantar stratum corneum cell turnovers (Thomas et al., 1985) and exceeded the timeframe of load application in previous skin studies, optimising the chance of skin change occurring. Testing of biophysical skin properties was performed on the first day of a week prior to any load application and then four weeks after the cessation of the minimum six weeks of load application. The post load measures were therefore taken at a minimum of 10 weeks after the start of the study. Biophysical skin data was collected by one investigator. The skin measures included skin hydration (Corneometer[®]), skin distensibility (Cutometer[®]) and skin surface topography measures of homogeneity and variance (Visioscan[®]). The skin sites measured included the intervention site – non-callused skin overlying the metatarsal head of choice; and two normal weight bearing skin sites including (1) a plantar metatarsal head, and (2) skin overlying the base of the fifth metatarsal. The skin sites were marked with a ball point pen. The first metatarsal was used as a control site unless any callus lay over this site, in which case the fourth metatarsal head was used. The fifth metatarsal base of the same foot was used as a second control site. Both control sites were identified through palpation and marked. The distance of the radii of the measurement probes were marked over the centre of each skin measurement site to allow accurate probe placement (Figure 6.1).

133



Figure 6.2 - The skin measurement sites included the load site (in this case the second metatarsal head) and skin overlying the PMA and base of the 5th metatarsal. A skin site on the opposite foot corresponding to the intervention site was also chosen (adapted from Hashmi and Malone-Lee, 2007 p.253).

Measurements were taken based on manufacturers' instructions but adjusted for the needs of plantar skin testing as follows. 10 measurements were taken per skin site using the Corneometer[®] and one image was taken per site for the Visioscan[®]. For the Cutometer[®], 500mbar of negative pressure was applied over a period of 30 seconds, and skin relaxation time was set to 30 seconds each per skin site, as described in previous chapters.

6.3.5 Activity monitoring

In order to express the applied load to the plantar skin in terms of accumulative weekly load, it was necessary to determine what the weekly load for each participant would be. Therefore, in order to estimate this, activity monitoring was employed. Approximately halfway through the study period, each subject was asked to wear an activPAL^{3TM} activity monitor for a period of seven days to record activity profile. The activPAL^{3TM} is a small device which contains an accelerometer and records data relating to body posture while worn on the anterior aspect of the thigh. The data is then analysed through software which classifies the data into sitting, standing and stepping categories, giving a profile of activity data over the duration which it was recorded. Parameters including stepping, standing and

sitting duration, number of steps, sitting to standing transfers, and cadence can be extracted. The activPALTM has previously been validated for walking and posture activity measurement in adults (Grant et al., 2006, Ryan et al., 2006, Godfrey et al., 2007) and infants (Davies et al., 2012) and has shown good reliability. Inter-device reliability has been shown to be very high (ICC > 0.99) and percentage differences in step number and cadence between device and observation have been reported at less than 1% difference (Ryan et al., 2006). The measurement accuracy between sitting, lying, standing and walking has been shown to be 98% in comparison to an accelerometer (Godfrey et al., 2007). In the study outlined in this chapter, step count, and stepping and standing time were the parameters of interest, best representing the cumulative loads that would be applied to the foot.

6.3.7 Barefoot pressure measurement

In addition to the in-shoe pressure data collected before the start of the study, barefoot pressure data was also collected from each participant during the study period, in order to estimate the accumulative pressure caused by walking barefoot, for example, when at home. The system used in this study was the Novel Emed[®] barefoot pressure platform which has recently shown good intra- and inter-platform reliability with ICC values greater than 0.7 showing high agreement (Hafer et al., 2013). The participants were asked to walk along a six metre walkway at a self-selected speed, measured by timing gates to ensure consistency, facing straight ahead. The platform with the foot of interest while walking normally. This is known as the midgait method and has shown to be very reliable with ICC values exceeding 0.75 with three or more trials used (Hughes et al., 1991, McPoil et al., 1999). Static pressures were also collected for each participant.

The pressure data was analysed using a MATLAB programme written in the Foot and Ankle Research department at the University of Salford. The programme calculates the peak pressure from masks inside the boundaries of the plantar surface of the foot. The mask dimensions and position are predefined and can be visually evaluated on the foot map to assist with measurement accuracy. For this study, a mask of consistent size was positioned over the region of interest for each study participant. In addition to peak pressures, the mean pressure data over each time point in the trial is also calculated within the mask and is exported to Microsoft Excel. This data was used to calculate PTI using the non-zero mean of all the sensors within the mask at the region of interest. Contact time was also calculated in this programme.

6.3.8 Foot loading data

During each loading session the device and foot were set up as previously described and loading programme 1 'Compression' was selected. Setting of the pressure values to be applied to the foot was achieved by first setting the pneumatic cylinder at 0 bars and gradually increasing pressure until the intended force had been reached. Once this was achieved, the device head was retracted and the loading cycles initialised.

The intended loading dose for each participant was 1,200 cycles of compressions at a maximum of three times normal pressure, over three sessions per week over a minimum period of six weeks. The loading dose of a maximum of three times normal pressure at 1,200 cycles was chosen as it was found to be tolerable in preliminary test sessions (section 6.2.3). Three loading sessions per week was chosen as it was the maximum that any of the participants could commit to. Load cell data was collected from the device which gave force and time values throughout the session. From this data the peak pressure, PTI and contact time for each loading session was calculated. The data from each session was used to calculate an overall peak pressure, PTI and contact time per compression for analysis. The activity monitoring data was used so that the loads applied by the device could be contextualised.

6.3.9 Data analysis

The design of this study is based on individual cases so significance tests are not appropriate. For skin data, in addition to plotting the trends from baseline on a line graph, the smallest detectable change (SDC) was calculated to help contextualise changes in skin biophysical data (as described in more depth in Chapter 3). The SDC is calculated from the SEM as follows:

 $SEM = \frac{SD_{Differences}}{\sqrt{2}}$ (de Vet et al., 2006, p.1037)

 $SDC = 1.96 \times \sqrt{2} \times SEM$ (de Vet et al., 2006, p.1038) The step count from the activity data collected from each participant was divided by two to get the number of strides for each foot. The stride number was multiplied by the peak pressure (PP), pressure time integral (PTI) and contact time (CT) of the barefoot and in-shoe pressure data. To estimate the total accumulative peak pressure, PTI and contact time for the week, it was assumed that 10% of walking occurred barefoot with the other 90% shod so this was applied to the barefoot and in-shoe data (0.1 and 0.9 times the total accumulative pressures respectively). These values were then summed for the overall weekly values.

Barefoot estimated PP, PTI or CT per week = $0.1 \times (PP, PTI \text{ or } CT \times Stride \text{ number})$

Shod estimated PP, PTI or CT per week = $0.9 \times (PP, PTI \text{ or } CT \times Stride number)$

Total accumulative plantar PP, PTI or CT per week = Barefoot estimated PP, PTI or CT + Shod estimated PP, PTI or CT

The loading device peak pressure, PTI and contact time per compression were multiplied by the number of compressions (3,600) applied to the foot each week to calculate the estimated average weekly compression dose. The weekly compression data was added to the overall estimated weekly accumulative peak pressure, PTI and contact time to calculate the total load beneath the region of interest over the week. The percentage of this total load applied by the device was calculated for these pressure parameters.

Average weekly compression dose = Average PP, PTI and CT per compression \times 3,600

Total estimated weekly load beneath load site

= Average weekly compression dose of PP,PTI and CT

+ Total accumulative plantar PP, PTI or CT

Percentage of weekly pressure from load device

 $= \frac{Average weekly compression dose}{Total estimated weekly dose beneath load site} \times 100$

For example, an individual with 150 kPa shod peak pressure at the region of interest would have a device peak pressure maximum target of 450 kPa (= 150×3). The device compression dose per week is 1,620,000 kPa (3,600 compressions x 450 kPa). If this person takes 25,000 steps per week on this foot, estimated accumulative shod peak pressure would be 3,375,000 kPa (25,000 steps x 150 kPa x 0.9 [estimated time shod]). Estimated accumulative barefoot peak pressure would be 500,000 kPa (25,000 steps x 200 kPa x 0.1 [estimated time barefoot]). This gives a total accumulative peak pressure of 3,875,000 kPa. The sum of the accumulative device load and the stepping load is 5,495,000 kPa. The load applied by the device therefore accounts for 29.48% of peak pressure load per week.

6.4 Case study – Subject 1

6.4.1 Participant profile

Subject 1 was a 45 year old, white British female with a BMI of 26.2. She was generally healthy, engaged in regular running and weight-lifting activities and had no medical conditions. On examination, bilateral callus beneath the second metatarsal heads and diffuse callus type skin over the first metatarsal heads was observed. It was determined that the most appropriate site for skin loading was the right fourth metatarsal head. The right foot was chosen as there was more 'normal' non-callused skin to provide control measurements. The control sites were the third metatarsal head, as it was callus free; the fifth metatarsal base, a semi weight-bearing site which seldom develops callus; and the fourth metatarsal head on the left foot to provide direct comparison with the loading site on the right foot.

6.4.2 Loading profile

Table 6.1 shows the loading profile for the loading site on Subject 1's foot which includes activity data, dynamic pressure data and pressure data collected from the loading device. It also shows the estimated accumulative dynamic loads and how much of this was

contributed by the loading device. The mean load applied by the device was approximately 2.9 times higher than normal pressure at the foot site.

Activity profile				
Number of strides per week	26,032			
Time stepping (s)	40,320			
Time standing (s)	120,240			
Plantar pressure per step at load	l site			
In-shoe PP (kPa)	117.20			
Barefoot PP (kPa)	138.00			
In-shoe PTI (kPa / s)	21.72			
Barefoot PTI (kPa / s)	24.49			
In-shoe contact time (s)	0.51			
Barefoot contact time (s)	0.44			
Estimated plantar pressure per week a	at load site			
PP (kPa)	3,105,097			
PTI (kPa / s)	572,525			
Contact time (s)	13,039			
Mean plantar pressure applied per device comp	pression at load site			
PP (kPa)	334.72			
PTI (kPa / s)	112.86			
Contact time (s)	0.57			
Mean plantar pressure applied by device per	week at load site			
PP (kPa)	1,204,992			
PTI (kPa / s)	406,297			
Contact time	2,058			
Total weekly plantar pressure expe under load site (% applied with d	rienced evice)			
PP (kPa)	4,310,089			
	(28.0)			
PTT (kPa / s)	978,822			
Contact time (s)	15,097			
	(13.6)			

Table 6.	l Subject 1	lloading	profile
----------	-------------	----------	---------

6.4.3 Skin data

Figures 6.2 - 6.5 and tables 6.3 - 6.6 show the raw data and smallest detectable change (SDC) collected from the skin for hydration, distensibility, homogeneity and variance respectively.



Figure 6.3 – Subject 1 raw hydration data.

Table	6.2 Subi	ect 1	smallest	detectable	change	for h	vdration
					· · · ·	-	

	Load site	PMA	Base of 5 th metatarsal	Opposite foot
Mean difference between time points	3.43	1.91	1.28	3.66
SD of differences between time points	2.16	1.93	1.99	1.41
SEM	1.53	1.37	1.41	1.00
SDC	4.23	3.79	3.90	2.76



Figure 6.4 – Subject 1 raw distensibility data.

	Load site	PMA	Base of 5 th metatarsal	Opposite foot
Mean difference between time points	0.09	0.05	0.12	0.07
SD of differences between time points	0.05	0.06	0.08	0.16
SEM	0.04	0.04	0.06	0.11
SDC	0.10	0.11	0.16	0.31

Table 6.3 Subject 1 smallest detectable change for distensibility



Figure 6.5 – Subject 1 raw homogeneity data.

Table 6.4 Subject 1	smallest detectable	change for	homogeneity

	Load site	PMA	Base of 5 th metatarsal	Opposite foot
Mean difference between time points	0.05	0.00	0.06	0.01
SD of differences between time points	0.05	0.08	0.06	0.05
SEM	0.03	0.06	0.04	0.03
SDC	0.10	0.15	0.11	0.09



Figure 6.6 – Subject 1 raw variance data.

Table 6.5 Subject 1 smallest detectable change for varia	ance
--	------

	Load site	PMA	Base of 5 th metatarsal	Opposite foot
Mean difference between time points	0.64	0.09	0.95	0.12
SD of differences between time points	0.44	0.86	0.75	0.68
SEM	0.31	0.61	0.53	0.48
SDC	0.86	1.69	1.46	1.33

6.5 Case study – Subject 2

6.5.1 Participant profile

Subject 2 was a 44 year old female with ancestry from the Indian Subcontinent with a BMI of 26.3. She did not undertake in regular physical activity and had no medical issues to contra-indicate participation in this study. On examination, bilateral callus beneath the second and third metatarsal heads was observed. The left foot had scaly skin over the metatarsal area, midfoot and heel, but the right foot had normal skin on callus-free areas. The right fourth metatarsal head was chosen as the loading site with controls including the first metatarsal head, which was callus free, fifth metatarsal base, and fourth metatarsal head of the left foot for direct comparison. The subject withdrew from the study after two appointments due to discomfort during the load application cycles.

6.6 Case study – Subject 3

6.6.1 Participant profile

Subject 3 was a 25 year old female of white Greek and Canadian descent. She was generally healthy with a BMI of 22.3, participating in regular physical activity, and had no medical issues to contra-indicate participation in this study. On examination, bilateral callus was observed beneath the second metatarsal heads with the rest of the forefoot showing healthy skin. The right fourth metatarsal head was chosen as the loading site with the first metatarsal head, fifth metatarsal base, and fourth metatarsal head of the left foot serving as control sites.

6.6.2 Loading profile

Table 6.6 shows the loading profile for the loading site on Subject 3's foot which includes activity data, dynamic pressure data and pressure data collected from the loading device. It also shows the estimated accumulative dynamic loads and how much of this was contributed by the loading device. The mean load applied by the device was approximately 2.2 times higher than normal pressure.

Activity profile				
Number of strides per week	46,043			
Time stepping (s)	55,440			
Time standing (s)	90,000			
Plantar pressure per step at loa	nd site			
In-shoe PP (kPa)	186.33			
Barefoot PP (kPa)	192			
In-shoe PTI (kPa / s)	31.96			
Barefoot PTI (kPa / s)	35.04			
In-shoe contact time (s)	0.59			
Barefoot contact time (s)	0.6			
Estimated plantar pressure per week	at load site			
PP (kPa)	8,605,299			
PTI (kPa / s)	1,485,574			
Contact time (s)	27,111			
Mean plantar pressure applied per device con	npression at load site			
PP (kPa)	411.02			
PTI (kPa / s)	144.8			
Contact time (s)	0.57			
Mean plantar pressure applied by device pe	r week at load site			
PP (kPa)	1,479,672			
PTI (kPa / s)	521,262			
Contact time	2,036.82			
Total weekly plantar pressure experienced				
PP (kPa)	10,084,970			
	(14.7)			
PTI (kPa / s)	2,006,837			
	(26.0)			
Contact time (s)	29,148			
	(7.0)			

Table 6.6 Subject 3 loading profile

6.6.3 Skin data

Figures 6.7 - 6.10 and Tables 6.7 - 6.10 show the raw data and SDC collected from the skin for hydration, distensibility, homogeneity and variance respectively.



Figure 6.7 – Subject 3 raw hydration data.

Table 6	5.7 Subjec	t 3 smallest	detectable	change :	for hv	dration

	Load site	PMA	Base of 5 th metatarsal	Opposite foot
Mean difference between time points	10.29	9.19	9.27	6.20
SD of differences between time points	5.03	5.45	1.41	4.61
SEM	3.56	3.85	0.99	3.26
SDC	9.86	10.67	2.75	9.04



Figure 6.8 – Subject 3 raw distensibility data.

	Load site	PMA	Base of 5 th metatarsal	Opposite foot
Mean difference between time points	0.02	0.10	0.05	0.01
SD of differences between time points	0.11	0.07	0.04	0.03
SEM	0.07	0.05	0.03	0.02
SDC	0.21	0.14	0.08	0.06

Table 6.8 Subject 3 smallest detectable change for distensibility



Figure 6.9 – Subject 3 raw homogeneity data.

Tabla	60	Subject 3	smallast	datactable	change	for	homogonai	t x 7
I abic	0.7	Subject 3	Smancsi	utituanit	t Unange	IUL	nomogenei	ιy

	Load site	PMA	Base of 5 th metatarsal	Opposite foot
Mean difference between time points	0.08	0.11	0.13	-0.03
SD of differences between time points	0.05	0.05	0.05	0.07
SEM	0.03	0.04	0.03	0.05
SDC	0.09	0.10	0.09	0.13



Figure 6.10 – Subject 3 raw variance data.

Tahla	6 10	Subjec	f 3 c	mallact	datactable	change	for	variance
Lanc	0.10	Subjec	1 3 3	manesi	ucicciable	unange	101	variance

	Load site	PMA	Base of 5 th metatarsal	Opposite foot
Mean difference between time points	0.79	1.51	1.74	0.07
SD of differences between time points	0.54	0.78	0.58	0.74
SEM	0.38	0.55	0.41	0.52
SDC	1.06	1.53	1.14	1.44

6.7 Case study – Subject 4

6.7.1 Participant profile

Subject 4 was a 23 year old female of white Irish descent. She undertook in regular physical activity including distance running and cycling, and had a BMI of 22.0. On examination, bilateral callus was observed beneath the second metatarsal heads with healthy skin over the rest of the forefoot. The right fourth metatarsal head was chosen as the loading site and the control sites included the first metatarsal head, fifth metatarsal base, and fourth metatarsal head of the left foot.

6.7.2 Loading profile

Table 6.11 shows the loading profile for the loading site on Subject 4's foot which includes activity data, dynamic pressure data and pressure data collected from the loading device. It also shows the estimated accumulative dynamic loads and how much of this was contributed by the loading device. The average load applied by the device was

approximately 1.5 times higher than normal pressure. Loads higher than this could not be tolerated.

Activity profile									
Number of strides per week	34,211								
Time stepping	35,169								
Time standing	103,392								
Plantar pressure per step at load site									
In-shoe PP (kPa)	247.66								
Barefoot PP (kPa)	266.00								
In-shoe PTI (kPa / s)	46.09								
Barefoot PTI (kPa / s)	37.68								
In-shoe contact time (s)	0.71								
Barefoot contact time (s)	0.61								
Estimated plantar pressure per week at load site									
PP (kPa)	8,535,439								
PTI (kPa / s)	1,548,132								
Contact time (s)	24,036								
Mean plantar pressure applied per device comp	pression at load site								
PP (kPa)	374.47								
PTI (kPa / s)	132.31								
Contact time (s)	0.58								
Mean plantar pressure applied by device per	week at load site								
PP (kPa)	1,348,092								
PTI (kPa / s)	476,308								
Contact time	2,072								
Total weekly plantar pressure experienced under load site (% applied with device)									
PP (kPa)	9,883,531								
$\mathbf{DTI}(\mathbf{l}_{\mathbf{D}}\mathbf{p}_{1}/\mathbf{p})$	(13.6)								
r11(Kra/s)	(23.5)								
Contact time (s)	26,108 (7.9)								

Table	6.11	Sub	iect 4	loading	profile
I GOIC		Duo.	JUCC I	iouuing	prome

6.7.3 Skin data

Figures 6.11 - 6.14 and Tables 6.12 - 6.15 show the raw data and SDC collected from the skin for hydration, distensibility, homogeneity and variance respectively.



Figure 6.11 – Subject 4 raw hydration data.

Τŧ	ıb	le	6.	12	2 9	Suk	Die	ect	4	sma	llest	de	tect	tabl	e (change	for	hv	dra	tion

	Load site	PMA	Base of 5 th metatarsal	Opposite foot
Mean difference between time points	3.81	6.05	0.28	8.39
SD of differences between time points	1.83	1.84	1.51	2.00
SEM	1.30	1.30	1.07	1.42
SDC	3.60	3.61	2.95	3.92



Figure 6.12 – Subject 4 raw distensibility data.

	Load site	PMA	Base of 5 th metatarsal	Opposite foot
Mean difference between time points	0.14	0.03	0.07	0.07
SD of differences between time points	0.12	0.06	0.09	0.08
SEM	0.09	0.05	0.06	0.06
SDC	0.24	0.13	0.17	0.16

Table 6.13 Subject 4 smallest detectable change for distensibility



Figure 6.13 – Subject 4 raw homogeneity data.

Table 6	.14 Sub	ject 4 sma	allest detec	table chang	e for	homogeneity
		U		C	,	

	Load site	PMA	Base of 5 th metatarsal	Opposite foot
Mean difference between time points	0.04	0.04	0.01	0.05
SD of differences between time points	0.05	0.06	0.04	0.08
SEM	0.04	0.04	0.03	0.05
SDC	0.10	0.11	0.07	0.15



Figure 6.14 – Subject 4 raw variance data.

Table	6.15	Subject 4	smallest	detectable	change	for	variance

	Load site	PMA	Base of 5 th metatarsal	Opposite foot
Mean difference between time points	0.40	0.39	0.28	0.56
SD of differences between time points	0.57	0.53	0.45	0.91
SEM	0.41	0.37	0.32	0.64
SDC	1.13	1.04	0.89	1.79

6.8 Case study – Subject 5

6.8.1 Participant profile

Subject 5 was a 20 year old, white British female. She was generally healthy and participated in regular sporting activities including netball and gym work. On examination, bilateral callus beneath the fifth metatarsal heads was observed with healthy skin on the rest of the forefoot. The right second metatarsal head was chosen as the loading site, and the first metatarsal head, fifth metatarsal base and second metatarsal head of the left foot were chosen for control measurements.

6.8.2 Loading profile

Table 6.16 shows the loading profile for the loading site on Subject 5's foot which includes activity data, dynamic pressure data and pressure data collected from the loading device. It also shows the estimated accumulative dynamic loads and how much of this was contributed by the loading device. The average load applied by the device was approximately 2.1 times higher than normal pressure.

Activity profile							
Number of strides per week	34,562						
Time stepping	44,640						
Time standing	159,840						
Plantar pressure per step at load	l site						
In-shoe PP (kPa)	182.19						
Barefoot PP (kPa)	291.00						
In-shoe PTI (kPa / s)	29.69						
Barefoot PTI (kPa / s)	62.10						
In-shoe contact time (s)	0.49						
Barefoot contact time (s)	0.58						
Estimated plantar pressure per week a	at load site						
PP (kPa)	6,672,919						
PTI (kPa / s)	1,138,116						
Contact time (s)	17,222						
Mean plantar pressure applied per device com	pression at load site						
PP (kPa)	384.87						
PTI (kPa / s)	128.41						
Contact time (s)	0.58						
Mean plantar pressure applied by device per	week at load site						
PP (kPa)	1,385,532						
PTI (kPa / s)	462,278						
Contact time	2,096						
Total weekly plantar pressure experienced under load site (% applied with device)							
PP (kPa)	8,058,451						
	(17.2)						
P11 (kPa / s)	1,600,394						
Contact time (s)	19,318						
	(10.9)						

Table 6.16 Subject 5 loading profile

6.8.3 Skin data

Figures 6.15 - 6.18 and Tables 6.17 - 6.20 show the raw data and SDC collected from the skin for hydration, distensibility, homogeneity and variance respectively.



Figure 6.15 – Subject 5 raw hydration data.

Ta	ab	le	6.	17	Su	biec	t 5	smallest	dete	ctable	change	for	hvdrat	tion
					~ ~									

	Load site	PMA	Base of 5 th metatarsal	Opposite foot
Mean difference between time points	6.11	9.11	5.85	0.50
SD of differences between time points	1.94	3.97	3.04	1.69
SEM	1.37	2.81	2.15	1.19
SDC	3.80	7.78	5.96	3.30



Figure 6.16 – Subject 5 raw distensibility data.

	Load site	PMA	Base of 5 th metatarsal	Opposite foot
Mean difference between time points	0.11	0.04	0.03	0.04
SD of differences between time points	0.13	0.09	0.08	0.09
SEM	0.09	0.07	0.05	0.06
SDC	0.25	0.18	0.15	0.17

Table 6.18 Subject 5 smallest detectable change for distensibility



Figure 6.17 – Subject 5 raw homogeneity data.

Table 6.19	Subject 5	smallest	detectable	change f	or homogeneity

	Load site	PMA	Base of 5 th metatarsal	Opposite foot
Mean difference between time points	0.11	0.09	0.07	0.05
SD of differences between time points	0.05	0.04	0.11	0.03
SEM	0.04	0.02	0.08	0.02
SDC	0.10	0.07	0.21	0.06



Figure 6.18 – Subject 5 raw variance data.

Tabla 6 30	Subject 5	amallaat	dataatabla	ahanga	for	vonionoo
1 able 0.20	Subject 5	smanest	detectable	change	10 Г	variance

	Load site	PMA	Base of 5 th metatarsal	Opposite foot
Mean difference between time points	1.03	0.77	0.65	0.50
SD of differences between time points	0.62	0.38	1.08	0.46
SEM	0.44	0.27	0.76	0.32
SDC	1.22	0.74	2.12	0.90

6.9 Discussion

Contrary to the widely held hypothesis, the data suggest that there was no skin response at the area of additional loading. There are a number of points related to methodology and implications which must be considered in the context of this unexpected nil-response.

Tables 6.1, 6.6, 6.11 and 6.16 show the estimated percentage of total weekly dynamic loading delivered during the loading sessions by the device to each participant's plantar skin. These show that while only three appointments per week were possible, the volume of load administered in relation to what the site of load might normally experience (albeit estimated) was relatively high. The percentage of peak pressure ranged from 13.6% to 28%; PTI, the sum of total loading 23.5% to 41.5%, and contact time 7% to 13.6%. It is important to highlight that the results of this study only reflect estimated loads due to walking and not due to standing. The percentage load contributed by the device to the total accumulative loads experienced by the foot site would be much lower if the periods of standing were taken into account. The activity data shows that the duration of standing is considerably higher than that of stepping; for Subjects 4 and 5 almost three times as much.

It is possible that much longer periods of loading per day and week provide optimum conditions for stimulation of callus, rather than cyclical loading during stepping activities. If so then the loads added to the feet of participants is relatively small compared to the loads experienced each week, perhaps too little additional load to stimulate a skin response, thus explaining the results. It might be that duration of loading as opposed to magnitude of peak pressure is the main factor in callus development. An increase in pressure resulting, for example from an anatomical deformity which might lead to callustype skin, is present in the individual's foot all the time, and in cases where footwear might be a factor, a period of several hours for at least several days per week. This amount of time would not be feasible in a laboratory controlled study and it would be unethical to request such a commitment from participants.

Conversely, if volume of accumulative load is the factor irrespective of duration of load, an increase in the level of pressure applied by the device at each loading session may have accelerated a skin response as it would have increased this volume of total load. However, increasing the pressure applied to the foot by the device was not appropriate. Only two fold normal in-shoe plantar pressure values could be tolerated by most of the participants for a short period of time. Higher pressures were tested in preliminary testing sessions and were found to cause discomfort, so for this, and also safety reasons, a maximum of three fold normal peak pressure was considered appropriate, but could be reduced if necessary to suit the participants. In these sessions, it was also deemed that 20 minutes was an appropriate timeframe for each session of loading. This was due to time commitment required and also comfort. In one case, Subject 2, it still proved to be a magnitude of load that was too great and could not be tolerated. Furthermore, only three days per week of loading was possible. All the individuals recruited were from the University and worked or studied in relatively close proximity to the author of this thesis. This was considered important to allow for frequent loading sessions with minimal disruption of work commitments, thus adding recruitment and retention of participants. Recruiting outside the institution would not have proved effective since the burden on participants would have been too great.

Other difficulties in the recruitment process involved the recruitment criteria. Originally, the criterion was for healthy females above the age of 40 years. Previous research has suggested that older females are more likely to develop callus (Menz et al., 2007) and while an inclusion age of 40 is not by any means elderly, having an age range too narrow would make it difficult to reach the intended sample size in a population of individuals working or studying at the University. However, due to the poor initial response, the inclusion criteria were changed to individuals of any age with the presence of callus somewhere on the foot. This was deemed to be acceptable because the presence of callus would suggest that the individual's skin would be prone to such a response with the correct conditions.

A potentially important factor in eliciting a skin response might have been the type of load administered to the plantar skin. This study focussed purely on application of vertical pressure, due to the fact that several studies (Potter and Potter, 2000b, Pataky et al., 2002, Duffin et al., 2003) have reported significant increases in vertical plantar pressures beneath callused skin compared to normal skin and control subjects. Further studies have investigated changes in normal pressure in response to treatment (Pitei et al., 1999, Potter and Potter, 2000b, Woodburn et al., 2000, Davys et al., 2005). However, shear force might also have been a beneficial topic to study in place of or in addition to vertical pressures, given that several studies have noted skin changes at a histological level resulting from shear loads (Rubin, 1949, Mackenzie, 1974a, Mackenzie, 1974b, Goldstein and Sanders, 1998). Given that these studies have shown skin responses in human and animal non-plantar sites in a relatively short timeframe (between seven and 35 days), it could be argued that a similar approach could have reaped a similar response in plantar skin. However, it must be stressed that these studies only investigated non-plantar skin sites, areas that are anatomically and functionally different from plantar skin, and are not designed for bearing loads, so this could explain why these studies found it relatively easy to drive skin change in a short timeframe. As plantar skin is designed for bearing loads, given that literature already supported the hypothesis of vertical pressure being related to plantar callus, the need to test the skin response to vertical pressure application took precedence over shear pressure. Vertical pressure is much easier to measure, and administer repeatably and safely. It is also expected to be in the region of six times greater than shear pressure, and therefore could be argued to be more worthy of investigation, at least initially. Shear pressure has been shown to be difficult to measure and studies have found conflicting results regarding the locations of peak shear pressures beneath the foot (Pollard et al., 1983, Tappin and Robertson, 1991, Hosein and Lord, 2000, Perry et al., 2002, Yavuz et al., 2007, Stucke et al., 2012, Mori et al., 2012).

While applying shear pressure using the loading device used in this chapter is technically possible, it is very challenging because the nature of the compression application required to allow the shear actuator to move whilst in contact with the skin is difficult to achieve. Therefore, the repeatability of compression to shear ratios within and between loading sessions could be poor with the current device. Also, if one were to apply similar levels of compression and shear as is present during gait (1.2 and 0.2 times bodyweight respectively (Richards, 2008)) a very large shear actuator and a more robust device design might be needed, perhaps with a more sophisticated pressure control system than manually adjustable pressure regulators. Applying such large shear forces artificially with a metal device could be potentially dangerous with the risk of skin tearing a possibility. For these reasons, the most feasible starting point for studying the relationship between loading and callus was vertical pressure. However, addressing the role of shear could be a potentially beneficial study to undertake at a later time, and is certainly an important area of study.

One challenge for this study and indeed any study of factors affecting skin properties over time is the apparently normal large variation of skin properties over time. As shown in the biophysical measurements data, the skin experiences apparently normal changes in its properties over time under typical circumstances; indeed this study is perhaps the first to reveal this in pedal skin. The reasons for this could include changes in footwear (and thus occlusion), weather and activity over time. The SDC values for skin hydration suggest that relatively large changes in skin properties at single skin sites are necessary in order for the change to be considered 'real' and not due to random error (i.e. statistically significant at 95% confidence level). However, the relatively large SDC values are likely to be a product of skin variation over time. The values might suggest that skin changes may be somewhat hidden by measurement error but control sites would also be susceptible to these changes, so comparing the load site to controls would still allow an effect to be identified.

Certainly, Studies on callus and heel fissure treatments that the author of this thesis co-authored at this institution (Hashmi et al., 2015; under review) clearly showed an obvious skin change in the regions of interest in comparison to the control sites which could not be attributed to normal variation over time. These studies showed that increases in hydration of up to 152%, and distensibility of up to 100% in callused skin sites, between baseline and post-treatment, were possible after podiatry treatment for callus. This magnitude of skin change is far larger than can be attributed to normal skin variation, and these changes were obvious in relation to the control site. In other studies of pedal skin properties, large changes in hydration within a smaller timeframe than the work in this chapter have been demonstrated post-treatment with topical moisturisers. Garrigue et al.

(2011) observed a 48.9% and 57.3% increase in hydration after 14 and 28 days (p = 0.0002and < 0.0001 respectively), and Papanas et al. (2011) observed an 8.9% and a 20.6% increase after seven and 14 days respectively (p < 0.001). In the context of the study in this chapter, if normal skin were to become 'callus-like', it should clearly show in the data. The fact that it has not suggests that, in this case, the loading protocol was not enough to elicit a skin change. It might be that this study was not conducted over a long enough period for any possible changes to manifest in relation to the control sites and that skin changes, if very small, were masked beneath normal week to week variation. However, as previous studies have shown obvious skin changes within 35 days (Rubin, 1949, Mackenzie, 1974a, Mackenzie, 1974b, Goldstein and Sanders, 1998), the period of loading used in this study was deemed a suitable timeframe for identifying skin changes. Because no skin change due to loading was identified, a future study investigating vertical pressure and skin response might have to be conducted over a period of many more months before a noticeable change can be detected. It might be that a large timeframe is the only way to cause a biophysical change in plantar skin, a site that is anatomically designed for load bearing. Such studies will prove very demanding for participants and researchers alike.

One area that must be addressed before concluding future study is safety. Preliminary tests were conducted, as discussed previously, to determine the loading dose and frequency of loading sessions for the study. However, these aspects were not completely rigid for the entire study and could have been subject to change if need be. At each session, the load was set for each participant and during the loading cycle, feedback was sought regarding comfort. This was important because it ensured participant safety. The second participant in the study was required to withdraw because the sensation of having the probe head applied to the skin was uncomfortable and caused a small amount of bruising to the skin. This was identified by participant feedback and highlights its importance. No further data was collected from this participant, but on examination it was noted that she had particularly thin tissue overlying the metatarsal heads compared to the other participants. The participant also reported not undertaking regular exercise, leading a relatively sedentary lifestyle compared to Subjects 1, 3, 4 and 5 who all undertook regular, strenuous exercise including running, sports and gym sessions. This might have meant that the tissues in her foot were not accustomed to higher loads and were therefore more sensitive and prone to injury. This could be a potential factor to consider in recruitment – seeking individuals who are physically active so they can tolerate higher loads. However, it seems logical that individuals that are sedentary might show a greater skin response to

laboratory-induced loads due to the fact that their skin is not as used to intense loading. The ability to apply person specific loads using the device used here addresses this concern to some degree, because even if high pressures are common for a specific foot, at least two fold these high pressures can be applied. Careful application of loads, beginning from a small load, such as applying the same level of pressure they would normally experience beneath the area of interest, and building up to a higher, but tolerable load, after several weeks in the study might be of benefit to pre-condition the skin. However, this would also further increase the burden on research participants.

Regular examination of the feet is also recommended to identify any inadvertent bruising or inflammation caused by the device, or any abrasions or blistering that could occur. Subject 5, during the second week of loading, developed a small abrasion at an adjacent area to the loading site, caused by the probe rubbing on the skin. The adjacent skin in question had a small patch of callus and was therefore stiffer than the healthy skin of the loading site. This stiffness of the adjacent skin was likely responsible for the small abrasion so the participant was sent for a podiatry assessment and had the hard skin debrided. The next loading session was delayed to allow time for the skin to heal, and then commenced as normal. Regular examination of the foot was undertaken for each participant subsequently to ensure safety.

In addition to those highlighted above, this study has several other limitations which need to be taken into account. Firstly, regarding activity monitoring, it is important to note that it only represents data collected in a given week and not necessarily every week. Levels of activity inevitably will vary from week to week so the estimated percentage of loading beneath the foot delivered via the loading device will be different each week. For optimum accuracy, one would have to wear the monitor for the entire duration of the study.

Another limitation is that pressures from barefoot and shod walking were used to create a target plantar pressure that would be applied using the device. Activity monitoring data assists in this profile but does not give an indication of how much time is spent walking shod and barefoot in real life. In this study, the author had to make an assumption of shod walking accounting for 90% and barefoot 10% of physical activity. It is likely the actual ratio between the two is different for different people and not consistent on a day-to-day basis.

When processing pressure data collect from the Pedar[®] and Emed[®] devices, a mask was used to overlay the loading site in order to capture the data. Due to the difference in

pressure sensors and software, the masks may have varied slightly in size. For Pedar[®] the sensor sizes are relatively large so potential error exists in that adjacent metatarsal data may have been collected within the mask. The metatarsal heads cannot be visualised using Pedar[®] and the sensors are too large to be able to have a mask overlay the exact area. The mask used was the only one underpinned by research relating to foot anatomy (Bontrager et al., 1997) – the mask in question overlaying the second, third and fourth metatarsal heads with 16 pressure sensors in an array of 4 x 4 (Figure 6.1). In this study, either the second or the fourth metatarsal head was used as the loading site so the mask was divided into two with the area overlying the region of interest used. There is a small possibility of some third metatarsal head data being collected within this mask, but if the mask was further reduced making it narrower, it may have overlaid the wrong area completely. For the Emed[®] MatLab software, it is possible to approximate the region of each metatarsal head visually from the pressure footmap which aids in creating the mask. Due to these factors, there was possibly a discrepancy between data collected between the two systems. If peak pressure data was actually from an adjacent metatarsal head as opposed to the area of interest, this could have led to over or underestimation of pressure magnitude. For PTI, the mean pressures of all the sensors within the mask were used. This was deemed a safer option in the case that other, unwanted metatarsal heads strayed into the mask. However, it may have had the effect of underestimating the PTI that was actually occurring at the load site. Using a PTI with the summed pressure values of all the sensors in the mask would have been most accurate provided the mask exactly overlaid the area which the contact pad of the device contacted and had the same surface area. However, this would have been exceptionally difficult due to the size and location of the Pedar[®] sensors. With advanced software, one might be able to achieve something similar in Emed[®] but as of yet, none is available.

6.10 Conclusion and implications for further research

Overall, while no real skin response was observed for any participant, this study proved to be a useful exercise towards improving our understanding of how external loads might affect plantar skin properties. The device and pilot work represent the design and implementation of a novel research study that has never been attempted on plantar skin previously.

Several possible reasons have been identified for the nil-response shown in this chapter, and recommendations for others attempting this type of research are highlighted.

The load dose used in this study may not have been high enough or administered over long enough duration of time to elicit a callus-like skin response. However, increasing this loading dose any more would make recruitment exceptionally difficult and raise possible safety concerns. Recruitment was undertaken to allow the greatest chance of a skin response (e.g. feet known to be able to produce callus), but a narrower recruitment criteria such as older females might have been beneficial (due to changes in skin properties with age). However, this too may make implementation of the study more difficult. The direction of pressure administered to the skin may not have provided optimum conditions for skin response. Research supports increased vertical pressure beneath callus, which was the rationale for further investigating this phenomenon, but shear pressure application might have been proven effective as has been highlighted in dermatology studies. In this case shear was too difficult to administer and vertical pressures were deemed a more suitable initial focus. Safety considerations including small initial doses of load, particularly for individuals leading a sedentary lifestyle whose feet are not used to being vigorously loaded and regular examination of the feet have also been recommended for skin loading research.

This project, to the author's knowledge, is the first of its kind investigating the effect of regular load doses on plantar skin and using biophysical skin measurement devices to record skin response. It is also the first to relate activity monitoring data to plantar pressure data to estimate accumulative loads. This is an approach that could be beneficial in other areas of research, such as assessment of loading profiles of individuals at risk of ulceration in diabetes and rheumatoid arthritis. The type of pressure research conducted in a laboratory, while useful, provides only a very narrow snapshot of individuals' loading profiles, and activity monitoring could allow a greater understanding of this and possibly identifying those at risk of ulceration or explain the presence of existing lesions.

To summarise, this study investigated the relationship between plantar loading and callus, implemented a novel loading device for the first time, and sought to track changes in plantar skin through a pilot study. The non-response observed in the skin is interesting because it points to the need for perhaps greater vertical plantar loads or need for shear loads to trigger skin responses. The next step in understanding the relationship between loading and callus is to investigate how existing calluses respond to pressure reduction; do the lesions regress and become more like 'normal' plantar skin? This will be investigated in the final study, Chapter 7.

Chapter 7: The effect of pressure reduction on the biophysical properties of plantar callus

7.1 Introduction

Orthotic insoles have been used as a conservative pressure reduction modality in podiatry and other physical therapy professions for many years. One of their key uses is to offload pressure areas on the plantar foot, which is particularly beneficial for individuals with diabetes and rheumatoid arthritis, as high plantar pressures have been associated with ulceration (Boulton et al., 1983, Murray et al., 1996). As a result, much of the research into orthotic insoles has focussed on these groups with minimal studies focussing on their use in treating pressure related lesions. However, in order to understand how insoles work it is necessary to explore the literature on their design and common uses. This chapter will begin with a review of the literature into insole design to set the context for the research before detailing the study.

7.2 Review of the literature

7.2.1. Efficacy of customised insoles

In the literature, the benefits of the use of custom moulded insoles to reduce plantar pressures under the forefoot have been widely reported. Raspovic et al. (2000) assessed the effect of using customised orthoses as a pressure relieving measure in individuals (n = 8) with healed diabetic neuropathic ulcers. F-Scan in-shoe pressure data revealed that customised insoles reduced peak pressure (p < 0.01) and pressure time integral (PTI) (p <0.05), and increased the contact area (p < 0.01) compared to the shoe-only condition. The actual pressure reductions ranged from 6% to 93%. They authors highlighted that the extent of pressure the reduction was highly variable between individuals.

Several studies have reported significant reductions in peak pressure and PTI with increases in surface area in custom moulded insoles compared to flat insoles. In 42 participants with metatarsalgia, Postema et al. (1998) reported a significant reduction in peak pressure at the distal central and lateral forefoot of 18.2% and 10.8% (p = 0.000 and 0.03 respectively) and significant reductions of force impulse at the distal central and lateral, and proximal central forefoot (p = 0.006, 0.005 and 0.000 respectively) with a custom insole compared to a standard insole. Birke et al. (1999) found a 55% reduction in pressure (p < 0.05) beneath high pressure areas in 19 participants with diabetes with a
history of ulceration in their own shoes with customised insoles compared with a standardised shoe alone and with flat Poron insoles of different densities. In 21 subjects with diabetes presenting with neuropathy plus foot deformities, Bus et al. (2004b) found a 15.6% decrease in peak pressure over the first metatarsal head (p < 0.05) but no significant reductions beneath the other metatarsals. However, they did report a significant 10.1% decrease (p < 0.05) in force/ time integral beneath the lateral metatarsal heads. The peak pressure and force/ time integral were transferred to the medial midfoot. Tsung et al. (2004) found, in a group consisting of 8 healthy individuals and 6 individuals with diabetes, that custom moulded insoles manufactured using scans taken at different loading conditions significantly reduced the peak pressures and increased the surface areas over the forefoot compared with shoe only conditions. The pressure reductions reported ranged from 2.6% to 17.9%. Mueller et al. (2006) studied the effects of pressure relieving modalities in a sample of individuals suffering from diabetes with a history of neuropathy and ulceration (n = 20). They reported a significant reduction in peak pressure (p < 0.03) and PTI (p < 0.036) with total contact insoles compared to shoes in each metatarsal head. Pressure reductions of between 19% and 24% were reported. In a randomised controlled trial (Burns et al., 2009), it was reported that customised insoles significantly decreased peak pressures in the forefoot (p = 0.034) and rearfoot (p < 0.001) compared to sham orthoses in individuals with diabetes with peripheral arterial disease and foot pain. The pressure reductions are the forefoot were 16.3% in the customised insole group and 10.6% in the sham group when compared to shoe only conditions (Burns et al., 2009).

However, not all studies have found significant improvements in using customised orthoses over prefabricated. Novak et al. (2009) conducted a randomised controlled study comparing the pressure reduction effects of customised orthoses versus flat insoles in a sample of rheumatoid arthritis patients (n = 40). They reported no significant differences in pressure reducing effects of prefabricated and customised insoles in randomised cross-over trial with individuals presenting with flat feet. They found that while both sets of insoles reduced peak pressure, mean pressure, PTI and force/ time integral at the forefoot compared to using a shoe only (p < 0.05), there was no significant difference noted between the insoles. In a group of 18 runners tested in normal and custom fit additive manufactured insoles, Salles and Gyi (2013) reported no significant differences in forefoot pressures between the insoles, noting only differences at the heel region.

The efficacy of full contact insoles versus flat insoles has also been explored using computational analysis. Chen et al. (2003) found, comparing finite element models of two different full contact insole types against a flat insole, that peak and normal stresses were reduced in all areas of the plantar foot except for the medial midfoot where the stresses increased as a result of stress redistribution.

These studies show that generally full contact customised insoles are more effective than no insole or flat insoles in reducing peak pressures. This is due to the fact that pressures are redistributed to the midfoot region, therefore offloading regions with higher pressures, such as the metatarsal heads.

7.2.2. Methods of insole design and manufacture

There are several papers exploring different methods of insole design. Tsung et al. (2004) explored the pressure relieving effects of insoles moulded using different loading conditions including non-weight-bearing, semi-weight-bearing and full-weight-bearing. They found that all conditions were significantly more effective than using flat insoles with reductions ranging from 2.6% to 17.9% across the metatarsal heads. When comparing the three different loading conditions, it was found that the insole defined using the semi-weight-bearing foot shape was the most effective at reducing peak pressures at the medial forefoot with reductions of 17.9% at the first to third metatarsal heads. The insole defined using the lateral metatarsal heads (14.8%). This insole also reduced PTIs at the forefoot up to 20.9%, and increased it at the midfoot by 23.7%, which the authors stated was due to the greater arch support compared to the other insoles.

Another study has compared the pressure relieving effects of custom moulded versus insoles fabricated from material inserted in participants' shoes and moulded through walking (termed dynamic impression insoles) (Chang et al., 2012). Beneath the forefoot, the custom moulded and dynamic impression insoles reduced peak pressures by 34.3% and 46.3% respectively compared with the flat insole (p < 0.05). The dynamic impression insoles significantly reduced peak pressure by 18.2% compared to the full contact insole (p < 0.05). The authors also observed significant reductions in PTI (p < 0.05) and increased contact areas (p < 0.05) for both insoles compared to the control, and for dynamic impression insoles compared to custom moulded insoles at the forefoot.

One study has compared CAD-CAM designed insoles with insoles made using foot shape, captured using a foam impression box (Ki et al., 2008) in 30 individuals. It was

reported that the peak pressures and force were significantly reduced in the heel region (p = 0.000 and 0.000 respectively) and increased in the midfoot region (p = 0.004 and 0.000 respectively) for both customised insoles compared to a flat insole. The PTI was also increased in the midfoot for both insoles. When comparing the CAD-CAM with the foam moulded insoles, the CAD-CAM insole delivered a 15.3% lower peak pressure beneath the mid forefoot region than the foam moulded insole (p = 0.01).

Further research has looked into CAD-CAM customised insoles. Owings et al. (2008) explored the effect of using pressure data combined with foot shape data in CAD-CAM insoles in 20 participants suffering from diabetes. They sent foam impressions from each participant to three insole companies for CAD-CAM of insoles and one company also received barefoot plantar pressure data to accompany the foam box impressions. From plantar pressure analysis of the forefoot using the three pairs of insoles, the insoles which were manufactured using pressure data plus foot shape derived from the foam box showed significantly lower peak pressures at 14.2% and 30.7% reduction compared to the two insoles that used foam box only (p < 0.0001). The force/ time integral (p < 0.0001) was also significantly lower at the forefoot, but increased at the midfoot (p < 0.01) for the insoles developed from foam box and plantar pressure data.

This section has highlighted the potential benefit of using semi-weight-bearing loading conditions and/or created from dynamic material impressions to relieve peak pressure. The use of CAD-CAM as an insole fabrication method may give quality insoles for pressure reduction and this could be further enhanced by the use of pressure data (Owings et al., 2008).

7.2.3. Insole materials

There is an abundance of studies addressing the efficacy of different materials in plantar pressure reduction and comfort. An early study by Leber and Evanski (1986) compared the pressure reduction qualities of seven different insole materials on patients with high plantar pressures (n = 26). The materials included latex foam, Plastazote, Dynafoam, Orthofelt, PPT, Spenco and Molo. Using the Harris and Beath footprinting technique, they observed that all conditions significantly reduced pressure with reductions ranging from 28% - 53% (p < 0.01) compared with the no-material control footprint. PPT, Plastazote and Spenco were significantly more effective at reducing pressure than the other materials, reducing the pressure between 51% and 53% compared to the control (p < 0.01). Barrow et al. (1992) further investigated the pressure relieving properties of PPT and Spenco.

Participants (n = 7) with high plantar pressure beneath the second metatarsal head wore a total of four insoles for one month each. The insoles included two PPT and two Spenco with a U-shaped cut-out beneath the second metatarsal head and a sponge-filled U-shaped cut-out beneath the second metatarsal head. Pressures were measured using pedobarography at monthly intervals to allow data to be collected with new insoles, then insoles after one month of wear. The authors found that all four insoles significantly reduced pressure (percentage reductions not published) (p < 0.0001) and there was no significant difference between the pressure relief of PPT and Spenco. There was no significant difference between new and old insoles in pressure relief. The Spenco insoles with the sponge filled cavity slightly offloaded pressure beneath the second metatarsal head, but this was not evident for the PPT insoles.

Birke et al. (1999) compared the pressure relieving effects of flat Poron insoles at different densities in 19 individuals with diabetes with a history of ulceration. They found that medium density Poron (shore 22, 27 and 32) was significantly more effective at relieving mean pressures (range 36% - 39.5% reduction) than soft density (shore 14 and 17) and high density (shore 40 and 55) Poron (range 20% - 33.7% reduction, p < 0.05). However all Poron densities were significantly more effective than a shoe only condition but not as effective as a moulded insole in the participants' own shoes which gave 55% reduction in mean pressure (p < 0.05).

In a population of German soldiers (n = 26), Hinz et al. (2008) compared conventional army boot insoles with prefabricated, contoured EVA and Neoprene insoles to assess their pressure relieving effects for use in injury prevention. They found that forces were significantly reduced in the Neoprene insoles compared with the conventional insoles (p < 0.0006) across each metatarsal head and the EVA insoles (p < 0.00001) across metatarsal heads three to five. Compared with the other insoles, peak pressures were significantly lower in the Neoprene insoles in the second metatarsal head (p < 0.003) and the third to fifth metatarsal heads (p < 0.0001) as was the force/ time integral across all metatarsal heads (p < 0.0002). Peak pressure reductions were shown to range from 20.6% to 31.4% for the Neoprene insoles compared to the conventional types beneath the second and third metatarsal heads.

Tong and Ng (2010) investigated the pressure reduction effects of slow recovery Poron, standard Poron, Poron with firm Plastazote and Poron with soft Plastazote. Simple 6.5mm thick insoles were made from the materials and pressures were measured using an F-Scan system in 5 subjects. They found that all materials could reduce pressure compared with no insole, but only the Poron with high density Plastazote reached significance reducing the mean peak pressure by 27% (p < 0.03).

Healy et al. (2012) studied EVA and PU materials in custom and flat insoles in 10 subjects. All materials significantly reduced peak pressures at the first and lateral metatarsal heads for both insole types with reductions ranging from 2.1% to 18.3% for the flat insoles and 6.5% to 18.6% for the custom insoles. Peak pressures were also significantly reduced at the heel for the custom moulded insole type. Comparing the materials, the authors found that PU was effective at reducing PTI and increasing contact area, while medium density EVA showed higher pressures in flat insoles than the other materials.

Two studies have assessed the effects of wear on pressure reduction capabilities of insoles. Rogers et al. (2006) investigated how two insole compositions, Poron and a mixture of Poron and Plastazote, reduced pressure before and after 50,000 steps assessed with a pedometer in 19 subjects. They observed significant reductions in peak pressures before (p < 0.05) and after (p < 0.05) 50,000 steps for both insole compositions compared to a shoe only condition. The pressure reductions were 29.9% and 28% before 50,000 steps; and 18.3% and 25.3% reduction after 50,000 steps for Poron and Poron/ Plastazote respectively. The Poron insoles' pressure reduction qualities were significantly reduced after 50,000 steps (16.6% increase in pressure compared to measurements taken before 50,000 steps; p < 0.05), but there was no significant impairment in pressure reduction for the Poron and Plastazote insole. Cronkwright et al. (2011) found similar results with a prefabricated, dual density Formthotic insole before and after one year. Compared to shoe only, the new insole reduced pressures beneath the forefoot and heel by 12% and 23% respectively (p < 0.05 and 0.01). The old insole still reduced pressures in these regions compared to the shoe, but only the heel reduction was significant (p < 0.01). There was no significant change in the forefoot between old and new insoles but the heel pressures were significantly higher in the old insole (p < 0.01). The maximum force was significantly increased in the forefoot and decreased in the heel in the old and new insole compared to the shoe only (p < 0.01), and significantly increased in the forefoot in the old insole (p < 0.01)(0.05). Compared to the new insole, the force was significantly higher in the forefoot (p < (0.05) and the heel (p < 0.01) in the old insole. Contact areas were also significantly increased in the new and old insoles compared to the shoe (p < 0.01) but not between insoles.

These studies show that most insole materials are capable of reducing peak pressures compared to wearing a shoe without an insole. The most effective materials are arguably PPT, Spenco, Poron combined with Plastazote, Neoprene and PU and these could be particularly effective in a customised insole. Insole wear also appears not to significantly affect pressure reduction in most cases, even after one year of wear.

7.2.4. Pressure relieving additions to insoles

There is a wealth of literature focussing on pressure relieving additions to insole design. These include metatarsal domes, pads and bars which are designed to offload all or individual metatarsal heads and relieve pressure and pain. Specific pressure reductions are shown in Table 7.1.

Study	Modality	Pressure without modality	Pressure with modality	Pressure reduction (%)
	Metatarsal pad - met head 1	2.43 kg/cm ²	2.16 kg/cm^2	-11.1
Holmes and	Metatarsal pad – met head 2	4.31 kg/cm ²	3.12 kg/cm^2	-27.6
Timmerman (1990)	Metatarsal pad – met heads 3 & 4	2.91 kg/cm ²	2.25 kg/cm ²	-22.7
	Metatarsal pad – met head 5	1.92 kg/cm ²	1.6 kg/cm^2	-16.7
Poon and Love (1997)	Metatarsal dome	2833 g/cm	2366 g/cm	-16.5
Hodge at al. (1000)	Orthosis with dome	25.87 N/cm ²	21.82 N/cm ²	-15.7
Houge et al. (1999)	Orthosis with bar	25.87 N/cm ²	20.57 N/cm ²	-20.5
Jackson et al. (2004)	Orthosis with dome	274.5 kPa	242 kPa	-11.8
	Orthosis with bar	274.5 kPa	216.1 kPa	-21.3
Kang et al. (2006)	Orthosis with dome	225.8 kPa	199 kPa	-11.9
Lott et al. (2007)	Orthosis with dome	176 kPa	143 kPa	-18.8
Lin et al. (2012)	Orthosis with cut-out at region of interest	262.5 kPa	149.9 kPa	-42.9
Liff et al. (2015)	Orthosis with cut-out plus arch support	262.5 kPa	135.6 kPa	-48.3
	Orthosis, dome 10mm proximal	399 kPa	364.7 kPa	-8.6
Lee et al. (2014)	Orthosis, dome 5mm distal	399 kPa	331.9 kPa	-16.8
	Orthosis with bar	399 kPa	358.2 kPa	-10.2

Table 7.1 – Pressure changes beneath forefoot using pressure relieving additions reported in the literature

Pressure changes calculated by new value – old value / old value x 100.

The use of metatarsal pads/ domes have been found to offer significant reductions in peak pressure and pain scores in individuals with metatarsalgia (Poon and Love, 1997);

diabetes and a history of neuropathy and ulceration (Lott et al., 2007, Guldemond et al., 2007, Mueller et al., 2006); rheumatoid arthritis (Jackson et al., 2004, Kang et al., 2006) and asymptomatic individuals (Holmes and Timmerman, 1990). Pads may also decrease soft tissue strain in healthy feet (Ibrahim et al., 2013) and increase tissue thickness in diabetic feet (Mueller et al., 2006). Extra arch support may decrease the peak pressures further in individuals suffering from diabetic neuropathy (Guldemond et al., 2007). Metatarsal bars also offer significant pressure reductions comparable to metatarsal pads/ domes (Hodge et al., 1999, Jackson et al., 2004) and may offer greater impulse reductions beneath the second metatarsal head (Deshaies et al., 2011).

The longitudinal placement of the metatarsal pad/ dome also has an influence on the extent of pressure reduction. One study (Hsi et al., 2005) reported that metatarsal pads placed 4.4mm proximal to the target area of pressure reduction was the most effective position, causing significant pressure reduction (p < 0.05) directly below the area of peak pressure at the second metatarsal head. Positioning the pad 8.8m proximal, or directly at the site of peak pressure would lead to significant reductions 4.4mm distal to those points. Hastings et al. (2007) observed that pads placed 6.1 - 10.6 mm proximal to the metatarsal heads may reduce pressure; pads placed between 1.8mm distal and 6.1mm proximal, and between 10.6 - 16.8mm proximal can have a variable pressure reducing effect; and pads placed more than 1.8mm distal or more than 16.8mm proximal to the metatarsal heads can lead to increased pressure (Hastings et al., 2007). Another study, however, found that placing the pad 5mm distal was a more effective method of reducing pressure beneath the metatarsal heads than 10mm proximal to the metatarsal line (Lee et al., 2014). The differences between results could be attributed to materials used: Hastings et al. (2007) used a metatarsal pad made of cork while Hsi et al. (2005) used foam rubber and Lee et al. (2014) used PPT foam. The measurement devices were also different.

One other study has also investigated the effect of longitudinal position of the metatarsal pad (between 0 - 25mm proximal to metatarsal line) combined with the use of two different thicknesses of pad (5mm and 10mm) (Brodtkorb et al., 2008). In contrast to the two studies highlighted previously, these authors found no significant impact of longitudinal axis position on mean plantar force values. While both pad thicknesses significantly decreased force beneath the second metatarsal head (p < 0.05), the 10mm pad was significantly more effective than the 5mm pad (p < 0.05).

While metatarsal pads and bars are very well publicised, another pressure reduction modality which exists is the use of cavities in the insole. Actis et al. (2008) designed a

cavity insole using finite element analysis based on the plantar geometry of two individuals with high peak plantar pressures. The insole in question had a cavity which could be filled with pressure reducing material positioned below the high pressure areas. It was found that the insole with a cavity filled with a matrix of soft Poron plugs that were 4mm in diameter was more effective at reducing high peak pressures than using a total contact insole, an insole with a forefoot inlay, and an insole with the forefoot area inserted with a matrix of 4mm plugs. A more recent study (Lin et al., 2013) assessed the pressure relieving effects of insoles with removable plugs at a region of interest. The subjects in question were 26 individuals presenting with diabetes and neuropathy. The forefoot area with the highest peak pressures as defined from in-shoe pressure tests were used as the regions of interest. The insoles were tested before and after plug removal, then with additional arch support. Peak pressures were found to be significantly reduced after insole plug removal (p < 0.001) and were further reduced when arch support was added (p < 0.001) 0.001). In the adjacent areas which were not regions of interest, there was no significant change between pre and post-plug removal. For both regions of interest and non-regions of interest, there was a significant reduction in peak pressures from baseline (6mm flat EVA insole) and the insoles with removable plugs (p < 0.001).

These studies show that the use of insole additions including metatarsal bars and pads to offload the metatarsal heads are effective in terms of pressure and pain reduction. This may be further enhanced with additional arch support (Guldemond et al., 2007). However, if using a pad, care should be taken to ensure optimal positioning in the longitudinal axis to achieve the best results (Hsi et al., 2005, Hastings et al., 2007, Lee et al., 2014). With these modalities, one should bear in mind that they are designed to relieve the pressure from several metatarsal heads and not specific points of high pressure such as an area of callus. For localised pressure reduction, insoles with cavities are a viable option (Actis et al., 2008, Lin et al., 2013).

7.2.5. Insoles used to treat plantar callus

Two studies have investigated the use of insoles for the treatment of callus (see Chapter 2). Briefly, Colagiuri et al. (1995) investigated the effect of a yearlong treatment using custom moulded plastic insoles worn seven hours per day in individuals with diabetes presenting with callus (n = 9). Clinical assessment of photographs taken before and after the study was used to assess callus grade and thus changes in callus. Results showed that there was a significant improvement (p = 0.02) in callus appearance compared with patients treated by

scalpel debridement (n = 11). However, plantar pressures were not assessed so whether or not these reductions in clinical appearance of callus can be attributed to pressure reduction is unclear.

Duffin et al. (2003) assessed pressure reducing measures a group of adolescents with type 1 diabetes with (n = 17) and without callus (n = 17). The pressure reduction measures included a custom moulded semi-rigid PE insole, PPT cushioning and both insole and cushioning combined. Cushioning reduced pressure by 21.8% and 20.2% (p = 0.001), the orthosis alone reduced pressures by 17.7% and 14.3% (p = 0.05) and the cushioning and orthoses combined reduced pressures by 27.7% and 31% (p < 0.001) in the groups with callus and high plantar pressure respectively. All pressure reduction measures also significantly reduced PTI (p < 0.05). Combined cushioning with orthoses was significantly more effective at reducing pressure than orthoses (p < 0.05) and cushioning (p < 0.001) alone. Out of their sample, they recruited 23 subjects who used custom made orthotic insoles and 67 control subjects who did not use insoles and reassessed them after a year. A significant reduction in pressure was found (p = 0.0003) with pressure changes ranging between +41% to -35% (mean reduction of 12 N/cm² / 120kPa) in the orthoses group; but no significant difference was found in the control group. In the insole intervention group, six participants had callus at the beginning of the trial but at the end of the trial two calluses had resolved. In the control group, seven had callus at the beginning of the trial and no calluses resolved. There were no quantitative measures of skin properties taken or any attempt to quantify the effect of pressure reduction on callus.

For the purposes of this thesis, these studies are very important as they highlight a possible link between pressure reduction and callus regression, but their methodologies are severely limited in that they do not use quantitative measures of skin properties. Also, it is unknown whether individuals were permitted to use foot treatments or whether there were any other factors which may have affected the skin. Furthermore, Duffin et al. (2003) provided no information as to whether the insoles were still delivering pressure reductions at the 12 month follow-up. Combined with skin biophysical measurements, a study which quantitatively measures pressure reduction using insoles would be beneficial to further understand the effect of offloading on the skin, but also for the efficacy of insoles as a pressure reduction measure in callused feet.

7.2.6. Conclusion

This review of the literature has highlighted that most commonly used insole materials are able to significantly reduce pressures, custom moulded insoles are particularly effective in offloading the metatarsal heads, particularly when used with a pressure relieving aid that is adequately positioned. The use of semi-weight-bearing or dynamic impressions and CAD-CAM fabrication are likely to result in the most effective pressure reduction. Offloading beneath callused skin may result in an improvement in skin characteristics. The subsequent study will provide participants with plantar pressure relief beneath forefoot calluses using the aforementioned literature as a guide, and investigate whether the callused skin changes over time.

7.3 Study aims and objectives

The aim of this study was to determine the effect on biophysical plantar skin properties of reducing pressure under the forefoot in individuals who present with plantar callus beneath the metatarsal heads. The objectives were to provide healthy participants with a plantar forefoot callus with a pressure relieving insole and assess the biophysical properties of the callused skin sites at baseline, and six and 12 weeks after using the insole.

7.4 Methodology

Ethical approval for this study was granted by the University of Salford's College Research Ethics Panel (application number HSCR14/41).

7.4.1. Subjects

The sample size for this study was informed using a paired T-test power calculation (http://biomath.info/power/prt.htm) using the skin characterisation data (presented in Chapter 4) from central callus and callus edge regions. The callus edge data was used because this skin site displays properties similar to central callus while still being significantly different to both this and normal skin, so in effect a halfway point between plantar callus and normal PMA skin. This would be an appropriate level of change to be confident of a skin response from pressure reduction. The results from the calculation suggested 16 datasets would be needed to determine whether or not significance would be achieved from this method of treatment.

Participants were recruited via flyers displayed in the waiting area of the Podiatry Clinic site and areas around the School of Health Sciences buildings at the University of Salford. Recruitment also involved radio advertisement, the Research Database at the University, and through an advert in a local newspaper. Participants presenting with plantar forefoot callus were included provided they did not have any of the following exclusion criteria which could potentially affect the skin's properties: compromised cardiovascular or neurological status, connective tissue disorders (such as lupus erythematosus), diabetes, autoimmune disorders (such as rheumatoid arthritis), peripheral vascular disease, or wounds/ ulcers of the legs and feet, and eczema, psoriasis or other dry skin disorders affecting the plantar skin.

Before being enrolled into the study, the foot was assessed for neuropathy and peripheral vascular disease. Pulses from the posterior tibial and dorsalis pedis arteries were assessed by palpation. Presence of neuropathy was assessed by vibration perception using a 128MHz tuning fork and 10g monofilament test on the hallux, first and fifth metatarsal heads. If these tests, skin assessment and medical history were satisfied, the volunteers were recruited to participate in the study.

7.4.2. Insole design and manufacture

Following recruitment and initial screening the participants attended a clinical screening which included an assessment of the foot. To allow a custom orthotic to be made, a 3D scan of the participants' feet were taken. The orthotic interventions were then designed and manufactured using CAD-CAM technology by a biomechanics researcher specialising in insole design and manufacture. This study formed part of a larger project comparing insole design, manufacturing methods and materials, and two designs and two materials were used in this study to enable reductions in forefoot plantar pressure. The insole materials used were medium density EVA and rubber, and the designs included a total contact insole or a total contact insole plus a cut-out area beneath the callused skin, both based on the 3D plantar surface geometry. The insoles were designed using the Custom 3Din plugin for Rhinoceros 5, with the basic design based on default templates within the programme. The basic design was using 3mm material with the top surface contoured based on the geometry of the individual's foot. The cut-out design was similar to this but with an extra 2 to 5mm of material added to the existing arch, up to a maximum height of 25mm, and an oval shaped cut-out positioned at the area of peak pressure corresponding to the site of callus. The latter was obtained from a pressure map image from the Emed[®] scan which was overlaid onto the CAD model of the insole. The cut-out dimensions were 20% insole length and 35% toe width. The insole designs were exported to INESCOP CAM software,

which generated milling paths and enabled the manufacture of the insoles in the CNC Router milling machine. The EVA insoles were made of 40Shore A hardness material. The rubber insoles were made off-site using additive manufacturing (AM) techniques, and were made of thermal rubber, also with a 40Shore A hardness. As the subject of this study is the effect of pressure reduction on callused skin, no further analysis of insole design and manufacture are presented; the only criteria for inclusion in the final data analysis was a reduction in peak pressure beneath the region of interest, i.e. the callused area.

7.4.3. Data collection

Biophysical skin measurements

The primary outcome of this study was callus lesion regression over a 12 week period. Baseline data included skin measurements and shod plantar pressure data. Skin biophysical properties data was collected by one investigator and plantar pressure data was collected by two investigators. The skin measures included skin hydration (Corneometer[®]), skin distensibility (Cutometer[®]) and skin surface topography parameters (Visioscan[®]), as in previous chapters. The skin sites measured included the centre and edge of the callus, skin adjacent to the callus, and two normal weight bearing skin sites including (1) a plantar metatarsal head free from callus, and (2) skin overlying the base of the fifth metatarsal. The skin sites were marked with a ball point pen. Both control sites were identified through palpation and marked. The distance of the radii of the measurement probes was marked over the centre of each skin measurement site to allow accurate probe placement (Figure 7.1).



Figure 7.1 - The measurement sites on the callus include the centre, edge, and skin directly adjacent to the callus; and skin overlying the PMA and base of the 5th metatarsal (adapted from Hashmi and Malone-Lee, 2007 p.253).

Measurements were conducted as in previous chapters: 10 measurements were taken per skin site using the Corneometer® and one image was taken per site for the Visioscan®. For the Cutometer®, 500mbar of negative pressure was applied for 30 seconds, then the skin was allowed to relax for 30 seconds.

Pressure data

The pressure parameter used in this study was peak pressure. It was deemed that using other parameters such as PTI would not be beneficial because peak pressures are most likely the main contributor to callus production in healthy feet (based on the literature), and thus almost all of the pressure data relating to callus has focussed on this parameter. Also, a recent systematic review (Bus and Waaijman, 2013) has shown that reporting PTI is of limited benefit due to the fact that the peak pressure causes the greatest amount of damage to the skin, and PTI strongly correlates with peak pressure data.

The plantar pressure data was collected in the participants' own shoes using Novel Pedar-X in-shoe pressure sensors (Novel, Germany) both without and with the orthotic insoles in situ. These measures were completed at each appointment. Pedar insoles have a matrix of 99 sensors (which vary in size according to the size of insole). The in-shoe

pressure measurements were obtained using the participants' own shoes. The participants were not required to wear the same footwear at each visit, because it was likely that, upon becoming accustomed to the insoles, they might use them in different shoes, and change preference as to which shoes to use the insoles with. This would give a more realistic 'reallife' pressure profile. For that reason, they were asked to bring the shoes they wore the insoles with most often. During data collection, the participants were asked to walk along an eight metre walkway five times at a self-selected walking speed, which was chosen as it would give the most realistic pressure profile for each participant. Timing gates were used to record speed to ensure consistency in the walking trials. The pressure data was analysed using InShoe Pressure Analyser[©] version 1.0 (2012), a pressure analysis programme written in the Foot and Ankle Research department at the University of Salford. The code separates the walking trial into blocks of steps. The first and last three steps were then removed as they represent gait initiation and end. The code then takes the peaks from all the sensors in each mask and then means them, so there is a mean peak pressure for each step, and then it gives a mean peak pressure in each mask for the entire walking trial. The masks include the first metatarsal head, second to fourth metatarsal heads, fifth metatarsal head, hallux and heel.

The peak pressure increase for callused subjects compared with controls is, across the literature, on average in the region of 35% (Potter and Potter, 2000b, Abouaesha et al., 2001, Pataky et al., 2002, Duffin et al., 2003, Menz et al., 2007). One orthotic study conducted in adolescents with plantar callus found a mean pressure reduction of 17.7% using orthotic insoles (Duffin et al., 2003) which is a fairly typical pressure reduction using contoured insoles. It is worth noting that while pressure may decrease as a result of the orthotic, as the skin of the callus becomes more compliant, this might further reduce pressure. In this study, participants were included if they experienced a reduction in plantar pressure under an area of callus due to the insoles, and this site was used for all data collection and analysis. In cases where the participant did not bring suitable shoes for the insoles, a stock trainer was used for data collection.

7.4.4. Statistical analysis

Means, 95% confidence intervals and percentage differences were calculated for hydration, distensibility, topography for all skin sites; and for peak pressure data beneath the region of interest. To assess whether these differences were statistically significant, the repeated measures Analysis of Variance (ANOVA) with post-hoc Bonferroni adjustment,

or Friedman's ANOVA with post-hoc Wilcoxon Signed-Ranks Test were used for parametric and non-parametric data distributions respectively. SDC (calculated from SEM) was calculated to help contextualise changes in skin biophysical data. Statistical analysis was undertaken on SPSS 20.0 and Microsoft Office Excel 2010.

7.5. Results

7.5.1. Participant and pressure profile

Thirty five healthy adults were enrolled into the study. Out of individuals who completed the study, pressure reductions were achieved in 15 with a total of 26 calluses (Table 7.2). Their data was then analysed. This sample of participants was 66.7% female and had a mean age of 53.6 (\pm 15.5). Out of the 26 calluses, 13 were consistent with Merriman's grade 1 'no specific callus plaque, but diffuse or pinch callus, or present in narrow bands' and 13 were consistent with grade 2 'circumscribed, punctuate oval or circular, welldefined thickening of keratinized tissue' (Springett and Merriman, 1995, p.207). On average, participants wore insoles for six days per week and 8.2 hours per day between baseline and week 6, and for 5.7 days per week and 7.5 hours per day between weeks 6 and 12. At baseline, the pressure was reduced at the region of interest by 18%; at six weeks the reduction was 15.2%; and at week 12, the reduction was 13.6%. The most common sites of maximum pressure increase were at the midfoot and fifth metatarsal head. The room conditions were monitored throughout the trial but not controlled. The mean room temperature and humidity at baseline were 21°c and 42.8% respectively; at week 6 they were 21.8°c and 47.4% respectively; and at week 12 they were 21.7°c and 53.6% respectively.

	Mean (SD)
Age	53.6 (± 15.5)
Height (cm)	165.5 (± 11.7)
Weight (kg)	72.5 (± 15.4)
BMI	26.5 (± 5)

 Table 7.2 Pressure reduction group participant profile

7.5.2. Skin hydration data

Figure 7.2 shows the raw data at each time point for skin hydration, Table 7.3 shows the percentage change at each skin site per time point and Table 7.4 shows the percentage change between callus and control sites at each time points. The SDC throughout the trial is shown for each skin site in Table 7.5. None of the skin sites deviated significantly in

their values between any of the time points. At baseline, there was a significant difference between the callus centre and PMA and base of fifth metatarsal skin sites (63.0% and 56.8% difference respectively; p = 0.002 and 0.000 respectively). The differences remained significant throughout the study period and were 62% and 52.5% different respectively at week 12 (p = 0.004 and 0.001 respectively). Similarly the skin of the callus edge was significantly different to the PMA and base of fifth metatarsal at baseline (49% and 40.1% respectively). This difference remained significant for the rest of the trial, and differences of 48.9% and 36.3% (p = 0.027 and 0.006 respectively) were recorded at week 12. Skin adjacent to callus was not significantly different to skin overlying the PMA or base of fifth metatarsal at any point in the trial. The SDC data over the trial ranged from 4.99au (callus centre) to 14.12au (skin adjacent to callus) over the study period.



Figure 7.2 – Raw hydration data

	Callus	Callus	Skin adjacent to	Normal	Normal 5 th met
	centre	edge	callus	PMA	base
Baseline – Week 6 (p	32.3				
value)	(0.178)	9.7 (1.0)	18.8 (0.419)	26.3 (0.370)	40.5 (0.005)
Baseline – Week 12 (p	26.5	23.3			
value)	(0.065)	(0.493)	22.2 (0.219)	23.2 (0.422)	15.1 (0.158)
Week 6 – Week 12 (p		12.4			
value)	-4.4 (1.0)	(0.583)	2.9 (1.0)	-2.5 (1.0)	-18.1 (0.334)

 Table 7.3 Percentage change between time points at each skin site

Table 7.4 Percentage change betwee	n callus and control	sites at each	time point
------------------------------------	----------------------	---------------	------------

	Callus and PMA (p value)	Callus and 5 th met (p value)	Edge and PMA (p value)	Edge and 5 th met (p value)	Adjacent and PMA (p value)	Adjacent and 5 th met (p value)
Baseline	-63.0 (0.002)	-56.8 (0.000)	-49 (0.029)	-40.1 (0.006)	1.15 (1.0)	18 (1.0)
Week 6	-61.2 (0.003)	-59.3 (0.000)	-55.7 (0.015)	-53.6 (0.000)	-4.9 (1.0)	-0.3 (1.0)
Week 12	-62 (0.004)	-52.5 (0.001)	-48.9 (0.027)	-36.3 (0.006)	0.3 (1.0)	25.3(1.0)

Table 7.5 Smallest detectable change for hydration data

	Callus centre	Callus edge	Skin adjacent to callus	Normal PMA	Normal 5 th met base
Mean					
difference	0.91	1.05	3.42	1.06	1.44
SD of					
differences	2.54	3.94	7.20	5.60	4.27
SEM	1.80	2.79	5.09	3.96	3.02
SDC	4.99	7.73	14.12	10.97	8.37

7.5.3. Skin distensibility data

Figure 7.3 shows the raw data at each time point for skin distensibility, Table 7.6 shows the percentage change at each skin site per time point and Table 7.7 shows the percentage change between callus and control sites at each time points. The SDC throughout the trial is shown for each skin site in Table 7.8. There were no significant changes at callus centre or edge site between time points, but there was a significant change at the skin adjacent to callus between baseline and week 6 (7.3% difference; p = 0.04). At baseline, skin at the centre of the callus was 26.5% less distensible than skin of the PMA (p = 0.005). This remained significantly different throughout the trial and the difference was 20.4% at week 12. The skin at the callus centre was 20.7% different to skin overlying the base of the fifth metatarsal at baseline, but this did not reach significance (p = 0.081). The difference increased by week 6, reaching significance (22.9%; p = 0.019) and fell slightly by week

12, but remained significant (16.6%; p = 0.041). There were no significant differences between skin of the callus edge and skin adjacent to callus, and the control sites at any time point. The SDC data over the trial ranged from 0.029mm (normal PMA skin) to 0.63mm (skin overlying the callus edge).



Figure 7.3 – Raw distensibility data

Tε	ıb	le '	7.6	Percenta	age change	between	time	points a	at each	skin	site
					-BB-	~~~~~		P 0 00 1		~	~~~

	Callus	Callus	Skin adjacent to	Normal	Normal 5 th met
	centre	edge	callus	PMA	base
Baseline – Week 6 (p		-9.2			
value)	-7.3 (0.089)	(0.406)	-7.3 (0.040)	-3.9 (0.576)	-4.6 (0.523)
Baseline – Week 12 (p		-7.8			
value)	2.3 (1.0)	(0.376)	-6.5 (0.106)	-5.5 (0.133)	-2.7 (0.697)
Week 6 – Week 12 (p	10.3			-1.61	
value)	(0.055)	1.6 (1.0)	0.9 (1.0)	(0.518)	2.0 (1.0)

	Callus and PMA (p	Callus and 5 th met (p	Edge and PMA (p	Edge and 5 th met (p	Adjacent and PMA (p value)	Adjacent and 5 th met (p
	value)	value)	value)	value)		value)
Baseline			-10.0			
	-26.5 (0.005)	-20.7 (0.081)	(0.298)	-2.9 (1.0)	-0.8 (1.0)	7.1 (1.0)
Week 6			-16.0			
	-29.0 (0.006)	-22.9 (0.019)	(0.465)	-7.6 (1.0)	-4.25 (1.0)	4.0 (1.0)
Week			-12.2			
12	-20.4 (0.026)	-16.6 (0.041)	(0.984)	-8.0 (1.0)	-1.8 (1.0)	2.9 (1.0)

	Callus centre	Callus edge	Skin adjacent to callus	Normal PMA	Normal 5 th met base
Mean					
difference	0.12	0.17	0.12	0.07	0.08
SD of					
differences	0.19	0.32	0.16	0.15	0.16
SEM	0.13	0.23	0.11	0.11	0.11
SDC	0.37	0.63	0.31	0.29	0.31

Table 7.8 Smallest detectable change for distensibility data

7.5.4. Skin homogeneity data

Figure 7.4 shows the raw data at each time point for skin hydration, Table 7.9 shows the percentage change at each skin site per time point and Table 7.10 shows the percentage change between callus and control sites at each time points. The SDC throughout the trial is shown for each skin site in Table 7.11. Statistically significant increases in homogeneity values were apparent between baseline and week 6 for all skin sites, but then plateaued between weeks 6 and 12. Significant differences were also apparent between baseline and week 12 for callus centre (5.7%; p = 0.000) and callus edge (5.2%; p = 0.000). There was no significant change between week 6 and week 12 for any callus site. At baseline, skin of the callus centre was significantly different to skin overlying the PMA and fifth metatarsal base (8.9% and 4.9% respectively; p = 0.000 and 0.021 respectively). These differences remained significant throughout the trial, with differences of 6.4% and 5.6% (p = 0.000and 0.007 respectively) recorded for these sites respectively at week 12. Skin overlying the callus edge was significantly different to skin overlying the PMA at baseline (6.1%); p = 0.003), and remained significant throughout the study period, ending on 4.1% difference at week 12 (p = 0.003). Skin of the callus edge was not significantly different to skin overlying the fifth metatarsal base at baseline (2%), however at week six, the difference did reach significance (3.5%; p = 0.039). By week 12, the difference was not significant, falling to 2.2%. The SDC data over the trial ranged from 0.16au (normal PMA skin) to 0.55au (callus centre skin) over the study period.



Figure 7.4 – Raw homogeneity data

Table	· 7.9	Percent	tage cl	nange	between	time	points	at	each	skin	site
										~	

	Callus centre	Callus edge	Normal PMA	Normal 5 th met base
Baseline – Week 6 (p value)	6.8 (0.001)	4.6 (0.028)	4.8 (0.002)	6.2 (0.011)
Baseline – Week 12 (p value)	5.7 (0.000)	5.2 (0.000)	2.9 (0.056)	5.4 (0.002)
Week 6 – Week 12 (p value)	-1.0 (1.0)	0.6 (1.0)	-1.8 (0.057)	-0.8 (1.0)

Table	7.10	Percentage	change	between	callus an	d control	l sites at	each	time	point
				~~~~~						P ~

	Callus and PMA (p value)	Callus and 5 th met (p value)	Edge and PMA (p value)	Edge and 5 th met (p value)
Baseline	-9.0 (0.000)	-4.9 (0.021)	-6.1 (0.003)	-2.0 (1.0)
Week 6	-7.2 (0.002)	-4.4 (0.010)	-6.4 (0.001)	-3.5 (0.039)
Week				
12	-6.4 (0.000)	-4.6 (0.007)	-4.1 (0.003)	-2.2 (0.566)

<b>Table 7.11</b>	<b>Smallest</b>	detectable	change for	• homogeneity	data
			0		

	Callus centre	Callus edge	Normal PMA	Normal 5 th met base
Mean difference	0.03	0.07	0.06	0.08
SD of differences	0.28	0.09	0.08	0.12
SEM	0.20	0.07	0.06	0.09
SDC	0.55	0.18	0.16	0.24

# 7.5.5. Skin variance data

Figure 7.5 shows the raw data at each time point for skin hydration, Table 7.12 shows the percentage change at each skin site per time point and Table 7.13 shows the percentage change between callus and control sites at each time points. The SDC throughout the trial

is shown for each skin site in Table 7.14. The variance values decreased from baseline to week 6 (changes at each skin site were significant except at the callus edge), but plateaued between weeks 6 and 12. At baseline, skin of the callus centre was significantly more variable to skin overlying the PMA and fifth metatarsal base (38.9%; p = 0.001). These differences remained significant throughout the trial, with a difference of 33.4% (p = 0.000) recorded at the end of the trial. The callus centre was not significantly different to the skin overlying the base of the fifth metatarsal (15.4%; p = 0.066), but this difference increased slightly, becoming significant at week 6 (22.1%; p = 0.004) and remaining significantly different to skin overlying the PMA at baseline (24.0%; p = 0.004). This difference remained significant throughout the trial, ending on 21.4% difference (p = 0.002) at week 12. The skin of the callus edge was not significantly different to skin overlying the fifth metatarsal base at any point during the study. The SDC data over the trial ranged from 0.16au (normal PMA) to 3.47au (callus edge) over the study period.



Figure 7.5 – Raw variance data

<b>Table 7.12</b>	Percentage	change	between	time	points	at	each	skin	site

	Callus centre	Callus edge	Normal PMA	Normal 5 th met base
Baseline – Week 6 (p value)	-19.08 (0.001)	-14.51 (0.099)	-18.89 (0.001)	-23.51 (0.002)
Baseline – Week 12 (p value)	-15.47 (0.002)	-14.0 (0.005)	-12.11 (0.031)	-18.27 (0.001)
Week 6 – Week 12 (p value)	4.46 (1.0)	0.6 (1.0)	8.36 (0.065)	6.84 (1.0)

	Callus and PMA (p	Callus and 5 th met (p	Edge and PMA (p	Edge and 5 th met (p
	value)	value)	value)	value)
Baseline	38.84 (0.000)	15.43 (0.066)	24.03 (0.004)	3.2 (1.0)
Week 6	38.41 (0.001)	22.1 (0.004)	30.74 (0.001)	15.33 (0.063)
Week				
12	33.43 (0.000)	19.38 (0.017)	21.37 (0.002)	8.6 (0.762)

Table 7.13 Percentage change between callus and control sites at each time point

#### Table 7.14 Smallest detectable change for variance data

	Callus centre	Callus edge	Normal PMA	Normal 5 th met base
Mean difference	1.38	-0.07	-0.06	-0.08
SD of differences	1.77	0.09	0.08	0.12
SEM	1.25	0.07	0.06	0.09
SDC	3.47	0.18	0.16	0.24

# 7.6 Discussion

The data from this study suggests over the population of this study (n = 15) that pressure reduction of the scale achieved by the insoles, and over a 12 week period, does not lead to a change in callused skin properties that is different than changes in normal plantar skin sites. If it is assumed that callus and loading are related, there are several possible reasons why the data might show this. Firstly, it could be that the skin of the foot has become accustomed to the level of load it receives on a daily basis and has become 'programmed' to form callus through biochemical changes including keratins, proteins involved in adhesion and differentiation, as highlighted by Kim et al. (2010) and discussed in Chapter 2. It might be that reducing the load on the skin is not enough to reverse this process once the biochemical triggers have become well established. If this is indeed the case, it can be suggested that callus is indeed a chronic skin lesion.

The above reason however assumes that the level of pressure reduction was sufficient, but in reality it might be the case that a larger reduction in pressure is a requirement to reverse the hyperkeratotic process. Whole foot and regional data from previous studies suggests that peak plantar pressures at sites of calluses is high compared to controls, ranging from 5% to 145% greater peak plantar pressures (Potter and Potter, 2000b, Abouaesha et al., 2001, Pataky et al., 2002, Duffin et al., 2003, Menz et al., 2007). As these studies collected data from both specific regions and the whole foot, it can only give an example of what pressure increases might be occurring specifically as a result of

callus but does highlight the possibility that in order for pressure reduction to cause a callus-reversing effect, the reduction in pressure has to be large. In Chapter 6 (skin loading pilot study), it was shown that increases in peak pressure (ranging from 13.6% to 28%) over a prolonged period did not lead to any changes in plantar skin properties. In this study, the magnitude of the change in peak pressure changes was similar. The mean reduction in pressure at the first appointment was 18%, 15.2% at week 6 and 13.6% at week 12. Whilst these reductions are in line with those in the literature, and thus the orthosis design was appropriate, perhaps a more aggressive approach to pressure reduction might be necessary to drive some form of skin change. In diabetes, there are several methods of offloading with the most notable examples including rocker shoes in prevention and total contact casting as a treatment for plantar ulceration, both of which can substantially reduce pressures by as much as 39% and 70% respectively (Burns and Begg, 2011, Chapman et al., 2013). While the populations in these studies may have had very high peak pressures to begin with, making pressure reduction easier to achieve, it is likely that offloading techniques like these would reduce pressures beneath callused skin. These would allow the greatest chance of skin change, but are largely impractical and perhaps unethical to use in healthy individuals given the aesthetic and mobility issues associated with them. In this study, the pressure reduction data included cases where the actual regional reduction was small. An alternative approach would have been to set a minimum pressure reduction required for inclusion in the study.

One issue affecting plantar pressure and skin in 'real world' is activity level. One may assume that the activity levels remain consistent throughout the study, but in reality these could and are likely to vary dramatically. This would not necessarily affect the reduction in pressure caused by the insole, but will increase cumulative loading on the skin to the point that it is actually being loaded more than before enrolment into the study. This could lead to an increased rate of hyperkeratosis and thus have affected skin data, cancelling out any effect of reductions in peak plantar pressure. Monitoring subjects using an activity monitor, or asking them to fill out questionnaires relating to physical activity levels would give information about how much loading the foot might experience (as undertaken in the previous chapter). However, the benefit of this would be minimal as in reality there is no way of controlling how much an individual's foot is loaded outside the laboratory setting, so it has to be accepted that the foot is never loaded in a homogenous manner over time. It could only assist in explaining the data. Similarly, while pressure data at each appointment was collected in a stock shoe and in participants' own shoes, there is no way of knowing what pressures are like in each pair of shoes the individual wears without testing them all of them – this is unrealistic due to the amount of time it would take to conduct all of the measurements. There is no control over which shoes individuals wear outside of the study, so again it must be accepted that different shoes with the insoles may distribute pressures differently beneath the feet, and potentially increase them in some cases. This could also have had an impact on skin data.

Another possible issue with this study is that it was conducted over a 12 week period. As this study was to an extent exploratory, there is no gold standard duration of insole wear which will lead to an effect on the skin and it is likely individuals will respond differently due to anatomy, wear time, skin properties and other factors. In this study, 12 weeks was deemed appropriate because it would allow approximately three cycles of skin cell turnover and thus theoretically enough time for some skin change to occur. Certainly, previous studies which have manipulated skin loading conditions have noted marked changes within 35 days (Rubin, 1949, Mackenzie, 1974a, Mackenzie, 1974b, Goldstein and Sanders, 1998). However, these studies were applying loads to the skin, not reducing them, so it could be that the skin's response to reductions in pressure is a process that takes longer to appear. It would have been interesting had any of the aforementioned authors studied how quickly the skin, having been loaded, recovered. This may have given a more realistic idea of how long to allow for skin changes.

The previous studies which have reported callus regression from insole wear (Colagiuri et al., 1995, Duffin et al., 2003) were undertaken over a period of 12 months which would allow a greater time for any effect of pressure reduction on callused skin to manifest. However, the participants in these studies had diabetes which means that a longer timeframe might be needed due to possible vascular impairment to the lower limb, and as these studies were conducted in hospital diabetes clinics, they would have had access to a large number of potential participants. In this study, 12 weeks was deemed appropriate due to the fact that keeping participants in a longitudinal study can be difficult, and given the low volume of potential participants, it was deemed important to maximise the chances of keeping them in the study. There are also possible pitfalls with extended insole use which are only just starting to emerge, such as an effect on soft tissue thickness in the foot (Sweeney et al., 2014), plantar somatosensory sensations (Vie et al., 2015), and possible changes in foot shape and function (D'Août et al., 2009).

The previous insole studies which reported callus reduction (Colagiuri et al., 1995, Duffin et al., 2003) did not report pressure data at follow-up time points, nor did they measure skin properties. While it can be suggested that pressure reduction may have been a factor in reducing callus, other factors may also have contributed to their findings. This study found that hydration and roughness over all the skin measurement sites improved slightly over the duration of the study. This is unlikely to be a product of pressure reduction because pressure reduction did not occur at each skin site. The most common areas of pressure increase included the midfoot and surrounding metatarsal heads. It is more likely that the skin improved as a result of either increasing temperature of the environment (the study was conducted between winter and summer, 2015) which caused an increase in heat inside the shoe leading to greater sweat production – studies have shown that skin hydration and mechanincal properties at non-plantar sites can be affected by seasonal variation (Cravello and Ferri, 2008); or a result of the material used for the insole itself causing an increase in heat and causing some degree of skin occlusion which has been shown to increase hydration (Zhai and Maibach, 2002). Some participants in this study mentioned that their feet felt 'warmer' and 'clammier' with the insole inside the shoe and this is likely to be due to the material. The materials used in this study were EVA and rubber. While there is literature investigating insole materials (reviewed earlier), they look at pressure reduction properties and not the effect of the materials on skin. This might be an interesting area of future study – general skin improvement using insoles could have clinical significance and might be helpful in individuals where pressure reduction is not a necessity.

One hypothetical reason why the callused skin did not regress significantly could be a result of the pressure reduction itself. Perhaps the offload of the callus was such that the most superficial layers of dead keratin, as is found in thick plaques in callused skin, were retained. Walking could be a mechanism for shedding the superficial keratin sheets through rubbing at the foot-shoe interface, but since the callused skin did not regress, it might mean that pressure reduction decreased the amount of skin shedding that may have otherwise have occurred had the pressure remained the same. There is no way of studying this other than through biopsies taken before and after the study and histologically analysing the superficial keratin layers of the stratum corneum. However, taking biopsies would be painful and thus may affect gait (as callus occurs at weight bearing areas). It would thus be unethical to do so on any more than one occasion and would likely severely impact upon recruitment. It might be that the severity of the calluses used in this study was such that pressure reduction was unable to reverse the process of hyperkeratosis, at least in this timeframe. Due to the relatively low volume of recruitment, individuals with calluses of all grades were enrolled into the study. If recruitment was more successful, it might have been possible to enrol only individuals with relatively light calluses which may have shown a response within the timeframe, compared to pressure reduction than thicker calluses which may require longer.

A potential area of weakness with the study methodology is that detailed activity data was only collected at baseline, as part of a questionnaire collecting participant background information. It may have been useful to detail whether or not activity levels increased over the duration of the study. If pressure reductions were achieved but the total accumulative load experienced under the region of interest increased, this could counteract the potential benefit of the pressure reduction causing increased total pressure on the skin. While it is unlikely that each participant would have dramatically increased the volume of exercise taken over the study period, it cannot be discounted as a potential factor affecting the results, particularly as weather conditions improved throughout the study (which took place from winter to summer). This methodological error is one that must be taken into account in future research studies where accumulative loads could affect data.

Other potential areas affecting the data lie with study recruitment and participant compliance. This study formed part of a larger project looking at insole pressure reduction properties and durability over time and cases selected for analysis in this study were drawn from a larger collection of data. Due to the low recruitment volume, the researcher was forced to be less selective in cases included in the final data analysis in terms of pressure reduction and compliance. While the mean pressure reduction and compliance was relatively good across the dataset, in a few cases they were small. Also, it must be stated that in a study like this, compliance stated by the participant does not necessarily reflect reality. There is always the possibility that participants can over-estimate the amount of time wearing the insoles, or due to fear of affecting the project, give false answers. While participants were encouraged to be honest in their responses and were reassured that there would be no repercussions regarding poor compliance, there is still the possibility of feedback lacking accuracy. There is no way this can be avoided other than participant reassurance and regular contact with them, both of which were done in this study.

The final area which warrants discussion in this chapter is the normal variation of the skin between data collection points. As found in Chapters 3 and 6, the SDC data suggested that in order for a change in properties to be considered 'real', it would have to be relatively large. If small skin changes in the callused regions did occur as a result of pressure reduction, there is a risk that these changes may have been less easy to identify given that normal variation in skin properties occurred over time. However, to minimise this effect, numerous skin sites were chosen with two control sites. Variation could affect the measurements longitudinally, but when performing comparisons between skin sites at each time point, any skin effect caused by the pressure reduction would likely show in the data as a change in callus skin properties in relation to the other skin sites. A very obvious change could not be explained by natural variation, and this type of skin testing has proved successful in studies on callus and heel fissure treatments, studies that are currently awaiting publication (Hashmi et al., 2015). This study showed that increases in hydration of up to 152%, and distensibility of up to 100% were possible in callused skin sites, before and after podiatry treatment. This magnitude of skin change is far larger than can be attributed to normal skin variation, and these changes were obvious in relation to the control site. Likewise, other studies on pedal skin, mentioned in previous chapters, have noted large increases in hydration post-treatment. Increases in hydration of up to 57.3% were reported by Garrigue et al. (2011) after 28 days, and 20.6% increase in hydration after 14 days were reported by Papanas et al. (2011). It is clear that pronounced skin changes would clearly show in the datasets, and there was no evidence of any callus skin reduction occurring in this study.

While the results of this study provide only a starting point from which further research can be developed, it might be worth noting that using insoles as a treatment modality for existing calluses would likely be improved when used in conjunction with other treatments aimed to remove the actual skin, such as debridement. It is possible that debridement of the lesion, which may provide some degree of pressure relief (Pitei et al., 1999, Pataky et al., 2002), followed by insole wear might have resulted in a slowing of callus regrowth by removing the mechanical stimulus and a later study could investigate this. It might be that using insoles as a preventative measure for plantar callus regrowth in conjunction with better established primary treatments might be more effective than using insoles alone.

# 7.7 Conclusion

As with previous sections of this thesis, this chapter demonstrates the author's ability to design, implement and evaluate research in autonomously. Like Chapter 6, reasons for a nil-response of the skin to changes in loading conditions are evaluated and discussed, and learnings which can be applied to further research have been made explicit. To summarise, the results from this chapter suggest that pressure reduction may not be enough to reverse the skin changes associated with callus. The results should be treated with caution as there are several factors which may have influenced them. The pressure reduction magnitude and/ or the duration of insole wear may have been insufficient; the possibility that pressure reduction may have slowed desquamation; and the callus grades used in the study are all factors which should be taken into account. These results may be useful in clinical practice when insoles are considered for callus treatment. Pressure reduction may have more benefit as a preventative measure as opposed to a primary treatment so combining insole prescription with debridement could be considered until more evidence is available supporting pressure reduction as a means of callus reduction.

# **Chapter 8: Overall discussion and conclusion**

# 8.1 Summary of findings

There were two main aims to this PhD: (1) to investigate the biophysical properties of callused skin; and (2) to quantify the skin's response to changes in external loads specifically, whether callus results from increased pressure and whether callus regresses from pressure relief. To accomplish these aims, six investigations were necessary: (1) a review of the literature surrounding callus and skin measurement; (2) an investigation into reliability of skin measurement devices; (3) an investigation into the biophysical characteristics of normal and callused plantar skin; (4) development of a novel device designed to load plantar skin; (5) the effect of loading normal plantar skin in individuals prone to plantar callus; and (6) the effect of pressure reduction beneath plantar callus.

The literature review found that callus is a common clinical and foot health problem. There is anecdotal evidence that one of the causative factors is increased plantar vertical pressure and shear pressure acting on the skin, and there is a small body of literature which has focussed on non-pedal (Rubin, 1949, Goldblum and Piper, 1954) and animal (Mackenzie, 1974a, Mackenzie, 1974b, Goldstein and Sanders, 1998, Sanders et al., 2002) skin sites which have shown how the skin adapts to external loads. These studies show that skin thickening can occur in a short space of time (within 35 days) given the right conditions. However, these studies are cited in the callus literature, but cannot be applied to the foot because the skin sites used are completely different anatomically and functionally to plantar skin, and are not designed for bearing loads. Plantar pressure studies have shown that peak pressures tend to be significantly higher beneath callused skin sites compared with control subjects (Potter and Potter, 2000b, Pataky et al., 2002, Duffin et al., 2003). However, conflicting evidence exists about whether or not the plantar pressure is the precursor to, or a result of callused skin (Pitei et al., 1999, Potter and Potter, 2000b, Woodburn et al., 2000, Pataky et al., 2002). Individuals with callus have been reported to also present with factors such as limited joint mobility, deformities and decreased tissue thickness which may result in increased pressures (Bevans and Bowker, 1999, Abouaesha et al., 2001, Menz et al., 2007), but no studies have attempted to investigate specifically whether a link between plantar pressure and callus exists, therefore this thesis aimed to address this. In addition to identifying the gap in the literature between callus and pressures, the review also identified that there are a range of potential measurement modalities that could be used in plantar skin testing, some of which could be beneficial in

callus profiling. Several instruments were highlighted which could be of potential use in the reliability study and the skin characterisation study.

The reliability study presented in Chapter 3 investigated several devices highlighted in the literature review which could be beneficial in testing callus. The Corneometer[®] (hydration), Cutometer[®] (distensibility -a surrogate measure of stiffness) and Visioscan[®] (roughness) were chosen for investigation. Eight volunteers with plantar forefoot callus were tested over two days by two investigators. Skin tests were undertaken on callused and non-callused sites. It was found that generally the inter-rater and inter-day reliability and agreement across each of the instruments was good but there were some discrepancies which were discussed. The reliability literature available for these devices is scant, but those studies which do exist generally agree with the results in this study. The ICC values for the Cutometer[®] on callused skin were similar to the values reported by Fong et al. (1997) and Draaijers et al. (2004) on scar tissue for inter and intra-rater reliability tests. Visioscan[®] data showed that the values reported in this study were slightly lower than those reported by Kottner et al. (2013) but these can be attributed to differences in methodology. As the instruments used in this study were able to identify differences between each skin site, these methods of skin profiling were deemed suitable for use on plantar skin, so long as control sites are also used as a reference, particularly when collecting longitudinal data. This study adds to both the literature in biophysical measurements and in foot research by identifying devices that can reliably measure plantar skin - this has only previously been done with one device (Hashmi and Malone-Lee, 2007). This will allow clinicians and researchers to measure plantar skin with these devices, with the knowledge that they can provide reliable measurements.

The skin characterisation study presented in Chapter 4 investigated the biophysical profile of callused and non-callused plantar skin. Forty six individuals with 61 calluses were recruited onto the study and data was collected at one appointment by one investigator. Measurements included hydration, stiffness and roughness parameters. The results showed that callus is significantly less hydrated, less distensible (more stiff) and less homogenous and more variable in appearance (rougher) than skin adjacent to the lesions and normal plantar skin of the PMA and base of the fifth metatarsal. There were also some differences noted between control sites. The parameters of hydration and distensibility were moderately and significantly correlated on callused sites.

This study adds to the general literature of skin biophysical characterisation, of which there are only a few studies (Ryu et al., 2008, Krueger et al., 2011, Luebberding et

al., 2014). While they report differences between regions, only two tested these for significance (Ryu et al., 2008), the others focussing more on age related changes. One very recent study by Nam et al. (2015) adds to this literature. The authors investigated seasonal variations at different skin sites. Using the Corneometer[®], they identified that skin of 'crow's feet' was significantly more hydrated than skin of forearm and cheek (p < 0.05) throughout the trial. Using the Cutometer[®], they observed significant differences in mechanical parameters between the forearm and the other skin sites (p < 0.05) throughout the study period. Skin scaliness (using Visioscan[®]) was significantly greater at the forehead compared to other skin sites (p < 0.05).

At the time of writing, no studies have attempted to characterise plantar skin using a range of parameters, and the study in Chapter 4 adds to this gap in the literature. These results in this study can be linked to the findings of previous immunohistochemistry papers focussing on callus. The increase in stiffness may be due to an increase in the expression of keratins and proteins involved in cohesion and differentiation which would increase the rate of keratinisation process (Kim et al., 2010) leading to a build-up of hard skin. The increased rate of proliferation may result in poor differentiation (Thomas et al., 1985) leading to intercellular lipids becoming fragile (Harding et al., 2003), impairing the skin barrier and leading to a reduction in hydration (Wickett and Visscher, 2006, Baroni et al., 2012). The rougher appearance of callused skin may be a result of altered desquamation in the stratum corneum, reported by Thomas et al. (1985). The results of this chapter make explicit the biophysical differences between normal and callused skin which can be used as a benchmark for further research studies which might look to 'change' callused skin properties through treatment.

With Chapter 4 providing data on differences between callused and normal skin sites, further work could then be done to establish whether a relationship between pressure and callus exists. Previous work investigating skin adaptation to external loads has work has been undertaken in humans (Rubin, 1949, Goldblum and Piper, 1954, Pinkus, 1952, Brophy and Lobitz, 1959) and animals (Mackenzie, 1974a, Mackenzie, 1974b, Goldstein and Sanders, 1998, Sanders et al., 2002) but never for investigating plantar skin properties. Therefore, Chapter 5 presents work undertaken to develop a device to apply loads to plantar skin in a safe and repeatable manner for use in a later skin loading study. The device in question needed to be powerful enough to deliver large forces to the foot in excess of what would normally be experienced during gait. The device built comprised of a horizontal and vertical actuator to allow device movement in these directions, and a

horizontal and vertical load cell to measure the forces in these directions. Tests using static weights confirmed that the load cells were accurate and consistent but when a horizontal force is added, there are some small discrepancies in vertical readings. These are discussed in the chapter but were not deemed a problem given that the work for this PhD was to focus on the delivery of vertical pressure. A computer was built for the device with a number of loading cycles built in to instruct the device as to which movements were needed. For the purposes of this thesis, the 'compression' programme was deemed the most important and was therefore tested on plywood and medium density EVA. The results showed a linear relationship between air pressure delivered to the actuator and the resulting force applied. Furthermore, the device was capable of producing very repeatable loads during each compression cycle and was therefore deemed suitable for use on plantar skin.

Chapter 6 focussed on using the loading device on plantar skin in a small pilot test over a minimum period of six weeks. Preliminary testing aimed to fine tune protocol considerations including foot positioning and force fine tuning on plantar skin which showed that compression forces could be applied within 10% of the target load. Five individuals (one of whom dropped out) with plantar callus were recruited onto the study. The skin loading site in all cases was a callus-free area either at the second or fourth metatarsal head. To establish a target load, in-shoe pressure measurements were obtained using the Pedar X measurement system and the target load was calculated. During the study activity monitoring data using ActivPALTM and barefoot pressures measured using Novel EMed[®] were also collected for calculating a loading profile for each participant to conceptualise the load applied with the device. Each individual attended three appointments per week for a minimum of six weeks and the target load was set at 2-3 times normal pressure for the loading site but adjusted based on participant feedback for comfort. Skin data as described previously was collected at weekly intervals, then at one appointment four weeks after the skin loading was concluded. The data showed that no real skin change occurred as a result of the skin loading. Since the percentage weekly load applied to the participants' feet using the device was relatively large, it suggests that a longer duration of study or the introduction of shear force in subjects recruited from narrower criteria might be necessary to elicit a skin change toward that of callus. Safety was also discussed and suggestions such as recruiting individuals who exercise regularly, who may have plantar skin accustomed to larger loads, or beginning with a relatively small load and slowly increasing this over a period of weeks to allow the foot to become accustomed to it, were made. Other factors such as the day-to-day variability of the skin's properties and the limitations measuring longitudinal changes in skin were also discussed and it was concluded that while variation of the skin could arguably make changes at the load site difficult to see, the use of control sites essentially minimised the risk of this affecting the results by providing data for comparison.

The final study in this thesis, Chapter 7, focussed on a 'real world' scenario looking at pressure reduction beneath callused skin using insoles. Each participant was given a pair of insoles to wear for a period of 12 weeks. 15 individuals with pressure reductions at 26 callused sites were included in the final analysis. Skin data using the same devices used in the other chapters were collected at baseline, six weeks and 12 weeks into the study. Plantar pressures were monitored in participants' own shoes and stock shoes using Novel Pedar X at these same appointments. The results suggested that plantar pressure reduction was not enough to cause a callus skin change. The reasons for this are multifactorial. Potentially, the pressure reduction was inadequate. At week 1, mean peak pressure reduction was 18%, at week 6, 15.2%, and at week 12, 13.6%. While variable, probably as a result of different footwear and insoles 'wearing in', these reductions are relatively high. It may be that the duration of insole wear was insufficient, the duration of the study was inadequate or the magnitude of pressure reduction itself was not high enough to trigger a change in callus properties. Normal variations in biophysical properties were observed across skin sites, but it was concluded that the use of control sites essentially minimised any negative effect on identifying callused skin changes. Other areas such as reduced desquamation as a result of the pressure reduction, the nature of the calluses themselves, and activity levels were also highlighted potential factors affecting the results.

Previous studies (Colagiuri et al., 1995, Duffin et al., 2003) have reported reductions in calluses as a result of insole wear, but the study presented in this chapter takes this further by specifically testing the skin to determine whether, over a 12 week period, pressure reduction leads to a change in callus skin properties. This chapter also reports pressures at each data time point which was not done in either of the previous papers but is important in ensuring that the intervention is working as intended.

### 8.2 Implications for clinicians and researchers

The research conducted for this PhD thesis has created knowledge of plantar callus from both a physiological and biomechanical standpoint. No previous work has been conducted that has both validated and then used a range of instruments to measure plantar skin biophysical properties. This is a small but necessary breakthrough in plantar skin research. While it has been postulated that callus and normal skin differ in terms of hydration, stiffness and roughness, measuring skin in-vivo without having to surgically remove it is beneficial in both single data collection appointments and particularly for longitudinal studies. It has confirmed the major differences between normal and callused skin. These methods of skin measurement could be far reaching in clinical research and practice, but despite the reliability of these methods, the normal variation of the skin can still affect the data, therefore the use of control sites which show consistent differences from the treatment site is essential. The reliability study from Chapter 3 can be used to inform other authors wishing to undertake plantar skin research as to the value of potential devices. As in other studies highlighted in the literature review on non-plantar skin, these devices could be used to characterise a whole range of skin disorders which may be beneficial in research and practice. For instance, skin measurement could be used to evaluate treatments such as therapeutic insoles for skin problems, as has been conducted in the final study of this thesis, or other modalities for different conditions. Differences between skin sites at baseline and after treatment could give an indication as to whether the treatment was successful, or if not, why it was not successful. Some of the work undertaken for this PhD was part of a larger project leading to papers published in instrument reliability (Hashmi et al., 2015b), skin characterisation (Hashmi et al., 2015a), and the efficacy of clinical and commercially available treatments for callus and heel fissures (Hashmi et al., 2015, papers under review) and the devices were able to successfully detect differences before and after treatment leading to more information about the value of the treatments. While this was in a research setting, these devices could theoretically be used in clinical practice also, in order to track the progress of podiatric treatments for xerotic skin disorders.

The skin loading study in this PhD (Chapter 6) is the first to try and develop a callus-like skin response mechanically in the laboratory setting. It is also the first to contextualise load applied in the laboratory as a percentage of weekly load experienced under the foot, calculated using activity monitoring data. While unsuccessful in defining a

link between callus and pressure, it provides readers with more information about callus, and potentially how complex a condition it is. The final study showed no skin change in callus as a result of pressure reduction. This study was the first to use skin measurements and pressure reduction measures to track callused skin during an insole trial. While several factors discussed previously might have affected the results, the study highlights the possibility that simply removing a mechanical stimulus is not enough to reverse the callused skin response. These two biomechanical studies could potentially benefit clinicians by challenging the 'traditional' beliefs of callus and pressure. While there is almost certainly a link between the two to some extent, it may be naïve to consider that pressure alone is enough to cause callus, and conversely, removing the mechanical stimulus and thus offloading the callus, will be enough to reverse the hyperkeratotic process. Perhaps the biochemical process is somewhat irreversible. However, there may potentially be a benefit if pressure reduction is used in conjunction with a primary treatment such as scalpel debridement. This may be an interesting avenue of research if one were to compare callus regrowth in debridement alone and debridement plus pressure reduction.

# 8.3 Recommendations for further research

The reliability study presented in Chapter 3 found that while reliability was good across the instruments, some variation was evident, particularly between days. Inter-rater and inter-day reliability was appropriate to test due to the intended use of the measurement devices. However measuring intra-rater reliability, involving separate measures at the same skin sites for one investigator during a single testing session, might have shown improved reliability scores, and would also have shed further light on the reliability and agreement measures of the devices. Comparing inter-day and intra-rater reliability, for example, would allow one to assess how much more variation might occur between days than within a single session, and in doing so, be able to better explain data from future studies.

The skin biophysical characterisation study presented in Chapter 4 showed significant differences between callus and non-callus skin sites in hydration, distensibility, homogeneity, and variance parameters. The main limitation associated with this study was the fact that no method was employed to visualise a cross-sectional area of the skin, which could have then been linked to biophysical properties. Researchers wishing to add to this

work may wish to seek a method of visualising a cross-sectional area of skin, whether this is by biopsy or by imaging, as it would be beneficial in linking biophysical properties to actual measures of anatomy. Similarly, cross-sectional anatomical measures would be beneficial in other types of study, such as the skin loading, and pressure reduction methods employed in this thesis. However, as previously stated, biopsies would cause discomfort and would thus affect participant recruitment so employing imaging techniques, such as OCT, would be the most beneficial.

The second limitation associated with this study was that 80% of the sample was female, which could have affected data, given that male and female skin may differ in properties (Luebberding et al., 2013, Luebberding et al., 2014). Researchers attempting similar work may want to address this by ensuring participants in the study are gender-matched. This would allow researchers a more accurate profile of normal and callused skin, and would also allow comparisons in biophysical plantar skin properties between genders, which might lead to other interesting skin studies.

Future researchers may also wish to include a control group of age- and gendermatched individuals without callus for comparison with the callus group. Comparison of general skin quality may be beneficial and allow inferences about whether individuals prone to callus display certain skin properties compared to those without. This could benefit future research and clinical practice by identifying individuals who might be at risk of callus development, based on their skin properties.

The study which assessed the skin's response to vertical pressure application (Chapter 6) found no effect of increased pressure on skin properties. Several factors which could have affected the data include dose of load (both magnitude and duration over which it was applied in a single session), number of sessions per week, and the direction in which the load was applied. It is unlikely that increasing the magnitude of pressure applied to the foot would be possible, and might be unsafe, and it is equally unlikely that participants would commit to more than three load application sessions per week. The most effective study design would be to change the direction of pressure applied to the foot, instead of applying vertical pressure, applying it horizontally and test the skin's response to shear pressure (Rubin, 1949, Mackenzie, 1974b, Mackenzie, 1974a) so this is a pertinent area to investigate.

Several device modifications would be needed (Appendix 1) if using the existing device, to better control the pressures applied and improve the horizontal movement of the
device. Researchers may wish to test the effect of shear with the contact pad rubbing over the skin (often referred to as 'friction') with a low vertical pressure, and with a higher vertical pressure where there is more resistance preventing the contact pad sliding over the skin. A series of case studies looking at the effects of these in different participants who have been carefully selected (such as over 40 years old, female and with some evidence of plantar callus) would allow comparison between these two shear components. A study such as this would add to the research in this thesis and might show different results, and improve understanding of callus.

The study presented in Chapter 7, which looked at the effect of reducing pressure beneath plantar calluses, found no skin response over a 12 week period. The factors which might explain this nil-response are the duration of the study and magnitude of pressure reduction. These are areas which can be addressed in further studies. In addition to increasing the duration of the study which might enable a greater chance of detecting skin change, methods of ensuring pressure reduction is substantial would also be beneficial. Researchers may wish to further reduce pressure, either by changing the geometry of the insole, such as increasing the arch height, or by changing the material to one which has been shown to be effective in pressure reduction, such as poron with plastazote (Tong and Ng, 2010), which has also been shown to retain its pressure reduction qualities after increased wear (Rogers et al., 2006). Also, using a larger sample size will enable researchers to be more selective in data that they analyse. It would be beneficial in this type of study to set a threshold for pressure reduction, e.g. a minimum of 10% pressure reduction beneath callused skin, which would mean that all data analysed are from calluses which have experienced a substantial reduction in pressure compared to a shoe-only condition.

One other area which is worth considering is the use of a control group. The pressure reduction study in this thesis identified that while no specific skin response to callus was detected, the overall quality of the skin improved slightly. This could have been due to seasonal variation or the effect of the insole on the skin, causing some degree of occlusion. Using a control group, a group of individuals with callus who receive no pressure reduction measures, would allow researchers to test the effects of the pressure reduction, by comparing callused skin properties between groups; and the effects of the insole itself on the skin, by comparing all skin sites between groups. In addition, if all skin sites improved between the insole group and the control group, this might suggest seasonal

variation is a key factor in skin change. This type of study would thus have direct clinical consequences.

#### 8.4 Conclusion to Ph. D. Thesis

This thesis presents several findings which include the biophysical properties of normal and callused skin, the nil-response of normal plantar skin to vertical pressure application, and the nil-response of callused skin to pressure reduction measures. Some of these findings may challenge the supposed link between pressure and callus, but cannot be generalised to the whole population – further research is needed to better understand the role that loading has in callus production. While none of the research conducted for this thesis has aimed to change practice – an area which may be hard to influence due to routines, cognition, attitudes and motivation (Grol and Grimshaw, 2003) – some of the research presented here might be enough to allow clinicians to think about the way they approach plantar callus and possibly influence existing knowledge of their pathophysiology. The work conducted for this PhD thesis has provided a good starting platform from which more research into the physiology and biomechanics of plantar callus can now be conducted.

The learnings from conducting this PhD have been substantial. Chapter 2 greatly improved knowledge and understanding of plantar skin physiology and biomechanics, and allowed an understanding of the main research problems surrounding plantar callus. Chapter 3 allowed the author to gain experience in research design, implementation and evaluation, and specifically biophysical plantar skin measurement methodology and statistical techniques. Understanding device limitations was critical in evaluating future data and making informed decisions. Chapter 4 allowed the author to further the body of knowledge in plantar skin physiology by conducting original research using the measurement techniques and learnings from Chapter 3. Chapters 5 and 6 involved design, implementation and evaluation of an experimental research project using new techniques, looking at the effect of increasing peak pressure under the plantar metatarsal area. It allowed the author to make informed decisions on methodology and a thorough evaluation of results using previous learnings. Chapter 7 demonstrated the author's ability to conduct a research project investigating plantar skin biophysical properties and biomechanics, this time from a different perspective – looking at skin properties after reducing peak pressure beneath callused skin. Again, the author was required to thoroughly analyse and evaluate

the results to explain a nil-response. This PhD has allowed the author to fulfil the criteria outlined in the Descriptor For A Higher Education Qualification At Level 8: Doctoral Degree (QAA, 2008, p.23) and has contributed to many transferable skills which can be used to fulfil a career in research and other academic professions.

# Appendix 1: Considerations for loading device improvements for further research projects using shear pressures

## Improve shear movement

Preliminary tests with the foot loading device showed that shear movement was very difficult to achieve, due to the fact that a dragging effect occurred. This occurred even on smooth surfaces. Several possible areas to consider include structure and function. The device components are made of aluminium. Replacing the components with larger, more robust components made of a stronger material, such as titanium, might help to reduce device compliance. This would also allow more accurate compression load cell measurements during the movement as device compliance would not have as much of an effect on the load cell.

If dragging still occurs, the researcher might wish to consider the precision of device positioning relative to the foot. The horizontal movement might be affected by indentation of the device contact pad being too deep in the material. A system which enables fine-tuning of the position of the device in the vertical direction and also takes account of the material's elasticity could be considered. An adaptation which allows measurement of the indentation depth would allow optimal control of device positioning and may reduce any dragging effect.

If these adaptations do not improve the quality of the shear movement, this may indicate that the actuator is not powerful enough to deliver shear at the intended compression. If reducing compression to minimum levels (and thus measuring rubbing effects) is not desirable, the researcher might wish to consider fitting a more powerful shear actuator to the device. Any work done on human feet after such an adaptation should be approached with care to avoid injury by using too high a shear pressure for the foot to tolerate.

#### Improvements in the accuracy of pressure delivery

In its current state, the air pressure to the device's actuators is controlled manually using pressure regulator gauges. To improve the design to allow more precise control of pressure delivery, the researcher might wish to employ a closed feedback loop system. Output signals from the load cells could be delivered to a conditioning unit. The conditioning unit would in turn send signals to an electronic pressure control system which would

automatically adjust the pressure, based on feedback from the load cells, to ensure that the correct pressure to the foot is being delivered. This method would allow excellent control of the pressures delivered, but might be expensive. An alternative method of fine tuning pressure delivery would be to use a pressure transducer which converts the air pressure reading into a digital display, rather than using a mechanical pressure regulator gauge. This would allow for more accurate manual adjustment of pressure settings.

## Improvements in controlling loading cycles

The device is controlled by a box with a series of pre-programmed options for loading cycles. For greater usability, further researchers might wish to employ a system whereby they could build their own loading sequences using a simple computer language.

# References

- ABOUAESHA, F., SCHIE, C. H. M., GRIFFITHS, G. D., YOUNG, R. J. & BOULTON,A. J. M. 2001. Plantar tissue thickness is related to peak plantar pressure in the high-risk diabetic foot. *Diabetes Care*, 24, 1270-1274.
- ACTIS, R. L., VENTURA, L. B., LOTT, D. J., SMITH, K. E., COMMEAN, P. K., HASTINGS, M. K. & MUELLER, M. J. 2008. Multi-plug insole design to reduce peak plantar pressure on the diabetic foot during walking. *Medical & Biological Engineering & Computing*, 46, 363-71.
- ADAMS, B. B. 2002. Dermatologic disorders of the athlete. / Problemes dermatologiques de l'athlete. *Sports Medicine*, 32, 309-321.
- AKDEMIR, O., BILKAY, U., TIFTIKCIOGLU, Y. O., OZEK, C., YAN, H., ZHANG, F. & AKIN, Y. 2011. New alternative in treatment of callus. *Journal of Dermatology*, 38, 146-50.
- ALTAMURA, D., ALTOBELLI, E., MICANTONIO, T., PICCOLO, D. & FARGNOLI, M. C. 2006. Dermoscopic patterns of acral melanocytic nevi and melanomas in a white population in central Italy. *Archives of Dermatology*, 1123-1128.
- ARMSTRONG, D. G. & LAVERY, L. A. 1998. Elevated peak plantar pressures in patients who have Charcot arthropathy. *Journal Of Bone And Joint Surgery*, 80, 365-369.
- ARMSTRONG, D. G., STACPOOLE-SHEA, S., NGUYEN, H. & HARKLESS, L. B. 1999. Lengthening of the Achilles Tendon in Diabetic Patients Who Are at High Risk for Ulceration of the Foot*. *Journal of Bone and Joint Surgery*, 81, 535-8.
- BAE, J. M., KANG, H., KIM, H. O. & PARK, Y. M. 2009. Differential diagnosis of plantar wart from corn, callus and healed wart with the aid of dermoscopy. *British Journal of Dermatology*, 160, 220-2.
- BALANOWSKI, K. R. & FLYNN, L. M. 2005. Effect of painful keratoses debridement on foot pain, balance and function in older adults. *Gait Posture*, 22, 302-7.
- BAREL, A. O. & CLARYS, P. 1995. Measurement of epidermal capacitance. *In:* SERUP,J. & JEMEC, G. B. (eds.) *Non-invasive methods and the skin.* USA: CRC Press.
- BAREL, A. O. & CLARYS, P. 1997. In vitro calibration of the capacitance method (Corneometer CM 825) and conductance method (Skicon-200) for the evaluation of the hydration state of the skin. *Skin Research and Technology*, 3, 107-113.

- BAREL, A. O. & CLARYS, P. 2006. Measurement of epidermal capacitance. *In:* SERUP,J., JEMEC, G. B. E. & GROVE, G. L. (eds.) *Handbook of non-invasive methods and the skin.* 2 ed. Boca Raton: CRC Press.
- BAREL, A. O., COURAGE, W. & CLARYS, P. 2006. Suction champer method for measurement of skin mechanics: the new digital version of the Cutometer. *In:* SERUP, J., JEMEC, G. B. E. & GROVE, G. L. (eds.) *Handbook of non-invasive methods and the skin.* 2 ed. Boca Raton: CRC Press.
- BAREL, A. O., HENAU, K. & CLARYS, P. 2005. In vitro calibration and validation of the Reviscometer using silicone polymers as simple skin model systems. *Skin Research and Technology, Abstract. P022*, 11, 294.
- BARNES, E. & BROCKLESBY, S. 2011. The use of Cica-Care in podiatry: a review of nine cases. *Podiatry Now*, 10, 18-22.
- BARONI, A., BUOMMINO, E., DE GREGORIO, V., RUOCCO, E., RUOCCO, V. & WOLF, R. 2012. Structure and function of the epidermis related to barrier properties. *Clinics In Dermatology*, 30, 257-262.
- BARROW, J., HUGHES, J., CLARK, P. & KLENERMAN, L. 1992. A study of the effect of wear on the pressure-relieving properties of foot orthosis. *The Foot*, 1, 195-199.
- BASSETT, D. R., WYATT, H. R., THOMPSON, H., PETERS, J. C. & HILL, J. O. 2010. Pedometer-Measured Physical Activity and Health Behaviors in United States Adults. *Medicine and science in sports and exercise*, 42, 1819-1825.
- BENVENUTI, F., FERRUCCI, L., GURALNIK, J. M., GANGEMI, S. & BARONI, A. 1995. FOOT PAIN AND DISABILITY IN OLDER PERSONS - AN EPIDEMIOLOGIC SURVEY. Journal of the American Geriatrics Society, 43, 479-484.
- BERARDESCA, E., CAMELI, N., PRIMAVERA, G. & CARRERA, M. 2006. Clinical and instrumental evaluation of skin improvement after treatment with a new 50% pyruvic acid peel. *Dermatologic Surgery*, 32, 526-31.
- BESSOUD, B., LASSAU, N., KOSCIELNY, S., LONGVERT, C., AVRIL, M.-F., DUVILLARD, P., ROUFFIAC, V., LECLÈRE, J. & ROCHE, A. 2003. Highfrequency sonography and color Doppler in the management of pigmented skin lesions. Ultrasound in Medicine & Biology, 29, 875-879.
- BEVANS, J. S. & BOWKER, P. 1999. Foot structure and function: aetiological risk factors for callus formation in diabetic and non-diabetic subjects. *The Foot*, 9, 120-127.

- BIRKE, J. A., FOTO, J. G. & PFIEFER, L. A. 1999. Effect of orthosis material hardness on walking pressure in high-risk diabetes patients. *Journal of Prosthetics and Orthotics*, 11, 43.
- BISIAUX, M. & MORETTO, P. 2008. The effects of fatigue on plantar pressure distribution in walking. *Gait Posture*, 28, 693-8.
- BLACK, J. R. & HALE, W. E. 1987. Prevalence of foot complaints in the elderly. *Journal Of The American Podiatric Medical Association*, 77, 308-311.
- BOBADILLA, F., WORTSMAN, X., MUNOZ, C., SEGOVIA, L., ESPINOZA, M. & JEMEC, G. B. 2008. Pre-surgical high resolution ultrasound of facial basal cell carcinoma: correlation with histology. *Cancer Imaging*, 8, 163-72.
- BONAPARTE, J. P. & CHUNG, J. 2014. The effect of probe placement on inter-trial variability when using the Cutometer MPA 580. *Journal of Medical Engineering and Technology*, 38, 85-9.
- BONAPARTE, J. P., ELLIS, D. & CHUNG, J. 2013. The effect of probe to skin contact force on Cutometer MPA 580 measurements. *Journal of Medical Engineering & Technology*, 37, 208-212.
- BONTRAGER, E. L., BOYD, L. A., HEINO, J. G., MULROY, S. J. & PERRY, J. 1997. Determination of Novel Pedar masks using Harris Mat imprints. *Gait Posture*, 5, 167-168.
- BOULTON, A. J. M., HARDISTY, C. H., BETTS, R. P., FRANKS, C. I., WORTH, R. C., WARD, J. D. & DUCKWORTH, T. 1983. Dynamic foot pressure and other studies as diagnostic and management aids in diabetic neuropathy. *Diabetes Care*, 6, 26-33.
- BOUWSTRA, J. A., DE GRAAFF, A., GOORIS, G. S., NIJSSE, J., WIECHERS, J. W. & VAN AELST, A. C. 2003. Water Distribution and Related Morphology in Human Stratum Corneum at Different Hydration Levels. *Journal of Investigative Dermatology*, 120, 750-758.
- BRANTHWAITE, H., CHOCKALINGAM, N. & GREENHALGH, A. 2013. The effect of shoe toe box shape and volume on forefoot interdigital and plantar pressures in healthy females. *Journal of Foot and Ankle Research*, 6, 28.
- BRAUN, R. P., RABINOVITZ, H., TZU, J. E. & MARGHOOB, A. A. 2009. Dermoscopy research--an update. *Seminars in Cutaneous Medicine and Surgery*, 28, 165-71.
- BRIGGAMAN, R. A. 1982. Biochemical composition of the epidermal-dermal junction and other basement membrane. *Journal of Investigative Dermatology*, 78, 1-6.

- BRIGGAMAN, R. A. & WHEELER, C. E. 1975. The epidermal-dermal junction. *Journal* of Investigative Dermatology, 65, 71-85.
- BRODTKORB, T. H., KOGLER, G. F. & ARNDT, A. 2008. The influence of metatarsal support height and longitudinal axis position on plantar foot loading. *Clinical Biomechanics*, 23, 640-7.
- BROPHY, D. & LOBITZ, W. C. 1959. Injury and reinjury to the human epidermis. II. Epidermal basal cell response. *Journal of Investigative Dermatology*, 32, 495-503.
- BRYANT, A., TINLEY, P. & SINGER, K. 1999. Plantar pressure distribution in normal, hallux valgus and hallux limitus feet. *The Foot*, 9, 115-119.
- BURNFIELD, J. M., FEW, C. D., MOHAMED, O. S. & PERRY, J. 2004. The influence of walking speed and footwear on plantar pressures in older adults. *Clinical Biomechanics*, 19, 78-84.
- BURNFIELD, J. M., JORDE, A. G., AUGUSTIN, T. R., AUGUSTIN, T. A. & BASHFORD, G. R. 2007. Variations in plantar pressure variables across five cardiovascular exercises. *Medicine & Science in Sports & Exercise*, 39, 2012-20.
- BURNS, J. & BEGG, L. 2011. Optimizing the offloading properties of the total contact cast for plantar foot ulceration. *Diabetic Medicine*, 28, 179-85.
- BURNS, J., CROSBIE, J., HUNT, A. & OUVRIER, R. 2005. The effect of pes cavus on foot pain and plantar pressure. *Clinical Biomechanics*, 20, 877-82.
- BURNS, J., WEGENER, C., BEGG, L., VICARETTI, M. & FLETCHER, J. 2009. Randomized trial of custom orthoses and footwear on foot pain and plantar pressure in diabetic peripheral arterial disease. *Diabetic Medicine*, 26, 893-9.
- BUS, S. A., MAAS, M., DE LANGE, A., MICHELS, R. P. & LEVI, M. 2005. Elevated plantar pressures in neuropathic diabetic patients with claw/hammer toe deformity. *Journal of Biomechanics*, 38, 1918-25.
- BUS, S. A., MASSA, M., CAVANAGH, P. R., MICHELS, R. P. J. & LEVI, M. 2004a. Plantar fat-pad displacement in neuropathic diabetic patients with toe deformity. *Diabetes Care*, 27, 2376-2381.
- BUS, S. A., ULBRECHT, J. S. & CAVANAGH, P. R. 2004b. Pressure relief and load redistribution by custom-made insoles in diabetic patients with neuropathy and foot deformity. *Clinical Biomechanics*, 19, 629-38.
- BUS, S. A. & WAAIJMAN, R. 2013. The value of reporting pressure-time integral data in addition to peak pressure data in studies on the diabetic foot: A systematic review. *Clinical Biomechanics*, 28, 117-121.

- BYGRAVE, C. J. & BETTS, R. P. 1992. The plantar tissue thickness in the foot: a new ultrasound technique for loadbearing measurements and a metatarsal head depth study. *The Foot*, 2, 71-78.
- CARL, H.-D., PAUSER, J., SWOBODA, B., JENDRISSEK, A. & BREM, M. 2014. Soccer boots elevate plantar pressures in elite male soccer professionals. *Clinical Journal of Sports Medicine*, 24, 58-61.
- CATALA-PETAVY, C., MACHET, L., GEORGESCO, G., PETAVY, F., MARUANI, A. & VAILLANT, L. 2009. Contribution of skin biometrology to the diagnosis of the Ehlers-Danlos syndrome in a prospective series of 41 patients. *Skin Research & Technology*, 15, 412-7.
- CAVANAGH, P. R. 1999. Plantar soft tissue thickness during ground contact in walking. *Journal of Biomechanics*, 32, 623-628.
- CHANG, B. C., WANG, J. Y., HUANG, B. S., LIN, H. Y. & LEE, W. C. 2012. Dynamic impression insole in rheumatoid foot with metatarsal pain. *Clinical Biomechanics*, 27, 196-201.
- CHAO, C. Y., ZHENG, Y. P. & CHEING, G. L. 2011. Epidermal thickness and biomechanical properties of plantar tissues in diabetic foot. *Ultrasound in Medicine* & *Biology*, 37, 1029-38.
- CHAPMAN, J. D. 2014. Improving the design of the curved rocker shoe for people with diabetes. Ph. D., University of Salford.
- CHAPMAN, J. D., PREECE, S., BRAUNSTEIN, B., HOHNE, A., NESTER, C. J., BRUEGGEMANN, P. & HUTCHINS, S. 2013. Effect of rocker shoe design features on forefoot plantar pressures in people with and without diabetes. *Clinical Biomechanics*, 28, 679-85.
- CHEN, C., LYONS-GIORDANO, B., LAZARUS, G. S. & JENSEN, P. J. 1993. Differential expression of plasminogen activators and their inhibitors in an organotypic skin coculture system. *Journal of Cell Science*, 106, 45-53.
- CHEN, W.-P., JU, C.-W. & TANG, F.-T. 2003. Effects of total contact insoles on the plantar stress redistribution: a finite element analysis. *Clinical Biomechanics*, 18, S17-S24.
- CHILCOTT, R. P., DALTON, C. H., EMMANUEL, A. J., ALLEN, C. E. & BRADLEY, S. T. 2002. Transepidermal Water Loss Does Not Correlate with Skin Barrier Function In Vitro. *Journal of Investigative Dermatology*, 118, 871-875.

- CHOI, J. W., KWON, S. H., HUH, C. H., PARK, K. C. & YOUN, S. W. 2012. The influences of skin visco-elasticity, hydration level and aging on the formation of wrinkles: a comprehensive and objective approach. *Skin Research & Technology*.
- CLARYS, P., BAREL, A. O. & GABARD, B. 1999. Non-invasive electrical measurements for the evaluation of the hydration state of the skin: comparison between three conventional instruments - the Cornoemeter, the Skicon and the Nova DPM. *Skin Research and Technology*, 5, 14-20.
- CLARYS, P., CLIJSEN, R. & BAREL, A. O. 2011. Influence of probe application pressure on in vitro and in vivo capacitance (Corneometer CM 825((R))) and conductance (Skicon 200 EX((R))) measurements. *Skin Research & Technology*, 17, 445-50.
- COLAGIURI, S., MARSDEN, L. L., NAIDU, V. & TAYLOR, L. 1995. The use of orthotic devices to correct plantar callus in people with diabetes. *Diabetic Research and Clinical Practice*, 28, 29-34.
- CONG, Y., CHEUNG, J. T., LEUNG, A. K. & ZHANG, M. 2011. Effect of heel height on in-shoe localized triaxial stresses. *Journal of Biomechanics*, 44, 2267-72.
- CORCUFF, P., CHATENAY, F. & LEVEQUE, J. L. 1984. A fully automated system to study skin surface patterns. *International Journal Of Cosmetic Science*, 6, 167.
- CORCUFF, P., DE LACHARRIERE, O. & LEVEQUE, J. L. 1991. Extension induced changes in the microrelief of the human volar forearm: variation with age. *Journal of Gerontology: Medical Sciences*, 46, 223-227.
- CORCUFF, P., DE RIGAL, J., MAKKI, S., LEVEQUE, J. L. & PORTE, G. 1983. Skin relief and aging. *Journal of the Society of Cosmetic Chemists*, 34, 177-189.
- CORCUFF, P., LEVEQUE, J. L., GROVE, G. L. & KLIGMAN, A. M. 1987. The impact of aging on the microrelief of periorbital and leg skin. *Journal of the Society of Cosmetic Chemists*, 82.
- COURAGE-KHAZAKA 2005. *The Reviscometer RVM 600*, Köln, Germany, Courage and Khazaka electronic.
- COURAGE-KHAZAKA 2009a. *Corneometer MPA 825*, Köln, Germany, Courage and Khazaka electronic.
- COURAGE-KHAZAKA 2009b. Information and operating instruction: Visioscan^R VC 98 and the software SELS (surface evaluation of the living skin). Köln, Germany, Courage and Khazaka electronic.

- COURAGE-KHAZAKA 2010a. Information and Operating Instructions for the Cutometer^R MPA 580 and the software Cutometer^R MPA Q, Köln, Germany, Courage and Khazaka electronic.
- COURAGE-KHAZAKA 2010b. Information and Operating Instructions: Visioline^R VL 650 Quantiride^R., Köln, Germany, Courage and Khazaka electronic.
- COURAGE-KHAZAKA 2012. Information and Operating Instructions for the Multi Probe Adapter MPA and its probes, Köln, Germany, Courage and Khazaka electronic.
- CRAVELLO, B. & FERRI, A. 2008. Relationships between skin properties and environmental parameters. *Skin Research & Technology*, 14, 180-6.
- CRONKWRIGHT, D. G., SPINK, M. J., LANDORF, K. B. & MENZ, H. B. 2011. Evaluation of the pressure-redistributing properties of prefabricated foot orthoses in older people after at least 12 months of wear. *Gait Posture*, 34, 553-7.
- D'AMBROGI, E., GIURATO, L., D'AGOSTINO, M. A., GIACOMOZZI, C., MACELLARI, V., CASELLI, A. & UCCIOLI, L. 2003. Contribution of plantar fascia to the increased forefoot pressures in diabetic patients. *Diabetes Care*, 26, 1525-1529.
- D'AOÛT, K., PATAKY, T. C., DE CLERCQ, D. & AERTS, P. 2009. The effects of habitual footwear use: foot shape and function in native barefoot walkers. *Footwear Science*, 1, 81-94.
- DANG, Y. Y., REN, Q. S., LIU, H. X., MA, J. B. & ZHANG, J. S. 2005. Comparison of histologic, biochemical, and mechanical properties of murine skin treated with the 1064-nm and 1320-nm Nd:YAG lasers. *Experimental Dermatology*, 14, 876-82.
- DAVIES, G., REILLY, J. J., MCGOWAN, A. J., DALL, P. M., GRANAT, M. H. & PATON, J. Y. 2012. Validity, Practical Utility, and Reliability of the activPAL[™] in Preschool Children. *Medicine & Science in Sports & Exercise*, 44, 761-768 10.1249/MSS.0b013e31823b1dc7.
- DAVYS, H. J., TURNER, D. E., HELLIWELL, P. S., CONAGHAN, P. G., EMERY, P. & WOODBURN, J. 2005. Debridement of plantar callosities in rheumatoid arthritis: a randomized controlled trial. *Rheumatology*, 44, 207-10.
- DE VET, H. C., TERWEE, C. B., KNOL, D. L. & BOUTER, L. M. 2006. When to use agreement versus reliability measures. *Journal of Clinical Epidemiology*, 59, 1033-9.

- DEMERJIAN, M., HACHEM, J. P., TSCHACHLER, E., DENECKER, G., DECLERCQ, W., VANDENABEELE, P., MAURO, T., HUPE, M., CRUMRINE, D., ROELANDT, T., HOUBEN, E., ELIAS, P. M. & FEINGOLD, K. R. 2008. Acute modulations in permeability barrier function regulate epidermal cornification: role of caspase-14 and the protease-activated receptor type 2. *American Journal of Pathology*, 172, 86-97.
- DESHAIES, A., ROY, P., SYMEONIDIS, P. D., LARUE, B., MURPHY, N. & ANCTIL, E. 2011. Metatarsal bars more effective than metatarsal pads in reducing impulse on the second metatarsal head. *The Foot*, 21, 172-5.
- DIMONTE, D. & LIGHT, H. 1983. Pathomechanics, gait deviations and treatment of the rheumatoid foot: a clinical report. *Physical Therapy*, 62, 1148-1156.
- DOBREV, H. 1998. Use of Cutometer to assess dermal oedema in erysipelas of the lower legs. *Skin Research & Technology*, 4, 155.
- DOBREV, H. 2000. In vivo Study of Skin Mechanical Properties in Psoriasis Vulgaris. Acta Dermato-Venereologica, 80, 263-266.
- DOBREV, H. 2005. Application of Cutometer area parameters for the study of human skin fatigue. *Skin Research & Technology*, 11, 120-122.
- DOBREV, H. 2007. In vivo study of skin mechanical properties in Raynaud's phenomenon. *Skin Research & Technology*, 13, 91-94.
- DRAAIJERS, L. J., BOTMAN, Y. A., TEMPELMAN, F. R., KREIS, R. W., MIDDELKOOP, E. & VAN ZUIJLEN, P. P. 2004. Skin elasticity meter or subjective evaluation in scars: a reliability assessment. *Burns*, 30, 109-14.
- DUFFIN, A. C., KIDD, R., CHAN, A. & DONAGHUE, K. C. 2003. High plantar pressure and callus in diabetic adolescents. Incidence and treatment. *Journal Of The American Podiatric Medical Association*, 93, 214-220.
- DUFFIN, A. C., LAM, A., KIDD, R., CHAN, A. K. F. & DONAGHUE, K. C. 2002. Ultrasonography of plantar soft tissues thickness in young people with diabetes. *Diabetic Medicine*, 19, 1009-1013.
- DUNN, J. E. 2004. Prevalence of Foot and Ankle Conditions in a Multiethnic Community Sample of Older Adults. *American Journal of Epidemiology*, 159, 491-498.
- EBERLEIN-KÖNIG, B., SCHÄFER, T., HUSS-MARP, J., DARSOW, U., MÖHRENSCHLAGER, M., HERBERT, O., ABECK, D., KRÄMER, U., BEHRENDT, H. & RING, J. 2000. Skin Surface pH, Stratum Corneum Hydration, Trans-epidermal Water Loss and Skin Roughness Related to Atopic Eczema and

Skin Dryness in a Population of Primary School Children: Clinical Report. *Acta Dermato-Venereologica*, 80, 188-191.

- EBRAHIM, S. B., SAINSBURY, R. & WATSON, S. 1981. Foot problems of the elderly: a hospital survey. *British Medical Journal*, 283, 949-950.
- ELIAS, P. M., COOPER, E. R., KORC, A. & BROWN, B. E. 1981. Percutaneous transport in relation to stratum corneum structure and lipid composition. *Journal of Investigative Dermatology*, 76, 297-301.
- ELIAS, P. M. & FRIEND, D. S. 1975. The permeability barrier in mammalian epidermis. *Journal Of Cell Biology*, 65, 180-191.
- ELKEEB, R., HUI, X., CHAN, H., TIAN, L. & MAIBACH, H. I. 2010. Correlation of transepidermal water loss with skin barrier properties in vitro: comparison of three evaporimeters. *Skin Research & Technology*, 16, 9-15.
- ENGELKE, M., JENSEN, J. M., EKANAYAKE-MUDIYANSELAGE, S. & PROKSCH,
  E. 1997. Effects of xerosis and ageing on epidermal proliferation and differentiation. *British Journal of Dermatology*, 137, 219-225.
- FARNDON, L., BARNES, A., LITTLEWOOD, K., HARLE, J., BEECROFT, C., BURNSIDE, J., WHEELER, T., MORRIS, S. & WALTERS, S. J. 2009. Clinical audit of core podiatry treatment in the NHS. *Journal of Foot and Ankle Research*, 2, 7.
- FERNANDEZ-SEGUIN, L. M., DIAZ MANCHA, J. A., SANCHEZ RODRIGUEZ, R., ESCAMILLA MARTINEZ, E., GOMEZ MARTIN, B. & RAMOS ORTEGA, J. 2014. Comparison of plantar pressures and contact area between normal and cavus foot. *Gait Posture*, 39, 789-92.
- FERNANDO, D. J. S., MASSON, E. A., VEVES, A. & BOULTON, A. J. M. 1991. Relationship of limited joint mobility to abnormal foot pressures and diabetic foot ulceration. *Diabetes Care*, 14, 8-11.
- FERNANDO, M., CROWTHER, R., LAZZARINI, P., SANGLA, K., CUNNINGHAM, M., BUTTNER, P. & GOLLEDGE, J. 2013. Biomechanical characteristics of peripheral diabetic neuropathy: A systematic review and meta-analysis of findings from the gait cycle, muscle activity and dynamic barefoot plantar pressure. *Clinical Biomechanics*, 28, 831-45.

FIELD, A. 2009. Discovering statistics using SPSS, London, SAGE Publications Ltd.

FIROOZ, A., SADR, B., BABAKOOHI, S., SARRAF-YAZDY, M., FANIAN, F., KAZEROUNI-TIMSAR, A., NASSIRI-KASHANI, M., NAGHIZADEH, M. M. &

DOWLATI, Y. 2012. Variation of Biophysical Parameters of the Skin with Age, Gender, and Body Region. *The Scientific World Journal*, 2012, 386936.

- FIRTH, J., NELSON, E. A., BRIGGS, M. & GORECKI, C. 2011. A qualitative study to explore the impact of foot ulceration on health-related quality of life in patients with rheumatoid arthritis. *International Journal of Nursing Studies*, 48, 1401-8.
- FLUHR, J. W., GLOOR, M., LAZZERINI, S., KLEESZ, P., GRIESHABER, R. & BERARDESCA, E. 1999a. Comparative study of five instruments measuring stratum corneum hydration (Corneometer CM 820 and CM 825, Skicon 200, Nova DPM 9003, DermaLab). Part I. In vitro. *Skin Research & Technology*, 5, 161.
- FLUHR, J. W., GLOOR, M., LAZZERINI, S., KLEESZ, P., GRIESHABER, R. & BERARDESCA, E. 1999b. Comparative study of five instruments measuring stratum corneum hydration (Corneometer CM 820 and CM 825, Skicon 200, Nova DPM 9003, DermaLab). Part II. In vivo. *Skin Research & Technology*, 5, 171.
- FONG, S. S. L., HUNG, L. K. & CHENG, J. C. Y. 1997. The cutometer and ultrasonography in the assessment of postburn hypertrophic scar — a preliminary study. *Burns*, 23, Supplement 1, S12-S18.
- FORNAGE, B. D. 1995. High-frequency sonography of the skin. European Journal of Ultrasound, 2, 173-182.
- FORNAGE, B. D., MCGAVRAN, M. H., DUVIC, M. & WALDRON, C. A. 1993. Imaging of the skin with 20-mhz US. *Radiology*, 189, 69-76.
- FOURCHET, F., KELLY, L., HOROBEANU, C., LOEPELT, H., TAIAR, R. & MILLET, G. P. 2012. Comparison of plantar pressure distribution in adolescent runners at low vs. high running velocity. *Gait Posture*, 35, 685-7.
- FREEMAN, D. B. 2002. Corns and calluses resulting from mechanical hyperkeratosis. *American Family Physician*, 65, 2277-2280.
- FRYKBERG, R. G., LAVERY, L. A., PHAM, H., HARVEY, C., HARKLESS, L. & VEVES, A. 1998. Role of neuropathy and high foot pressures in diabetic foot ulceration. *Diabetes Care*, 21, 1714-1719.
- GALICA, A. M., HAGEDORN, T. J., DUFOUR, A. B., RISKOWSKI, J. L., HILLSTROM, H. J., CASEY, V. A. & HANNAN, M. T. 2013. Hallux valgus and plantar pressure loading: The Framingham foot study. *Journal of Foot and Ankle Research*, 6, 42-49.
- GAMBICHLER, T., BOMS, S., STUCKER, M., KREUTER, A., MOUSSA, G., SAND, M., ALTMEYER, P. & HOFFMANN, K. 2006. Epidermal thickness assessed by

optical coherence tomography and routine histology: preliminary results of method comparison. *Journal of the European Academy of Dermatology and Venereology*, 20, 791-5.

- GAMBICHLER, T., MOUSSA, G., REGENITER, P., KASSECK, C., HOFMANN, M. R., BECHARA, F. G., SAND, M., ALTMEYER, P. & HOFFMANN, K. 2007. Validation of optical coherence tomography in vivo using cryostat histology. *Physics in Medicine and Biology*, 52, N75-85.
- GAMBICHLER, T., MOUSSA, G., SAND, M., SAND, D., ALTMEYER, P. & HOFFMANN, K. 2005. Applications of optical coherence tomography in dermatology. *Journal of Dermatological Science*, 40, 85-94.
- GARRIGUE, E., MARTINI, J., COUSTY-PECH, F., ROUQUIER, A. & DEGOUY, A. 2011. Evaluation of the moisturizer Pedimed((R)) in the foot care of diabetic patients. *Diabetes & Metabolism*, 37, 330-5.
- GIRARD, P., BERAUD, A. & SIRVENT, A. 2000. Study of three complementary techniques for measuring cutaneous hydration in vivo in human subjects: NMR spectroscopy, transient thermal transfer and corneometry - application to xerotic skin and cosmetics. *Skin Research & Technology*, 6, 205-213.
- GODFREY, A., CULHANE, K. M. & LYONS, G. M. 2007. Comparison of the performance of the activPAL[™] Professional physical activity logger to a discrete accelerometer-based activity monitor. *Medical Engineering & Physics*, 29, 930-934.
- GOLDBLUM, R. W. & PIPER, W. M. 1954. Artificial litchenification produced by scratching machine. *Journal of Investigative Dermatology*, 22, 405-415.
- GOLDSTEIN, B. & SANDERS, J. 1998. Skin response to repetitive mechanical stress: A new experimental model in pig. Archives of Physical Medicine and Rehabilitation, 79, 265-272.
- GORTER, K. J., KUYVENHOVEN, M. M. & DE MELKER, R. A. 2000. Nontraumatic foot complaints in older people. A population-based survey of risk factors, mobility, and well-being. *Journal Of The American Podiatric Medical Association*, 90, 397-402.
- GRANT, P. M., RYAN, C. G., TIGBE, W. W. & GRANAT, M. H. 2006. The validation of a novel activity monitor in the measurement of posture and motion during everyday activities. *British Journal of Sports Medicine*, 40, 992-997.

- GRANT, W. P., SULLIVAN, R., SONENSHINE, D. E., ADAM, M., SLUSSER, J. H., CARSON, K. A. & VINIK, A. I. 1997. Electron microscopic investigation of the effects of diabetes mellitus on the Achilles tendon. *Journal of Foot and Ankle Surgery*, 36, 272-278.
- GRAVANTE, G., POMARA, F., RUSSO, G., AMATO, G., CAPPELLO, F. & RIDOLA, C. 2005. Plantar pressure distribution analysis in normal weight young women and men with normal and claw feet: a cross-sectional study. *Clinical Anatomy*, 18, 245-50.
- GROL, R. & GRIMSHAW, J. 2003. From best evidence to best practice: effective implementation of change in patients' care. *The Lancet*, 362, 1225-1230.
- GROUIOS, G. 2004. Corns and calluses in athletes feet: a cause for concern. *The Foot*, 14, 175-184.
- GULDEMOND, N. A., LEFFERS, P., SCHAPER, N. C., SANDERS, A. P., NIEMAN, F., WILLEMS, P. & WALENKAMP, G. H. 2007. The effects of insole configurations on forefoot plantar pressure and walking convenience in diabetic patients with neuropathic feet. *Clinical Biomechanics*, 22, 81-7.
- HAFER, J. F., LENHOFF, M. W., SONG, J., JORDAN, J. M., HANNAN, M. T. & HILLSTROM, H. J. 2013. Reliability of plantar pressure platforms. *Gait Posture*, 38, 544-548.
- HARDING, C. R., LONG, S., RICHARDSON, J., ROGERS, J., ZHANG, Z., BUSH, A.
  & RAWLINGS, A. V. 2003. The cornified cell envelope: an important marker of stratum corneum maturation in healthy and dry skin. *International Journal Of Cosmetic Science*, 15, 157-167.
- HASHMI, F. & MALONE-LEE, J. 2007. Measurement of skin elasticity on the foot. *Skin Research & Technology*, 13, 252-8.
- HASHMI, F., MALONE-LEE, J. & HOUNSELL, E. 2006. Plantar skin in type II diabetes: an investigation of protein glycation and biomechanical properties of plantar epidermis. *European Journal of Dermatology*, 16, 23-32.
- HASHMI, F., NESTER, C., WRIGHT, C., NEWTON, V. & LAM, S. 2015a. Characterising the biophysical properties of normal and hyperkeratotic foot skin. *Journal of Foot and Ankle Research*, 8, 35-44.
- HASHMI, F., RICHARDS, B. S., FORGHANY, S., HATTON, A. L. & NESTER, C. J. 2013. The formation of friction blisters on the foot: the development of a laboratory-based blister creation model. *Skin Research & Technology*, 19, e479-89.

- HASHMI, F., WRIGHT, C., NESTER, C. & LAM, S. 2015b. The reliability of noninvasive biophysical outcome measures for evaluating normal and hyperkeratotic foot skin. *Journal of Foot and Ankle Research*, **8**, 28-39.
- HASTINGS, M. K., MUELLER, M. J., PILGRAM, T. K., LOTT, D. J., COMMEAN, P.
  K. & JOHNSON, J. E. 2007. Effect of metatarsal pad placement on plantar pressure in people with diabetes mellitus and peripheral neuropathy. *Foot & Ankle International*, 28, 84-8.
- HEALY, A., DUNNING, D. N. & CHOCKALINGAM, N. 2012. Effect of insole material on lower limb kinematics and plantar pressures during treadmill walking. *Prosthetics and Orthotics International*, 36, 53-62.
- HEDA, G. D. & ROBERTS, L. K. 2008. Role of tissue-type plasminogen activator in salicylic acid-induced sloughing of human corn tissue. *Journal Of The American Podiatric Medical Association*, 98, 345-352.
- HELFAND, A. 2003. Assessing hyperkeratotic lesions in the older patient. *Clinics in Podiatric Medicine and Surgery*, 20, 443-452.
- HELFAND, A. E., COOKE, H. L., WALINSKY, M. D. & DEMP, P. H. 1998. Foot problems associated with older patients. A focused podogeriatric study. *Journal Of The American Podiatric Medical Association*, 88, 237-241.
- HERMANNS-LÊ, T., JONLET, F., SCHEEN, A. & PIÉRARD, G. E. 2001. Age- and body mass index-related changes in cutaneous shear wave velocity. *Experimental Gerontology*, 36, 363-372.
- HINZ, P., HENNINGSEN, A., MATTHES, G., JAGER, B., EKKERNKAMP, A. & ROSENBAUM, D. 2008. Analysis of pressure distribution below the metatarsals with different insoles in combat boots of the German Army for prevention of march fractures. *Gait Posture*, 27, 535-8.
- HIRAO, T., DENDA, M. & TAKAHASHI, M. 2001. Identification of immature cornified envelopes in the barrier-impaired epidermis by characterization of their hydrophobicity and antigenicities of the components. *Experimental Dermatology*, 10, 35-44.
- HO, I. J., HOU, Y. Y., YANG, C. H., WU, W. L., CHEN, S. K. & GUO, L. Y. 2010. Comparison of plantar pressure distribution between different speed and incline during treatmill jogging. *Journal of Sports Science and Medicine*, 9, 154-160.
- HODGE, M. C., BACH, T. M. & CARTER, G. M. 1999. Orthotic management of plantar pressure and pain in rheumatoid arthritis. *Clinical Biomechanics*, 14, 567-575.

- HOLM, E. A., WULF, H. C., THOMASSEN, L. & JEMEC, G. B. 2006. Instrumental assessment of atopic eczema: validation of transepidermal water loss, stratum corneum hydration, erythema, scaling, and edema. *Journal of the American Academy of Dermatology*, 55, 772-80.
- HOLMES, G. B. & TIMMERMAN, L. 1990. A quantitative assessment of the effect of metatarsal pads on plantar pressures. *Foot & Ankle*, 11, 141-145.
- HOLSTEIN, P., LOHMANN, M., BITSCH, M. & JØRGENSEN, B. 2004. Achilles tendon lengthening, the panacea for plantar forefoot ulceration? *Diabetes/Metabolism Research and Reviews*, 20, S37-S40.
- HOSEIN, R. & LORD, M. 2000. A study of in-shoe plantar shear in normals. *Clinical Biomechanics*, 15, 46-53.
- HOSTE, E., KEMPERMAN, P., DEVOS, M., DENECKER, G., KEZIC, S., YAU, N., GILBERT, B., LIPPENS, S., DE GROOTE, P., ROELANDT, R., VAN DAMME, P., GEVAERT, K., PRESLAND, R. B., TAKAHARA, H., PUPPELS, G., CASPERS, P., VANDENABEELE, P. & DECLERCQ, W. 2011. Caspase-14 is required for filaggrin degradation to natural moisturizing factors in the skin. *Journal of Investigative Dermatology*, 131, 2233-41.
- HSI, W.-L., KANG, J.-H. & LEE, X.-X. 2005. Optimum Position of Metatarsal Pad in Metatarsalgia for Pressure Relief. American Journal of Physical Medicine & Rehabilitation, 84, 514-520.
- HSU, C. C., TSAI, W. C., CHEN, C. P., SHAU, Y. W., WANG, C. L., CHEN, M. J. & CHANG, K. J. 2005. Effects of aging on the plantar soft tissue properties under the metatarsal heads at different impact velocities. *Ultrasound in Medicine & Biology*, 31, 1423-9.
- HUGHES, J., PRATT, L., LINGE, K., CLARK, P. & KLENERMAN, L. 1991. Reliability of pressure measurements: the EM ED F system. *Clinical Biomechanics*, 6, 14-18.
- IBRAHIM, M., EL HILALY, R., TAHER, M. & MORSY, A. 2013. A pilot study to assess the effectiveness of orthotic insoles on the reduction of plantar soft tissue strain. *Clinical Biomechanics*, 28, 68-72.
- IDY-PERETTI, I., BITTOUN, J., ALLIOT, F. A., RICHARD, S. B., QUERLEUX, B. G. & CLUZAN, R. W. 1998. Lymphedematous skin and subcutis: In vivo high resolution magnetic resonance imaging evaluation. *Journal of Investigative Dermatology*, 110, 782-787.

- IMOKAWA, G., KUNO, H. & KAWAI, M. 1991. Stratum corneum lipids serve as a bound-water modulator. *Journal of Investigative Dermatology*, 96, 845-851.
- JACKSON, L., BINNING, J. & POTTER, J. 2004. Plantar pressures in rheumatoid arthritis using prefabricated metatarsal padding. *Journal Of The American Podiatric Medical Association*, 94, 239-245.
- JASAITIENE, D., VALIUKEVICIENE, S., LINKEVICIUTE, G., RAISUTIS, R., JASIUNIENE, E. & KAZYS, R. 2011. Principles of high-frequency ultrasonography for investigation of skin pathology. *Journal of the European Academy of Dermatology and Venereology*, 25, 375-82.
- JEMEC, G. B., SELVAAG, E., AGREN, M. & WULF, H. C. 2001. Measurement of the mechanical properties of skin with ballistometer and suction cup. *Skin Research & Technology*, 7, 122-126.
- JENSEN, J. M., SCHÜTZE, S., NEUMANN, C. & PROKSCH, E. 2000. Impaired cutaneous permeability barrier function, skin hydration, and sphingomyelinase activity in keratin 10 deficient mice. *Journal of Investigative Dermatology*, 115, 708-713.
- JEONG, J. H., FAN, Y., YOU, G. Y., CHOI, T. H. & KIM, S. 2015. Improvement of photoaged skin wrinkles with cultured human fibroblasts and adipose-derived stem cells: A comparative study. *Journal of Plastic, Reconstructive & Aesthetic Surgery*, 68, 372-381.
- JEONG, S. K., KIM, H. J., YOUM, J. K., AHN, S. K., CHOI, E. H., SOHN, M. H., KIM, K. E., HONG, J. H., SHIN, D. M. & LEE, S. H. 2008. Mite and cockroach allergens activate protease-activated receptor 2 and delay epidermal permeability barrier recovery. *Journal of Investigative Dermatology*, 128, 1930-9.
- JOSSE, G., GEORGE, J. & BLACK, D. 2011. Automatic measurement of epidermal thickness from optical coherence tomography images using a new algorithm. *Skin Research & Technology*, 17, 319-319.
- JOSSE, G., ROUVRAIS, C., MAS, A., HAFTEK, M., DELALLEAU, A., FERRAQ, Y., OSSANT, F., GEORGE, J., LAGARDE, J. M. & SCHMITT, A. M. 2009. A multitechnique evaluation of topical corticosteroid treatment. *Skin Research & Technology*, 15, 35-9.
- KANG, J. H., CHEN, M. D., CHEN, S. C. & HSI, W. L. 2006. Correlations between subjective treatment responses and plantar pressure parameters of metatarsal pad

treatment in metatarsalgia patients: a prospective study. BMC Musculoskeletal Disorders, 7, 95.

- KHAN, M. T., POTTER, M. & BIRCH, I. 1996. Podiatric treatment of hyperkeratotic plantar lesions with marigold tagetes erecta. *Phytotherapy Research*, 10, 211-214.
- KI, S. W., LEUNG, A. K. & LI, A. N. 2008. Comparison of plantar pressure distribution patterns between foot orthoses provided by the CAD-CAM and foam impression methods. *Prosthetics and Orthotics International*, 32, 356-62.
- KIM, S. D., HUH, C. H., SEO, K. I., SUH, D. H. & YOUN, J. I. 2002. Evaluation of skin surface hydration in Korean psoriasis patients: a possible factor influencing psoriasis. *Clinical & Experimental Dermatology*, 27, 147-152.
- KIM, S. H., KIM, S., CHOI, H. I., CHOI, Y. J., LEE, Y. S., SOHN, K. C., LEE, Y., KIM, C. D., YOON, T. J., LEE, J. H. & LEE, Y. H. 2010. Callus formation is associated with hyperproliferation and incomplete differentiation of keratinocytes, and increased expression of adhesion molecules. *British Journal of Dermatology*, 163, 495-501.
- KIRTLEY, C. 2014. *Ground reaction forces diagram* [Online]. Available: http://www.clinicalgaitanalysis.com/teach-in/friction.html [Accessed 28/02/2014.
- KLAESNER, J. W., HASTINGS, M. K., ZOU, D., LEWIS, C. & MUELLER, M. J. 2002. Plantar tissue stiffness in patients with diabetes mellitus and peripheral neuropathy. *Archives of Physical Medicine and Rehabilitation*, 83, 1796-801.
- KLEINERMAN, R., WHANG, T. B., BARD, R. L. & MARMUR, E. S. 2012. Ultrasound in dermatology: Principles and applications. *Journal of the American Academy of Dermatology*, 67, 478-87.
- KNAPP, A. C., FRANKE, W. W., HEID, H., HATZFELD, M., JORCANO, J. L. & MOLL, R. 1986. Cytokeratin No. 9, an epidermal type I keratin characteristic of a special program of keratinocyte differentiation displaying body site specificity. *Journal Of Cell Biology*, 103, 657-667.
- KO, P. H., HSIAO, T. Y., KANG, J. H., WANG, T. G., SHAU, Y. W. & WANG, C. L. 2009. Relationship between plantar pressure and soft tissue strain under metatarsal heads with different heel heights. *Foot & Ankle International*, 30, 1111-6.
- KOEHLER, M. J., PRELLER, A., KINDLER, N., ELSNER, P., KONIG, K., BUCKLE,
  R. & KAATZ, M. 2009. Intrinsic, solar and sunbed-induced skin aging measured in vivo by multiphoton laser tomography and biophysical methods. *Skin Research* & *Technology*, 15, 357-63.

- KOTTNER, J., SCHARIO, M., GARCIA BARTELS, N., PANTCHECHNIKOVA, E., HILLMANN, K. & BLUME-PEYTAVI, U. 2013. Comparison of two in vivo measurements for skin surface topography. *Skin Research & Technology*, 19, 84-90.
- KREMKAU, F. W. 2006. *Diagnostic ultrasound: Principles and instruments*, St Louis, Elsevier.
- KRUEGER, N., LUEBBERDING, S., OLTMER, M., STREKER, M. & KERSCHER, M. 2011. Age-related changes in skin mechanical properties: a quantitative evaluation of 120 female subjects. *Skin Research and Technology*, 17, 141-148.
- LANDIS, R. J. & KOCH, G. G. 1977. The measurement of observer agreement for categorical data. *Biometrics*, 33, 159-174.
- LANDORF, K. B., MORROW, A., SPINK, M. J., NASH, C. L., NOVAK, A., POTTER, J. & MENZ, H. B. 2013. Effectiveness of scalpel debridement for painful plantar calluses in older people: a randomized trial. *Trials*, 14, 243-251.
- LANE, T. J., LANDORF, K. B., BONANNO, D. R., RASPOVIC, A. & MENZ, H. B. 2014. Effects of shoe sole hardness on plantar pressure and comfort in older people with forefoot pain. *Gait Posture*, 39, 247-51.
- LANG, L. M. G., SIMMONITE, N., WEST, S. G. & DAY, S. 1994. Salicylic acid in the treatment of corns. *The Foot*, 4, 145-150.
- LASSAU, N., LAMURAGLIA, M., KOSCIELNY, S., SPATZ, A., ROCHE, A., LECLERE, J. & AVRIL, M. F. 2006. Prognostic value of angiogenesis evaluated with high-frequency and colour Doppler sonography for preoperative assessment of primary cutaneous melanomas: correlation with recurrence after a 5 year follow-up period. *Cancer Imaging*, 6, 24-9.
- LASSAU, N., MERCIER, S., KOSCIELNY, S., AVRIL, M. F., MARGULIS, A., MAMELLE, G., DUVILLARD, P. & LECLERE, J. 1999. Prognostic value of high-frequency sonography and colour doppler sonography for the preoperative assessment of melanomas. *American Journal of Roentgenology*, 172, 457-461.
- LAVERY, L. A., ARMSTRONG, D. G. & BOULTON, A. J. M. 2002. Ankle Equinus Deformity and Its Relationship to High Plantar Pressure in a Large Population with Diabetes Mellitus. *Journal Of The American Podiatric Medical Association*, 92, 479-482.
- LEBER, C. & EVANSKI, P. M. 1986. A comparison of shoe insole materials in plantar pressure relief. *Prosthetics and Orthotics International*, 10, 135-138.

- LEE, P. Y., LANDORF, K. B., BONANNO, D. R. & MENZ, H. B. 2014. Comparison of the pressure-relieving properties of various types of forefoot pads in older people with forefoot pain. *Journal of Foot and Ankle Research*, 7, 18.
- LEVIN, J. & MAIBACH, H. 2005. The correlation between transepidermal water loss and percutaneous absorption: an overview. *Journal of Controlled Release*, 103, 291-9.
- LIN, T.-L., SHEEN, H.-M., CHUNG, C.-T., YANG, S.-W., LIN, S.-Y., LUO, H.-J., CHEN, C.-Y., CHAN, I.-C., SHIH, H.-S. & SHEU, W. H.-H. 2013. The effect of removing plugs and adding arch support to foam based insoles on plantar pressures in people with diabetic peripheral neuropathy. *Journal of Foot and Ankle Research*, 6, 29.
- LOCKE, M., PERRY, J., CAMPBELL, J. & THOMAS, L. 1986. Ankle and subtalar motion during gait in arthritic patients. *Physical Therapy*, 64, 504-509.
- LORD, M. & HOSEIN, R. 2000. A study of in-shoe plantar shear in patients with diabetic neuropathy. *Clinical Biomechanics*, 15, 278-283.
- LOTT, D. J., HASTINGS, M. K., COMMEAN, P. K., SMITH, K. E. & MUELLER, M. J. 2007. Effect of footwear and orthotic devices on stress reduction and soft tissue strain of the neuropathic foot. *Clinical Biomechanics*, 22, 352-9.
- LUEBBERDING, S., KRUEGER, N. & KERSCHER, M. 2013. Skin physiology in men and women: in vivo evaluation of 300 people including TEWL, SC hydration, sebum content and skin surface pH. *International Journal Of Cosmetic Science*, 35, 477-483.
- LUEBBERDING, S., KRUEGER, N. & KERSCHER, M. 2014. Mechanical properties of human skin in vivo: a comparative evaluation in 300 men and women. *Skin Research & Technology*, 20, 127-35.
- LUND, A. & LUND, M. 2013. *Spearman's rank order correlation* [Online]. Laerd Statistics. Available: https://statistics.laerd.com/statistical-guides/spearmans-rank-order-correlation-statistical-guide.php [Accessed 29th January 2015].
- LYONS-GIORDANO, B. & LAZARUS, G. S. 1995. Skin abnormalities in mice transgenic for plasminogen activator inhibitor 1: implications for the regulation of desquamation and follicular neogenesis by plasminogen activator enzymes. *Developmental Biology*, 170, 289-298.
- MACHET, L., BELOT, V., NAOURI, M., BOKA, M., MOURTADA, Y., GIRAUDEAU,B., LAURE, B., PERRINAUD, A., MACHET, M. C. & VAILLANT, L. 2009.Preoperative measurement of thickness of cutaneous melanoma using high-

resolution 20 MHz ultrasound imaging: A monocenter prospective study and systematic review of the literature. *Ultrasound in Medicine & Biology*, 35, 1411-20.

- MACKENZIE, I. C. 1974a. The effects of frictional stimulation on mouse ear epidermis II. Histologic appearance and cell counts. *Journal of Investigative Dermatology*, 63, 194-198.
- MACKENZIE, I. C. 1974b. The effects of frictional stimulation on mouse ear epidermis. I. Cell proliferation. *Journal of Investigative Dermatology*, 62, 80-85.
- MACKIE, R. M. 2003. Clinical dermatology, Oxford University Press.
- MADISON, K. C. 2003. Barrier function of the skin: "La raison d'etre" of the epidermis. *Journal of Investigative Dermatology*, 121, 231-241.
- MAN, M. Q., XIN, S. J., SONG, S. P., CHO, S. Y., ZHANG, X. J., TU, C. X., FEINGOLD, K. R. & ELIAS, P. M. 2009. Variation of Skin Surface pH, Sebum Content and Stratum Corneum Hydration with Age and Gender in a Large Chinese Population. *Skin Pharmacology and Physiology*, 22, 190-199.
- MANDATO, M. G. & NESTER, E. 1999. The effects of increasing heel height on forefoot peak pressure. *Journal Of The American Podiatric Medical Association*, 89, 75-80.
- MANNY-AFRAMIAN, V. & DIKSTEIN, S. 1995. Indentometry. *Handbook of noninvasive methods and the skin*. USA: CRC Press.
- MARRAKCHI, S. & MAIBACH, H. I. 2007. Biophysical parameters of skin: map of human face, regional, and age-related differences. *Contact Dermatitis*, 57, 28-34.
- MARTINEZ-NOVA, A., SANCHEZ-RODRIGUEZ, R., PEREZ-SORIANO, P., LLANA-BELLOCH, S., LEAL-MURO, A. & PEDRERA-ZAMORANO, J. D. 2010.
   Plantar pressures determinants in mild Hallux Valgus. *Gait Posture*, 32, 425-7.
- MATSUMOTO, T., YUASA, H., KAI, R., UEDA, H., OGURA, S. & HONDA, Y. 2007. Skin capacitance in normal and atopic infants, and effects of moisturizers on atopic skin. *Journal of Dermatology*, 34, 447-50.
- MCGRATH, J. A. & UITTO, J. 2008. The filaggrin story: novel insights into skin-barrier function and disease. *Trends in Molecular Medicine*, 14, 20-7.
- MCPOIL, T. G., CORNWALL, M. W., DUPUIS, L. & CORNWELL, M. 1999. Variability of plantar pressure data. A comparison of the two-step and midgait methods. *Journal Of The American Podiatric Medical Association*, 89, 495-501.

- MENZ, H. B. & LORD, S. R. 2001. Foot pain impairs balance and functional ability in community-dwelling older people. *Journal Of The American Podiatric Medical Association*, 91, 222-229.
- MENZ, H. B., ZAMMIT, G. V. & MUNTEANU, S. E. 2007. Plantar pressures are higher under callused regions of the foot in older people. *Clinical & Experimental Dermatology*, 32, 375-80.
- MICKLE, K. J., MUNRO, B. J., LORD, S. R., MENZ, H. B. & STEELE, J. R. 2010. Foot pain, plantar pressures, and falls in older people: a prospective study. *Journal of the American Geriatrics Society*, 58, 1936-1940.
- MIRRASHED, F. & SHARP, J. C. 2004a. In vivo morphological characterisation of skin by MRI micro-imaging methods. *Skin Research & Technology*, 10, 149-160.
- MIRRASHED, F. & SHARP, J. C. 2004b. In vivo quantitative analysis of the effect of hydration (immersion and Vaseline treatment) in skin layers using high-resolution MRI and magnetic transfer contrast. *Skin Research & Technology*, 10, 14-22.
- MIYAZAKI, A., SAIDA, T., KOGA, H., OGUCHI, S., SUZUKI, T. & TSUCHIDA, T. 2005. Anatomical and histopathological correlates of the dermoscopic patterns seen in melanocytic nevi on the sole: a retrospective study. *Journal of the American Academy of Dermatology*, 53, 230-6.
- MORI, T., HAMATANI, M., NOGUCHI, H., OE, M. & SANADA, H. 2012. Insole-type simultaneous measurement system of plantar pressure and shear force during gait for diabetic patients. *Journal of Robotic and Mechatronics*, 24, 766-772.
- MUELLER, M. J., DIAMOND, J. E., A., D. & SINACORE, D. R. 1989. Insensitivity, limited joint mobility and plantar ulcers in patients with diabetes mellitus. *Physical Therapy*, 69, 453-459.
- MUELLER, M. J., HASTINGS, M., COMMEAN, P. K., SMITH, K. E., PILGRAM, T. K., ROBERTSON, D. & JOHNSON, J. 2003. Forefoot structural predictors of plantar pressures during walking in people with diabetes and peripheral neuropathy. *Journal of Biomechanics*, 36, 1009-1017.
- MUELLER, M. J., LOTT, D. J., HASTINGS, M. K., COMMEAN, P. K., SMITH, K. E. & PILGRAM, T. K. 2006. Efficacy and mechanism of orthotic devices to unload metatarsal heads in people with diabetes and a history of plantar ulcers. *Physical Therapy*, 86, 833-842.

- MURRAY, B. C. & WICKETT, R. R. 1997. Correlations between Dermal Torque Meter®, Cutometer®, and Dermal Phase Meter® measurements of human skin. *Skin Research and Technology*, 3, 101-106.
- MURRAY, H. J., YOUNG, M. J., HOLLIS, S. & BOULTON, A. J. M. 1996. The association between callus formation, high pressures and neuropathy in diabetic foot ulceration. *Diabetic Medicine*, 13, 979-982.
- NAGEL, A., FERNHOLZ, F., KIBELE, C. & ROSENBAUM, D. 2008. Long distance running increases plantar pressures beneath the metatarsal heads: a barefoot walking investigation of 200 marathon runners. *Gait Posture*, 27, 152-5.
- NAM, G. W., BAEK, J. H., KOH, J. S. & HWANG, J. K. 2015. The seasonal variation in skin hydration, sebum, scaliness, brightness and elasticity in Korean females. *Skin Research and Technology*, 21, 1-8.
- NETO, P., FERREIRA, M., BAHIA, F. & COSTA, P. 2013. Improvement of the methods for skin mechanical properties evaluation through correlation between different techniques and factor analysis. *Skin Research and Technology*, 19, 405-416.
- NGUYEN, D. Q. A., POTOKAR, T. S. & PRICE, P. 2010. An objective long-term evaluation of Integra (a dermal skin substitute) and split thickness skin grafts, in acute burns and reconstructive surgery. *Burns*, 36, 23-28.
- NISHIDE, K., NAGASE, T., OBA, M., OE, M., OHASHI, Y., IIZAKA, S., NAKAGAMI, G., KADOWAKI, T. & SANADA, H. 2009. Ultrasonographic and thermographic screening for latent inflammation in diabetic foot callus. *Diabetes Research and Clinical Practice*, 85, 304-9.
- NOVAK, P., BURGER, H., TOMSIC, M., MARINCEK, C. & VIDMAR, G. 2009. Influence of foot orthoses on plantar pressures, foot pain and walking ability of rheumatoid arthritis patients--a randomised controlled study. *Disability and Rehabilitation*, 31, 638-45.
- OHSHIMA, H., TADA, A., KANAMARU, A., AKAMATSU, H., SAKAI, Y., ITOH, M. & KANTO, H. 2011. Relevance of the directionality of skin elasticity to aging and sagging of the face. *Skin Research & Technology*, 17, 101-7.
- OTTER, S. J., BOWEN, C. J. & YOUNG, A. K. 2004. Forefoot plantar pressures in rheumatoid arthritis. *Journal Of The American Podiatric Medical Association*, 94, 255-260.

- OWINGS, T. M., WOERNER, J. L., FRAMPTON, J. D., CAVANAGH, P. R. & BOTEK, G. 2008. Custom therapeutic insoles based on both foot shape and plantar pressure measurement provide enhanced pressure relief. *Diabetes Care*, 31, 839-844.
- PALASTANGA, N. & SOAMES, R. W. 2012. Anatomy and Human Movement: Structure and function. Edinburgh: Churchill Livingstone Elsevier.
- PAPANAS, N., PAPAZOGLOU, D., PAPATHEODOROU, K. & MALTEZOS, E. 2011. Evaluation of a new foam to increase skin hydration of the foot in type 2 diabetes: a pilot study. *International Wound Journal*, 8, 297-300.
- PATAKY, Z., GOLAY, A., FARAVEL, L., DA SILVA, J., MAKOUNDOU, V., PETER-RIESCH, B. & ASSAL, J. P. 2002. The impact of callosities on the magnitude and duration of plantar pressure in patients with diabetes mellitus. A callus may cause 18,600 kilograms of excess plantar pressure per day. *Diabetes & Metabolism*, 28, 356-361.
- PAYE, M., MAC-MARY, S., ELKHYAT, A., TARRIT, C., MERMET, P. & HUMBERT, P. H. 2007. Use of the Reviscometer for measuring cosmetics-induced skin surface effects. *Skin Research & Technology*, 13, 343-9.
- PEDERSEN, L., HANSEN, B. & JEMEC, G. B. 2003. Mechanical properties of the skin: A comparison between two suction cup methods. *Skin Research & Technology*, 9, 111-115.
- PELLACANI, G. & SEIDENARI, S. 2003. Preoperative melanoma thickness determination by by 20-mhz sonography and digital videomicroscopy in combination. *Archives of Dermatology*, 139, 293-298.
- PENA FERREIRA, M. R., COSTA, P. C. & BAHIA, F. M. 2010. Efficacy of anti-wrinkle products in skin surface appearance: a comparative study using non-invasive methods. *Skin Research & Technology*, 16, 444-9.
- PERRY, J. E., HALL, J. O. & DAVIS, B. L. 2002. Simultaneous measurement of plantar pressure and shear forces in diabetic individuals. *Gait Posture*, 15, 101-107.
- PETIT, L., FOGOUANG, L., UHODA, I., SMITZ, S., PIERARD-FRANCHIMONT, C. & PIERARD, G. E. 2003. Regional variability in mottled subclinical melanoderma in the elderly. *Experimental Gerontology*, 38, 327-331.
- PIERARD-FRANCHIMONT, C., QUATRESOOZ, P. & PIERARD, G. E. 2011. Specular light reflectance of flakes in seborrhoeic dermatitis of the scalp: a pilot study. *Clinical & Experimental Dermatology*, 36, 793-6.

- PIERARD, G. E., PIERARD-FRANCHIMONT, C. & QUATRESOOZ, P. 2012. Field melanin mapping of the hairless scalp. *Skin Research & Technology*, 18, 431-5.
- PINKUS, H. 1952. Examination of the epidermis by the strip method. II. Biometric data on regeneration of the human epidermis. *Journal of Investigative Dermatology*, 19.
- PITEI, D. L., FOSTER, A. & EDMONDS, M. 1999. The effect of regular callus removal on foot pressures. *Journal of Foot & Ankle Surgery*, 38, 251-255.
- PLANK, M. J. 1995. The pattern of forefoot pressure distribution in hallux valgus. *The Foot*, 5, 8-14.
- POLLARD, J. P. & LE QUESNE, L. P. 1983. Method of healing diabetic forefoot ulcers. *British Medical Journal*, 286, 436-437.
- POLLARD, J. P., LE QUESNE, L. P. & TAPPIN, J. W. 1983. Forces under the foot. *Journal Of Biomedical Engineering*, 5, 37-40.
- POON, C. & LOVE, B. 1997. Efficacy of foot orthotics for metatarsalgia. 1997, 7, 202-204.
- POSTEMA, K., BURM, P. E. T., V. D. ZANDE, M. E. & V. LIMBEEK, J. 1998. Primary metatarsalgia: the influence of a custom moulded insole and a rockerbar on plantar pressure. *Prosthetics and Orthotics International*, 22, 35-44.
- POTTER, J. 2000. Regrowth patterns of plantar callus. *The Foot*, 10, 144-148.
- POTTER, J. & POTTER, M. 2000a. Regrowth patterns of plantar callus. *The Foot*, 10, 144-148.
- POTTER, J. & POTTER, M. J. 2000b. Effect of callus removal on peak plantar pressures. *The Foot*, 10, 23-26.
- POTTS, R. O. & FRANCOEUR, M. L. 1990. Lipid biophysics of water loss through the skin. *Proceedings of the National Academy of Sciences of the United States of America*, 87, 3871-3873.
- PRICE, D. D., FINNISS, D. G. & BENEDETTI, F. 2008. A comprehensive review of the placebo effect: recent advances and current thought. *Annual Review of Psychology*, 59, 565-90.
- PUTTI, A. B., ARNOLD, G. P., COCHRANE, L. A. & ABBOUD, R. J. 2008. Normal pressure values and repeatability of the Emed ST4 system. *Gait Posture*, 27, 501-5.
- PYUN, H.-B., KIM, M., PARK, J., SAKAI, Y., NUMATA, N., SHIN, J.-Y., SHIN, H.-J., KIM, D.-U. & HWANG, J.-K. 2012. Effects of Collagen Tripeptide Supplement on Photoaging and Epidermal Skin Barrier in UVB-exposed Hairless Mice. *Preventive Nutrition and Food Science*, 17, 245-253.

- QAA 2008. The framework for higher education qualifications in England, Wales and Northern Ireland, Mansfield, UK, The Quality Assurance Agency for Higher Education.
- QUATRESOOZ, P., HENRY, F., PAQUET, P. & PIERARD, G. E. 2011. Photoaging under recreational sunbeds. *Skin Research & Technology*, 17, 309-313.
- QUATRESOOZ, P., HERMANNS, J. F., PAQUET, P. & PIÉRARD, G. E. 2006. Mechanobiology and force transduction in scars developed in darker skin types. *Skin Research & Technology*, 12, 279-282.
- QUEEN, R. M., ABBEY, A. N., WIEGERINCK, J. I., YODER, J. C. & NUNLEY, J. A. 2010. Effect of shoe type on plantar pressure: a gender comparison. *Gait Posture*, 31, 18-22.
- RAHMANIAN-SCHWARZ, A., BEIDERWIEDEN, A., WILLKOMM, L.-M., AMR, A., SCHALLER, H.-E. & LOTTER, O. 2011. A clinical evaluation of Biobrane® and Suprathel® in acute burns and reconstructive surgery. *Burns*, 37, 1343-1348.
- RASPOVIC, A., NEWCOMBE, L., LLOYD, J. & DALTON, E. 2000. Effect of customized insoles on vertical plantar pressures in sites of previous neuropathic ulceration in the diabetic foot. *The Foot*, 10, 133-138.
- RAYMOND, A. A., MECHIN, M. C., NACHAT, R., TOULZA, E., TAZI-AHNINI, R., SERRE, G. & SIMON, M. 2007. Nine procaspases are expressed in normal human epidermis, but only caspase-14 is fully processed. *British Journal of Dermatology*, 156, 420-7.
- REDMOND, A., ALLEN, N. & VERNON, W. 1999. Effect of scalpel debridement on the pain associated with plantar hyperkeratosis. *Journal Of The American Podiatric Medical Association*, 89, 515-519.
- REDMOND, A. C., LANDORF, K. B. & KEENAN, A. M. 2009. Contoured, prefabricated foot orthoses demonstrate comparable mechanical properties to contoured, customised foot orthoses: a plantar pressure study. *Journal of Foot and Ankle Research*, 2, 20.
- REIBER, G. E., VILEIKYTE, L., BOYKO, E. J., DEL AGUILA, M., SMITH, D. G., LAVERY, L. A. & BOULTON, A. J. M. 1999. Causal pathways for incident lower-extremity ulcers in patients with diabetes from two settings. *Diabetes Care*, 22, 157-162.
- RIBU, L., BIRKELAND, K., HANESTAD, B. R., MOUM, T. & RUSTOEN, T. 2008. A longitudinal study of patients with diabetes and foot ulcers and their health-related

quality of life: wound healing and quality-of-life changes. *Journal of Diabetes and its Complications*, 22, 400-7.

- RICHARD, S., QUERLEUX, B., BITTOUN, J., IDY-PERETTI, I., JOLIVET, O., CERMAKOVA, E. & LEVEQUE, J. 1991. In vivo proton relaxation times analysis of the skin layers by magnetic resonance imaging. *Journal of Investigative Dermatology*, 97, 120-125.
- RICHARD, S., QUERLEUX, B., BITTOUN, J., JOLIVET, O., IDY-PERETTI, I., DE LACHARRIERE, O. & LEVEQUE, J. 1993. Characterization of the skin in vivo by high resolution magnetic resonance imaging: water behaviour and age-related effects. *Journal of Investigative Dermatology*, 100, 705-709.
- RICHARDS, J. 2008. Ground reaction forces, impulse and momentum. *In:* RICHARDS, J. (ed.) *Biomechanics in clinic and research*. Philadelphia: Churchill Livingstone Elsevier.
- RINN, J. L., WANG, J. K., ALLEN, N., BRUGMANN, S. A., MIKELS, A. J., LIU, H., RIDKY, T. W., STADLER, H. S., NUSSE, R., HELMS, J. A. & CHANG, H. Y. 2008. A dermal HOX transcriptional program regulates site-specific epidermal fate. *Genes and Development*, 22, 303-7.
- ROBBINS, J. M., ROTH, L. S. & VILLANUEVA, M. C. 1996. "Stand down for the homeless". Podiatric screening of a homeless population in Cleveland. *Journal Of The American Podiatric Medical Association*, 86, 275-279.
- ROGERS, K., OTTER, S. J. & BIRCH, I. 2006. The effect of poron and plastazote insoles on forefoot plantar pressures. *British Journal of Podiatry*, 9, 111-114.
- RUBIN, L. 1949. Hyperkeratosis in Response to Mechanical Irritation1. *Journal of Investigative Dermatology*, 13, 313-315.
- RYAN, C. G., GRANT, P. M., TIGBE, W. W. & GRANAT, M. H. 2006. The validity and reliability of a novel activity monitor as a measure of walking. *British Journal of Sports Medicine*, 40, 779-784.
- RYU, H. S., JOO, Y. H., KIM, S. O., PARK, K. C. & YOUN, S. W. 2008. Influence of age and regional differences on skin elasticity as measured by the Cutometer®. *Skin Research & Technology*, 14, 354-358.
- SAGE, R. A., WEBSTER, J. K. & FISHER, S. G. 2001. Outpatient care and morbidity reduction in diabetic foot ulcers associated with chronic pressure callus. *Journal Of The American Podiatric Medical Association*, 91, 275-279.

- SAIDA, T., MIYAZAKI, A., OGUCHI, S., ISHIHARA, Y., YAMAZAKI, Y., MURASE, S., YOSHIKAWA, S., TSUCHIDA, T., KAWABATA, Y. & TAMAKI, K. 2004. Significance of Dermoscopic patterns in detecting malignant melanoma on acral volar skin. *Archives of Dermatology*, 140, 1233-1238.
- SALLES, A. S. & GYI, D. E. 2013. An evaluation of personalised insoles developed using additive manufacturing. *Journal of Sports Sciences*, 31, 442-50.
- SANDERS, J. E., GOLDSTEIN, B. S. & LEOTTA, D. F. 1995. Skin response to mechanical stress: Adaptation rather than breakdown - A review of the literature. *Journal of Rehabilitation Research and Development*, 32, 214-226.
- SANDERS, J. E., MITCHELL, S. B., WANG, Y. & WU, K. 2002. An explant model for the investigation of skin adaptation to mechanical stress. *IEEE transactions on biomedical engineering*, 49, 1626-1631.
- SANDILANDS, A., SUTHERLAND, C., IRVINE, A. D. & IRWIN MCLEAN, W. H. 2009. Filaggrin in the frontline: role in skin barrier function and disease. *Journal of Cell Science*, 122, 1285-1294.
- SANS, N., FARUCH, M., CHIAVASSA-GANDOIS, H., DE RIBES, C. L., PAUL, C. & RAILHAC, J. J. 2011. High-resolution magnetic resonance imaging in study of the skin: normal patterns. *European Journal of Radiology*, 80, e176-81.
- SANTOS, M., PARAMIO, J. M., BRAVO, A., RAMIREZ, A. & JORCANO, J. L. 2002. The expression of keratin k10 in the basal layer of the epidermis inhibits cell proliferation and prevents skin tumorigenesis. *Journal of Biological Chemistry*, 277, 19122-30.
- SATOR, P. G., SCHMIDT, J. B. & HONIGSMANN, H. 2003. Comparison of epidermal hydration and skin surface lipids in healthy individuals and in patients with atopic dermatitis. *Journal of the American Academy of Dermatology*, 48, 352-8.
- SCHMID-WENDTNER, M. H. & DILL-MULLER, D. 2008. Ultrasound technology in dermatology. *Seminars in Cutaneous Medicine and Surgery*, 27, 44-51.
- SERUP, J., KEIDING, J., FULLERTON, A., GNIADECKA, M. & GNIADECKI, R. 1995. High-frequency ultrasound examination of skin: introduction and guide. *In:* SERUP, J. & JEMEC, G. B. (eds.) *Handbook of non-invasive methods and the skin*. USA: CRC Press.
- SHARMA, M., DHANENDRAN, M., HUTTON, W. C. & CORBETT, M. 1979. Changes in load bearing in the rheumatoid foot. *Annales of the Rheumatic Diseases*, 38, 549-552.

- SHROUT, P. E. & FLEISS, J. L. 1979. Intraclass correlations: uses in assessing rater reliability. *Psychological Bulletin*, 86, 420-428.
- SIDDLE, H. J., REDMOND, A. C., WAXMAN, R., DAGG, A. R., ALCACER-PITARCH, B., WILKINS, R. A. & HELLIWELL, P. S. 2013. Debridement of painful forefoot plantar callosities in rheumatoid arthritis: the CARROT randomised controlled trial. *Clinical Rheumatology*, 32, 567-74.
- SILVER, R., HELMS, A., FU, W., WANG, H., DIACONU, D., LOYD, C. M., ROLLINS, A. M. & WARD, N. L. 2012. Using optical coherence tomography for the longitudinal non-invasive evaluation of epidermal thickness in a murine model of chronic skin inflammation. *Skin Research & Technology*, 18, 225-31.
- SIMKIN, A. 1981. The dynamic vertical force distribution during level walking under normal and rheumatic feed. *Rheumatology and Rehabilitation*, 20, 88-97.
- SIMON, M., HAFTEK, M., SERBAG, M., MONTEZIN, M., GIRBAL-NEUHAUSER, E., SCHMITT, D. & SERRE, G. 1996. Evidence that filaggrin is a component of cornified cell envelopes in human plantar epidermis. *Journal of Biochemistry*, 317, 173-177.
- SÍN, P., STUPKA, I. & BRYCHTA, P. 2010. Evaluation and comparison of composite and split-thickness skin grafts using Cutometer MPA 580. Annals of Burns and Fire Disasters, 23, 208-213.

SINGH, D. 1996. Callosities, corns and calluses. British Medical Journal, 312, 1403-1406.

- SMITH, W. P., CHRISTENSEN, M. S., NACHT, S. & GANS, E. H. 1982. Effect of lipids on the aggregation and permeability of human stratum corneum. *Journal of Investigative Dermatology*, 78, 7-11.
- SNOW, R. E., WILLIAMS, K. R. & HOLMES, G. B. 1992. The effects of weating high heeled shoes on pedal pressure in women. *Foot & Ankle*, 13, 85-92.
- SPEKSNIJDER, C. M., VD MUNCKHOF, R. J. H., MOONEN, S. A. F. C. M. & WALENKAMP, G. H. I. M. 2005. The higher the heel the higher the forefootpressure in ten healthy women. *The Foot*, 15, 17-21.
- SPINK, M. J., MENZ, H. B. & LORD, S. R. 2009. Distribution and correlates of plantar hyperkeratotic lesions in older people. *Journal of Foot and Ankle Research*, 2, 8.
- SPRINGETT, K. 1993. The influences of forces generated during gait on the clinical appearance and physical properties of skin callus. Doctoral thesis, University of Brighton.

- SPRINGETT, K. & MERRIMAN, L. 1995. Assessment of the Skin and its Appendages. In: MERRIMAN, M. M. & TOLLAFIELD, R. T. (eds.) Assessment of the Lower Limb. London: Churchill Livingstone.
- SPRINGETT, K. P., WHITING, M. F. & MARRIOT, C. 2003. Epidemiology of plantar forefoot corns and callus, and the influence of dominant side. *The Foot*, 13, 5-9.
- STEINER, M., AIKMAN-GREEN, S., PRESCOTT, G. J. & DICK, F. D. 2011. Side-byside comparison of an open-chamber (TM 300) and a closed-chamber (Vapometer) transepidermal water loss meter. *Skin Research & Technology*, 17, 366-372.
- STEINERT, P. M. & MAREKOV, L. N. 1997. Direct evidence that involucrin is a major early isopeptide cross-linked component of the keratinocyte cornified cell envelope. *Journal of Biological Chemistry*, 272, 2021-2030.
- STUCKE, S., MCFARLAND, D., GOSS, L., FONOV, S., MCMILLAN, G. R., TUCKER, A., BERME, N., CENK GULER, H., BIGELOW, C. & DAVIS, B. L. 2012. Spatial relationships between shearing stresses and pressure on the plantar skin surface during gait. *Journal of Biomechanics*, 45, 619-22.
- SUN, J. H., CHENG, B. K., ZHENG, Y. P., HUANG, Y. P., LEUNG, J. Y. & CHEING, G. L. 2011. Changes in the thickness and stiffness of plantar soft tissues in people with diabetic peripheral neuropathy. *Archives of Physical Medicine and Rehabilitation*, 92, 1484-9.
- SUNGYEON, A., SEUNGHUN, K., HAEKWANG, L., SEONGJOON, M. & IHSEOP, C. 2007. Correlation between a Cutometer® and quantitative evaluation using Moire topography in age-related skin elasticity. *Skin Research & Technology*, 13, 280-284.
- SWEENEY, D., NESTER, C. J., PREECE, S. & MICKLE, K. J. 2014. The effect of antipronation foot orthoses geometry on compression of heel and arch soft tissues. *Under review*.
- SWENSSON, O., LANGBEIN, L., MCMILLAN, J. R., STEVENS, H. P., LEIGH, I. M., MCLEAN, W. H. I., LANE, E. B. & EADY, R. A. J. 1998. Specialised keratin expression pattern in human ridged skin as an adaptation to high physical stress. *British Journal of Dermatology*, 139, 767-775.
- TAGAMI, H., KOBAYASHI, H. & KIKUCHI, K. 2002. A portable device using a closed chamber system for measuring transepidermal water loss: comparison with the conventional method. *Skin Research & Technology*, 8, 7-12.

- TAPPIN, J. W. & ROBERTSON, D. A. 1991. Study of the relative timing of shear forces on the sole of the forefoot during walking. *Journal Of Biomedical Engineering*, 13, 39-42.
- TASTEKIN, N., TUNA, H., BIRTANE, M. & UZANCA, K. 2009. Plantar pressure changes of patients with heel valgus in rheumatoid arthritis. *Turkish Journal of Rheumatology*, 26, 67-71.
- THOMAS, S. E., DYKES, P. J. & MARKS, R. 1985. Plantar Hyperkeratosis: A Study of Callosities and Normal Plantar Skin. *Journal of Investigative Dermatology*, 85, 394-397.
- TONG, J. W. & NG, E. Y. 2010. Preliminary investigation on the reduction of plantar loading pressure with different insole materials (SRP--Slow Recovery Poron, P--Poron, PPF--Poron +Plastazote, firm and PPS--Poron+Plastazote, soft). *The Foot*, 20, 1-6.
- TRONNIER, H. 1999. Results of the skin surface evaluation. *Cosmetics & Toiletries Manufacture Worldwide*, 213-219.
- TSUNG, B. Y. S., ZHANG, M., MAK, A. F. T. & WONG, M. W. N. 2004. Effectiveness of insoles on plantar pressure redistribution. *Journal of Rehabilitation Research and Development*, 41, 767.
- TU, C. L., CHANG, W., XIE, Z. & BIKLE, D. D. 2008. Inactivation of the calcium sensing receptor inhibits E-cadherin-mediated cell-cell adhesion and calciuminduced differentiation in human epidermal keratinocytes. *Journal of Biological Chemistry*, 283, 3519-28.
- TURKSEN, K. & TROY, T.-C. 2003. Overexpression of the calcium sensing receptor accelerates epidermal differentiation and permeability barrier formation in vivo. *Mechanisms of Development*, 120, 733-744.
- UHODA, I., FASKA, N., ROBERT, C., CAUWENBERGH, G. & PIERARD, G. E. 2002. Split face study on the cutaneous tensile effect of 2-dimethylaminoethanol (deanol) gel. *Skin Research & Technology*, 8, 164.
- VAN DER LEEDEN, M., STEULTJENS, M., DEKKER, J. H., PRINS, A. P. & DEKKER, J. 2006. Forefoot joint damage, pain and disability in rheumatoid arthritis patients with foot complaints: the role of plantar pressure and gait characteristics. *Rheumatology*, 45, 465-9.
- VAN ENGELEN, A., VLIG, M., BOEKEMA, B., SMIT, T., VERHAEGEN, P. D., RES,E. M., MIDDELKOOP, E. & VAN ZUIJLEN, P. P. 2008. Collagen orientation in

healthy skin and scar tissue: Measurements with the Reviscometer and digital image analysis.: Faculty of Sciences, Vrije Universiteit Amsterdam.

- VERHAEGEN, P. D., RES, E. M., VAN ENGELEN, A., MIDDELKOOP, E. & VAN ZUIJLEN, P. P. 2010. A reliable, non-invasive measurement tool for anisotropy in normal skin and scar tissue. *Skin Research & Technology*, 16, 325-31.
- VEXLER, A., POLYANSKY, I. & GORODETSKY, R. 1999. Evaluation of skin Viscoelasticity and anisotropy by measurement of speed of shear wave propogation with viscoelasticity skin analyzer. *Journal of Investigative Dermatology*, 113, 732-739.
- VIE, B., NESTER, C. J., PORTE, L. M., BEHR, M., WEBER, J. P. & JAMMES, Y. 2015. Pilot study demonstrating that sole mechanosensitivity can be affected by insole use. *Gait Posture*, 41, 263-8.
- VILANA, R., PUIG, S., SANCHEZ, M., SQUARCIA, M., LOPEZ, A., CASTEL, T. & MALVEHY, J. 2009. Preoperative assessment of cutaneous melanoma thickness using 10-MHz sonography. *American Journal of Roentgenology*, 193, 639-43.
- VINCENT, J. 1982. Structural biomaterials, New Jersey, Princeton University Press.
- VISWANATHAN, V., SNEHALATHA, C., SIVAGAMI, M., SEENA, R. & RAMACHANDRAN, A. 2003. Association of limited joint mobility and high plantar pressure in diabetic foot ulceration in Asian Indians. *Diabetes Research and Clinical Practice*, 60, 57-61.
- WANG, Y. N. & SANDERS, J. E. 2003. How does skin adapt to repetitive mechanical stress to become load tolerant? *Medical Hypotheses*, 61, 29-35.
- WARNER, R. R., BOISSY, Y. L., LILLY, N. A., SPEARS, M. J., MCKILLOP, K., MARSHALL, J. L. & STONE, K. J. 1999. Water disrupts stratum corneum lipid lamellae: damage is similar to surfactants. *Journal of Investigative Dermatology*, 113, 960-966.
- WERTZ, P. W. & VAN DEN BERGH, B. 1998. The physical, chemical and functional properties of lipids in the skin and other biological barriers. *Chemistry and Physics* of Lipids, 91, 85-96.
- WHITE, E. G. & MULLEY, G. P. 1989. Footcare for very elderly people: a community survey. *Age And Ageing*, 18, 276-278.
- WICKETT, R. R. & VISSCHER, M. O. 2006. Structure and function of the epidermal barrier. *American Journal of Infection Control*, 34, S98-S110.

- WIGLEY, C., KIRCKPATRICK, W. N. A., MILLER, M. C. & NANCHAHAL, J. 2008. Skin and its appendages. *In:* STANDRING, S. (ed.) *Gray's Anatomy: The anatomical basis of clinical practice*. 40 ed.: Churchill Livingstone Elsevier.
- WILLEMS, T. M., DE RIDDER, R. & ROOSEN, P. 2012. The effect of a long-distance run on plantar pressure distribution during running. *Gait Posture*, 35, 405-9.
- WOO, M. S., MOON, K. J., JUNG, H. Y., PARK, S. R., MOON, T. K., KIM, N. S. & LEE, B. C. 2014. Comparison of skin elasticity test results from the Ballistometer® and Cutometer®. *Skin Research and Technology*, 20, 422-428.
- WOODBURN, J. & HELLIWELL, P. S. 1996. Relation between heel position and the distribution of forefoot plantar pressures and skin callosities in rheumatoid arthritis. *Annals Of The Rheumatic Diseases*, 55, 806-810.
- WOODBURN, J., STABLEFORD, Z. & HELLIWELL, P. S. 2000. Preliminary investigation of debridement of plantar callosities in rheumatoid arthritis. *Rheumatology*, 39, 652-654.
- WU, K. S., VAN OSDOL, W. W. & DAUSKARDT, R. H. 2006. Mechanical properties of human stratum corneum: effects of temperature, hydration, and chemical treatment. *Biomaterials*, 27, 785-95.
- XHAUFLAIRE-UHODA, E., PIERARD-FRANCHIMONT, C., PIERARD, G. E. & QUATRESOOZ, P. 2010. Weathering of the hairless scalp: a study using skin capacitance imaging and ultraviolet light-enhanced visualization. *Clinical & Experimental Dermatology*, 35, 83-5.
- XIN, S., MAN, W., FLUHR, J. W., SONG, S., ELIAS, P. M. & MAN, M. Q. 2010. Cutaneous resonance running time varies with age, body site and gender in a normal Chinese population. *Skin Research & Technology*, 16, 413-21.
- YAFFE, M. B., BEEGEN, H. & ECKERT, R. L. 1992. Biophysical characterization of involucrin reveals a molecule ideally suited to function as an intermolecular crossbridge of the keratinocyte cornified envelope. *Journal of Biological Chemistry*, 267, 12233-12238.
- YAVUZ, M., ERDEMIR, A., BOTEK, G., HIRSCHMAN, G. B., BARDSLEY, L. & DAVIS, B. L. 2007. Peak plantar pressure and shear locations: relevance to diabetic patients. *Diabetes Care*, 30, 2643-5.
- YAVUZ, M., TAJADDINI, A., BOTEK, G. & DAVIS, B. L. 2008. Temporal characteristics of plantar shear distribution: relevance to diabetic patients. *Journal of Biomechanics*, 41, 556-9.
- YOUNG, B., LOWE, J. S., STEPHENS, A. & HEATH, J. W. 2006. *Wheater's Functional Histology: A test and colour atlas,* Philadelphia, Churchill Livingstone Elsevier.
- YOUNG, M. J., COFFEY, J., TAYLOR, P. M. & BOULTON, A. J. M. 1995. Weightbearing ultrasound in diabetic and rheumatoid arthritis patients. *The Foot*, 5, 76-79.
- YUNG-HUI, L. & WEI-HSIEN, H. 2005. Effects of shoe inserts and heel height on foot pressure, impact force, and percieved comfort during walking. *Applied Ergonomics*, 36, 355-362.
- ZAMMIT, G. V., MENZ, H. B., MUNTEANU, S. E. & LANDORF, K. B. 2008. Plantar pressure distribution in older people with osteoarthritis of the first metatarsophalangeal joint (hallux limitus/rigidus). *Journal of Orthopaedic Research*, 26, 1665-9.
- ZHAI, H. & MAIBACH, H. I. 2002. Occlusion vs. skin barrier function. Skin Research and Technology, 8, 1-6.